

INSTITUTE OF PUBLIC ADMINISTRATION THE PENNSYLVANIA STATE UNIVERSITY 211 BURROWES BUILDING UNIVERSITY PARK, PA 16802



NCJRS

FEB 9 1979

ACQUISITIONS

REPORT ON PRODUCTIVITY IMPROVEMENT IN THE BRIDGEPORT POLICE DEPARTMENT

6

ببين به أعملهم

by

John M. Stevens

and

Thomas C. Webster

Institute of Public Administration The Pennsylvania State University

December, 1977

The preparation of this report was made possible by the "City Hall/ University Application of Urban Research and Decision Technology through Continuing Education" Project, funded by the U.S. Department of Health, Education, and Welfare, Office of Education, under Community Service and Continuing Education, Title IA of the Higher Education Act of 1965 as amended, Grant No. G00-75-02104, Robert J. Mowitz, Project Director.

Title: Errata for the "Report on Productivity Improvement in the Bridgeport Police Department." Institute of Public Administration, December, 1977.

The following editorial changes should be made to this report:

at a se

- 1. Page xvi, EXECUTIVE SUMMARY, Paragraph 4, sentence 12, change relative, costs, to read "relative costs, ...
- 2.
- Page 3, paragraph 2, line 7, change Figure I-1 to Table I-1. Page 17, Figure II-4, Right Hand Bottom Code. Add words "Cross Hatched" 3. to read "Cross Hatched = High Incident"
- Page 20, Paragraph 2, line 3, change presents to "prevents". 4.
- Page 49, sub-paragraph 2, line 3, change spelling of Emphasise to 5. "Emphasize",
- 6. BIBLIOGRAPHY, correct alphabetical ordering of O's and P's.

TABLE OF CONTENTS

EXECUTI	VE S	UMMARY	x
I.	INI	RODUCTION	1
	Α.	Goals and Objectives	1
	B.	Major Strategies and Emphases	2
	c.	Scope of the Study	3
	D.	Central Issues in Police Productivity	6
	E.	Organization of the Report	7
II.	UNI	VARIATE ANAYLSIS	9
	Α.	Demand for Services	10
	В.	Location	13
	C.	Victim Information	18
	D.	Time	20
	Ε.	Conclusions	20
III.	BIV	ARIATE ANALYSIS	23
	A.	Demographic Information	23
	Β.	Location of Incident	25
	C.	Time of Incident	27
	D.	Time and Location of Incident	28
	E.	Reporting	29
IV.	IMP	ACT-EFFECTIVENESS — EXTENDED ANALYSIS	30
	A.	Response Time	30
	в.	Extended Analysis	31
	С.	Major Car Post Sectors by Crime Type by Twelve	20
	n	Noior Origo Turos by Colosted Orida Controlling	32
	D.	For Time of Day	35
	F	Major Crime Tune by Major Car Doct Sectors Controlling	55
	و بنا	For Sessonal Trande	36
	ਸ	Crime Type by Location Status Controlling For the	50
	£. •	Major Car Post Sectors	38

Page

TABLE OF CONTENTS (Continued)

ν.	SUM	MARY OF MAJOR FINDINGS	42
	A. B. C. D.	General Overview Frequency Analysis Impact-Effectiveness and Extended Analysis Extended Analysis of Crime Code by Location Controlling for Major Car Post Sectors	42 42 45 46
VI.	PRO	DUCTIVITY IMPROVEMENT RECOMMENDATIONS	48
	A. B. C.	Information System Patrol Activities Organizational Realignment	48 50 51
Appendix	A:	Detailed Crime Code Classification	52
Appendix	B:	Bridgeport Incident Report Form	59
Appendix	С:	Chapter II Tables	60
Appendix	D:	Chapter III Tables	77
Appendix	E:	Chapter IV Tables1	.34
Bibliogra	ıphy	for Police Productivity Studies	

D

Page

iii

LIST OF FIGURES

Chapter I

Figure

I-1

Page

Frequency Percentage Comparing Crime Type in Bridgeport Population Data and Sample Data for 1975 4

LIST OF FIGURES

15

Chapter II

Figure		Page
11-1	Crime Code High Frequency Incident Types Sample Data 1975	12
II-2	Location Index for Bridgeport Police Department	14
II~3	Grids with 50 Plus Incidents	15
II-4	High Incident Frequencies by Street Location Plotted by Grid	17
11-5	Incident Location Type Sample Data 1975	19
II-6	Month Incident Occurred Sample Data 1975	21

LIST OF TABLES

CHAPTER 11

<u>Table</u>		Page
II- 1	Frequency Percentage Comparing Crime Type in Bridgeport Population Data and Sample Data for 1975.	59
II- 2	Crime Code - Collapsed Categories	61
II- 3	Car Post Sectors	62
II- 4	Grid Code Sample Data Street Locations with High Frequencies of Incidents	63
II- 5	Grid Code - Population Data	64
II- 6	Incident Location Type	65
II- 7	Age of Victims	66
II- 8	Sex of Victim	67
II- 9	Race of Victim	68
II-10	Month Incident Occurred	69
II-11	Month Incident Reported	70
II-12	Day of Month Incident Occurred	71
II-13	Day of Month Incident Reported	72
II-14	Time Incident Occurred/Reported: A.M P.M.	73
II-15	Time Incident Occurred/Reported: 24 Hour Clock	74
II-16	Time Incident Occurred/Reported: 12 Hour Segments	75

LIST OF TABLES

Chapter III

0

.

Table		Page
III-1	Victim's Sex by Victim's Race	76
III-2	Victim's Age by Victim's Sex	77
III-3	Victim's Age by Victim's Race	78
III-4	Victim's Sex by Newcrime	79
III-5	Victim's Race by Newcrime	81
III-6	Victim's Age by Newcrime	83
III-7	Victim's Sex by Car Post Sector	85
III-8	Victim's Race by Car Post Sector	87
III-9	Vicitm's Age by Car Post Sector	89
III-10	Car Post Sector by Newcrime	91
III-11	Grid by Crime Code	95
III-12	Victim's Sex by Incident Location Type	97
III-13	Incident Location Type by Victim's Race	100
III-1 4	Car Post Sector by Incident Location Type	102
III-15	Time Incident Occurred (A.MP.M.) by Date Incident Occurred (Month)	105
III-16	Time Incident Occurred (3 P.M./2 A.M.—3 A.M./2 P.M.) by Date Incident Occurred (Month)	106
III-17	Date Incident Occurred (Month) by Time Incident Occurred (24 Hour Clock)	107
III-18	Time Incident Occurred (A.MP.M.) by Newcrime	110
III-19	Time of Incident (3 P.M./2 A.M.—3 A.M./2 P.M.) by Newcrime	112
III-20	Time Incident Occurred (24 Hour Clock) by Newcrime	113

<u>Table</u>		Page
III-21	Date Incident Occurred (Month) by Newcrime	115
III-22	Time Incident Occurred (A.MP.M.) by Car Post Sector	117
III-23	Time Incident Occurred (3 P.M./2 A.M3 A.M./2 P.M.) by Grid	118
III-24	Time Incident Occurred (A.MP.M.) by Incident Location Type	119
III-25	Date Incident Occurred (Month) by Car Post Sector	121
III-26	Date Incident Occurred (Month) by Grid	124
III-27 [.]	Date Incident Occurred (Month) by Incident Location Type	126
III-28	Time Incident Occurred (A.MP.M.) by Time Incident Reported (A.MP.M.)	130
III-29	Date Incident Occurred (Month) by Date Incident Reported (Month)	131

LIST OF TABLES

Chapter IV

Table]	Page
IV-1	Response Time by Crime Code	133
IV-2	Frequency Distribution of the Four Major Car Post Sectors	135
IV-3	Frequency Distribution of Major Grid Locations	136
IV-4A	Major Car Post Sector by Crime Code Controlling for the 3 P.M. to 2:59 A.M. Time Segment	137
IV-4B	Major Car Post Sector by Crime Code Controlling for the 3 A.M. to 2:59 P.M. Time Segment	138
IV-5A	Major Gild Gategories by Crime Code Controlling for the 3 P.M. to 2:59 A.M. Time Segment	139
IV-5B	Major Grid Categories by Crime Code Controlling for the 3 A.M. to 2:59 P.M. Time Segment	140
IV-6A	Car Post Sector by Crime Code Controlling for the Winter Season	141
IV-6B	Car Post Sector by Crime Code Controlling for the Spring Season	142
IV-6C	Car Post Sector by Crime Code Controlling for the Summer Season	143
IV-6D	Car Post Sector by Crime Code Controlling for the Fall Season	144
IV-7A	Crime Code by Location Status Controlling for the Red Car Post Sector	145
IV-7B	Crime Code by Location Status Controlling for the Green Car Post Sector	146
IV-7C	Crime Code by Location Status Controlling for the Amber Car Post Sector	147
IV-7D	Crime Code by Location Status Controlling for the Blue Car Post Sector	148

Đ

...)

EXECUTIVE SUMMARY

a

Ó

 \bigcirc

11

3

Û

EXECUTIVE SUMMARY

A. Introduction

This report presents the results of a productivity analysis of the Bridgeport, Connecticut, Department of Police conducted by the Institute of Public Administration of The Pennsylvania State University from December 1976 through August 1977. This study is part of a continuing education project conducted by the Institute in cooperation with the Bridgeport City government. The overall objective of the study is to develop and apply management technology, information processing, and analytical methods to improve decisionmaking and effectiveness in meeting public needs.

B. Objectives of the Report

The primary focus of this study was to determine if realistic increases in productivity can be made through improved management of patrol operations in the Bridgeport Department of Police. Other specific objectives of the study were to: (a) assess the demand for police services in Bridgeport; (b) examine and evaluate time and location patterns of calls for crime and non-crime service; (c) develop methods to assess effectiveness; (d) derive methods for reducing crime and increasing apprehension rates, and (e) provide recommendations directed towards increasing police productivity.

х

C. Methodology for the Analysis

To achieve the stated objectives, researchers chose to analyze crime and noncrime demands and activities for one calendar year as documented in agency records. Data from the 1975 information system was utilized instead of 1976 because much of the 1976 data were not recorded on the computer file. The 1977 data were not available because of manpower shortages that occurred when certain CETA positions were not filled. Because of the large number of service calls, a 10 percent sample was drawn from the 80,000 incidents recorded on the computerized 1975 police incident reports. This sample was validated against the population distribution of crime and noncrime calls and was found to be consistent across all categories of calls for service. Two tables comparing the results are presented in Chapter I of this report.

The report also discusses data collection, analytical methods, and reviews some of the substantive and methodological issues in studies of police productivity. The analytical techniques employed range from univariate to bivariate analysis with control methods applied to elaborate on significant trends identified in antecedent analyses.

D. Major Findings

Demand for Police Services

The National Advisory Commission on Criminal Justice Standards and Goals states that every police agency should develop a patrol deployment system that is responsive to the demand for services.

xi

Measures of the amount and type of crime occurring at specified times and in certain locations provide valid indicators of the intensity and type of demand for police action. For example, the crime and noncrime distribution patterns in Bridgeport indicate that police respond most frequently to reports of medical problems (10 percent), motor vehicle theft (10 percent), nuisances (10 percent), property damage (7.5 percent), and alarms (7 percent). In addition to motor vehicle theft, burglary (5 percent), larceny offenses (10 percent), assault (4.6 percent), disorderly conduct (4.3 percent), and motor vehicle violations (4.2 percent) constitute major crime categories. Family offenses (3 percent) also require significant amounts of police service time.

Location of Patterns of Calls

The demand for police services are very frequent within car post sectors — green, amber, blue and red. These four sectors account for approximately 66 percent of the total incidents recorded. Also, two special sections — ambulance and dog warden — accounted for an additional 14.5 percent of the incidents served. In addition, high incident grids were computed by selecting street locations which registered frequencies of 50 or more incidents and matching them with the proper grid number. The grids with street locations having the most frequent rate of incidents are 008A, 009, 017, 011, 016, 003, 010, 027, 008b, and 999. The nine grids which are illustrated in Figure II-4 in the text account for approximately 76 percent of the incidents. This location-based segment of the analysis indicates that there are definite patterns of demand for police services in Bridgeport.

Time Variables

The distribution of incidents by month indicates a relative pattern of increased demand for service during the summer months---May through August. A definite pattern also exists for the distribution of incidents by hour using the 24-hour clock. When the 24-hour clock was divided into two 12-hour time segments, 3 p.m. to 2:59 a.m. and 3 a.m. to 2:59 p.m., it was found that 61.5 percent of the incidents occur and are reported in the 3 p.m. to 2:59 a.m. time period. In addition, the hours between 0300 and 0700 are the low frequency hours and calls tend to increase during the morning and afternoon hours with a noticeable increase around 1600. The peak demand period for calls is 2100.

Incident Type and Location

The examination of crime types by specific locations can provide useful information for devising optimum patrol strategies. For example, family offenses are a frequent source of incidents in the red car post sector; robberies, burglaries, and family offenses are frequent in the blue district; and the amber sector exhibits a definite pattern of larceny incidents. Family offenses are high in Grid 003; Grid 009 has a high percentage of larcenies and Grid 009a has a high percentage of medical problem calls.

The major Part I crime across 2 ne four major car post sectors between the hours of 3 p.m. to 2:59 a.m. is motor vehicle theft. Alarm and miscellaneous juvenile type calls are also common occurrences across these four car post sectors during this 12-hour

xiii

period. In addition, burglary is a common call serviced across the four major car post sectors during the early morning and daytime time segment--3 a.m. to 2:59 p.m.

Impact-Effectiveness

Documentation of data on time of call, response time, on-scene time, on-scene service time, back in-service time, number of units dispatched, number of officers dispatched, and the clearance status of incidents is necessary if management is to evaluate performance. However, because certain complications with CETA workers arose, the data was not properly transposed from the reports to the computer files. This type of information loss has precluded the use of the 1975 and 1976 file date for an adequate impact/effectiveness analysis of police services. Therefore, for illustrative purposes, methodological procedures for measuring effectiveness using available data have been presented in Chapter IV.

E. Productivity Improvement Recommendations

The following recommendations are based upon the analyses performed in the study. They represent some of the alternatives available to the Bridgeport police department for examining their operations and improving productivity. It must also be noted here that officials of the Bridgeport Police Department have recognized some of the needs documented here and have already taken remedial actions.

Information System Improvements

The police management officers should establish a priority system for the most critical information. For example, information relating to response times, witnesses, and clearance of crimes may be considered more important than other variables such as age. Whatever guidelines are adopted, a systematic approach to reporting and data collection is required to conduct a valid work load analysis. The major recommendation is to improve the collection, reporting, coding, and audit process to provide the basis for a patrol centered productivity analysis. The following steps may be taken:

1. Determine the most active time, location, and type of call information, then strictly supervise the collection, transcription, and quality control of the data entered into the computer records system. In conjunction with this control procedure, management must emphasize the importance of complete and accurate reporting by periodically providing patrolmen with feedback about how the reports are utilized to help them perform their functions.

2. Management should institute a formal periodic auditing procedure to review incident reports, computer entry and summaries of data that are being stored in the computer file. This auditing responsibility should reside with the officer who has authority over the management information system.

xv

- 3. Collect relevant resources and effectiveness data, such as the number of officers and units responding to a call; response time; time on the scene; action time, and the manner in which the incident was cleared. This information should be maintained on file to enable police management to determine response time patterns, patrol and unit work load, and the impact or effectiveness of police patrol operations.
- 4. Explore the possibility of procuring a flexible, analytical software system which is compatible with the existing IBM hardware. For instance, the analyses in this report were produced by the SPSS system (Statistical Package for the Social Sciences). This package is designed to allow for relatively rapid learning and manipulation of data for statistical analysis by line personnel who are not technical experts in computer programming. Latest quotations indicate that the SPSS system capability costs approximately \$1,400.00 for the first year and \$600.00 for maintenance in subsequent years. Other alternatives may also be examined in light of the relative, costs, needs of the department, and benefits.

Patrol Activities

Patrol scheduling and assignment should be developed to correspond to demand in grid areas with the highest frequency of incidents, i.e., Grids 009a, 009, 017, 011, 016, 003, 010, 027, and 008b. Furthermore, calls for service tend to increase between

xvi

the hours of 3 p.m. and 3 a.m.; therefore, appropriate scheduling could match manpower resources to meet demand for service according to time and location. It is realized that union actions and negotiated contracts have an impact upon scheduling of manpower, however, the results indicate that the public would be better served if patrol resources were matched with demands.

Organizational Structure Examination

Explore the feasibility of transferring the housing and administration of the ambulance service from the Police Department to the Fire Department. A realignment may relieve the Police Department of some of the administrative burden imposed by responding to medical problems which presently account for 10.3 percent of the demand for police service in the City of Bridgeport.



INTRODUCTION

£



Chapter I

INTRODUCTION

This report presents the results of a productivity study conducted in the City of Bridgeport, Connecticut, Police Department during the period of December, 1976 to August, 1977. This study was carried out as one segment of a continuing education project whose objective is to apply management technology and methodology to small and medium-size municipalities.

The central thrust of this report was to focus upon actual police activities as documented in official reports and records in order to provide the foundation for informed productivity recommendations and actions. The approach taken was to formulate a demand and response framework for police activities which could be analyzed to elicit useful information for planning and management purposes in the city.

A. Goals and Objectives

Since a common oversight of research in police work is the failure to articulate goals and objectives clearly with no clear translation into actions, the objectives of this study are presented below (Gass and Dawson, 1974).

67)

A major objective of this study is to determine what increases in productivity can be made through the management of patrol operations in the Bridgeport Police Department. Other specific objectives are to arrive at some determination about the present level of police protection in Bridgeport; examine time and location patterns of crime and calls for service; arrive at methods for reducing crime and increasing apprehension rates; and to provide general or specific recommendations that are oriented towards increases in overall police productivity.

Finally, it is the purpose of this report to provide an analytical basis for examining police activities across various size cities so that the police management process focuses upon the realities rather than images of police work.

B. Major Strategies and Emphases

The focus in this study was upon examining and evaluating actual police activities as documented in the computerized records system of the Bridgeport Police Department. The information in the computerized files include time, location, victim, crime type, and general administrative information relevant to the incident. Given the informational nature of an updated computer file, the aggregation of the data in the file constitutes an overall measure of crime and noncrime demand for police services in the city of Bridgeport.

The purpose of using this source of data was to measure actual police activities in terms of community demand and police response characteristics. This approach allowed the study members to concentrate upon the interaction between the demands for police services and productivity related tasks performed in conjunction with the demand. The critical elements of the police task have to be first defined and circumscribed within the overall context of police work before legitimate, work-related, productivity strategies are to be formulated.

Another major objective of this strategy was to demonstrate the utility of evaluating existing data documentation sources for developing management, planning, and productivity actions. This approach also provides for the capability to evaluate the adequacy of existing procedures and improve the information gathering system and procedures. It is intended that the information and strategies being developed in this report will be used specifically to analyze ongoing police activities in Bridgeport and provide a basis for future management evaluations in Bridgeport or other municipalities.

C. Scope of the Study

The Bridgeport Police Department experienced approximately 81,000 calls for service in the base year for the analysis - 1975. Data from the 1975 base year was utilized because the 1976 data file was found to be essentially incomplete, inconsistent and hence virtually nonanalyzable. However, due to the large number of cases in the 1975 file a representative 10% sample was drawn from the 50 major crime categories. Figure I-1 presents a number and percentage comparison between the population data and sample to validate the accuracy and representativeness of the sample.

Figure I-1

FREQUENCY PERCENTAGE COMPARING CRIME TYPE IN BRIDGEPORT POPULATION DATA AND SAMPLE DATA FOR 1975

	Population	(80,966)	Sample	(8,017)
Crime Type	Number	Percent	Number	Percent
Murder-Manslaughter	19	0.0%	1	0.0%
Robbery	697	0.9	59	0.7
Assaults	3851	4.8	365	4.6
Burglary	3806	4.7	391	4.9
Larceny LE \$50	3486	4.3	334	4.2
Larceny GT \$50	4295	5.3	444	5.5
Motor Vehicle Theft	7720	9.5	815	10.2
Forgery	187	0.2	15	0.2
Other Larceny/ Fraud Embezzlement	419	0.5	45	0.6
Stolen Property	31	0.0	6	0.1
Weapons	155	0.2	18	0.2
Vice	155	0.2	11	0.1
Sex Offenses	138	0.2	18	0.2
Obscenity	5	0.0	1	0.0
Drugs	562	0.7	63	0.8
Liquor Laws	157	0.2	20	0.2
Drunkedness	821	1.0	88	1.1
Kidnapping	26	0.0	1	0.0
Arson	117	0.1	16	0.2
Gambling	252	0.3	17	0.2
Disorderly Conduct	3382	4.2	342	4.3
Bribery	17	0.0	2	0.0

Figure I-1 (Continued)

FREQUENCY PERCENTAGE COMPARING CRIME TYPE IN BRIDGEPORT POPULATION DATA AND SAMPLE DATA FOR 1975

	Population (80,966)		Sample (8,017)	
Crime Type	Number	Percent	Number	Percent
Family Offenses	2417	3.0%	238	3.0%
Property Damage	6290	7.8	598	7.5
License Violations	29	0.0	2	0.0
Code Violations	44	0.1	5	0.1
Court Violations	116	0.1	12	0.1
Impeding Justice	228	0.3	29	0.4
Nuisances	8617	9.9	811	10.1
Escapes	66	0.1	5	0.1
Suicide	90	0.1	9	0.1
Dead on Arrival	320	0.4	38	0.5
Missing Persons	1208	1.5	115	1.4
Found Persons	110	0.1	9	0.1
Suspicion	1476	1.8	140	1.7
Alarms	5738	7.1	554	6.9
Property	816	1.0	78	1.0
Investigations	276	0.3	31	0.4
Accidents	905	1.1	88	1.1
Motor Vehicle Violations	3550	4.4	340	4.2
Medical Problems	8402	10.4	824	10.3
Warrant-Subpoena	834	1.0	76	0.9
No Incident	2407	3.0	233	2.9
Miscellaneous	7279	9.0	710	8.9
Totals	80916	99.8%	8017	100.0%

D. Central Issues in Police Productivity

Costs are easier to identify than are the elements of police services, therefore, it is difficult to equate any given financial figure with either increases or decreases in police performance. A recent report on police productivity presents four ways in which police productivity may be considered:

- 1. Increasing police productivity means improving current police practices to the best level known, to get better performance without a proportionate increase in cost.
- 2. Increasing police productivity means allocating resources to activities which give the highest return for each additional dollar spent.
- 3. Given the uncertainties of police work, increasing productivity means increasing the probability that a given objective will be met.
- 4. Increasing productivity in police work means making the most of the talents of police personnel (National Commission on Productivity, 1973).

In addition to the resource constraints and lack of comprehensive frameworks for studying police productivity, another substantial factor remains largely unexplored. A recent report on efficiency and effectiveness in "Big City Police Departments," concedes that police perform a multiplicity of tasks but the self-image of police officers is rooted in their crime fighting function (Skogan, 1976). This substantive issue is addressed in the study by focusing upon what duties are actually performed in response to the demands of the community.

The operating premise of this study is that some hard, analytical

data is needed for the purposes of comparison, planning and controlling from a management perspective. The data collected utilize both the self-image (Skogan, 1976) activities of the police departments (Part I and Part II) crimes, and the other support services provided to the community such as responding to vagrancy, serving warrants, or handling traffic complaints. Only in this way can police management begin to plan for more efficient use of constrained resources and plan for more effective provision of <u>critical</u> services. A valid foundation based upon accurate information is needed before future goals and objectives can even be considered. In order to increase productivity, we must first know what demands are being made by the citizens and to what extent scarce resources can be allocated to meet these demands.

E. Organization of the Report

The report is designed to present the results of the productivity study to both city and police department management officials. Chapter Two presents the specific demand analysis for police services. This includes both crime (Parts I and II) and other noncrime police activities. Chapter Three presents bivariate analyses that examine several crime code associations. This chapter also covers bivariate analyses of time and location data. Chapter Four presents impact and effectiveness assessments that are based upon an extended analyses of crime and response patterns. Chapter Five includes the summary and conclusions for the major findings. Chapter Six presents the productivity improvement recommendations that correspond to the major findings and conclusions.

The specific analytic techniques used will be described and discussed as they are introduced into the specific chapters.

II.

 \mathfrak{h}

6 H

0

UNIVARIATE ANALYSIS



Chapter II

UNIVARIATE ANALYSIS

This chapter discusses distributional patterns of crime-type, situation, location, and time demands for police service in the City of Bridgeport. Demand for police service was analyzed based on data gathered from a sample of 8017 incidents drawn from records of over 80,000 incidents recorded on the 1975 police incident-file reports.

The 1975 sample was used for analysis rather than the 1976 sample because it provided more accurate and consistent data on the distribution of incidents across the crime-type situation, location, and time variables identified in the study. The sample was validated by comparing the adjusted frequency of the population distribution (N=80, 966) of the 50 category crime code with the adjusted frequency of the sample distribution (N=8017) of the 50 category crime code. The difference between the population and sample adjusted frequency distributions ranged between .1 to .7 percent for each crime code. Based on the relative similarity between the adjusted frequencies, the 1975 sample was judged as a reliable indicator of the amount and type of crime in the City of Bridgeport (See Table II-1, APPENDIX C).

Analyses utilizing the 1975 sample data revealed inconsistencies with coding and missing data. Large amounts of missing data prevented an exhaustive analysis of many operation centered variables in the sample. Many variables that contained less than 25 percent valid data were deleted from the analysis. Examples of deletions are: age of victims, victim-offender relationship, incident type, manner committed, police division to which the incident was referred, witness, suspect, and response time to incidents. Many of these variables would have provided a sound basis for analysis and an assessment of operational effectiveness.

However, sufficient patterns have been identified which demonstrate the distribution of crimes, noncrime, location, and time characteristics of incidents occurring in the City. The remainder of this Chapter discusses the distribution of consistent patterns as they relate to demands for police services.

A. Demand for Services

The amount and type of crime occurring per year provides a good indicator of the level and particular kinds of demand placed on police services. To provide this information - crime codes from the <u>Bridgeport Police Department - Code Book*</u> were matched with corresponding data on reported yearly levels of types of crime to create two variables: Variable 105 - crime code-recoded, and Newcrimecollapsed categories of crime. Variable 105 - collapses the four digit crime codes in the <u>Code Book</u> into 50 coded categories. Variable Newcrime further reduces the 50 coded categories into 19 categories. Variable Newcrime captures the crime-type categories with high frequencies but combines the low frequency categories into three general categories: Vice, other-person, and other-property. The more inclusive, with 50 categories was used for the frequency distributions, used with the univariate descriptive, while Newcrime *See Appendix A - Bridgeport - Crime Code Book

with 19 categories was used for the bivariate analysis.

The analysis of frequency distributions presented in Appendix indicates that police respond most frequently to reports of medical problems (10.3 percent), motor vehicle thefts (10.2 percent), nuisances (10.0 percent), property damage (7.5 percent), and alarms (6.9 percent). In addition to these calls, moderately high demand exists for service in response to Part I crimes such as larceny greater than \$50 (5.5 percent), larceny less than \$50 (4.2 percent), and burglary (4.9 percent). Significant amounts of police time are also committed to incidents such as assault (4.6 percent), disorderly conduct (4.3 percent), motor vehicle violations (4.2 percent), and family offenses (3.0 percent). Figure II-1 and Tables II-1 and II-2 demonstrate these patterns.

Many of the incidents in the 50 category crime code contain high percentage distributions of specific degrees of the various types of crime. For instance, medical assistance (8.2 percent) is the primary component of medical "problems" to which police respond. Animal nuisances comprise 7.2 percent of the nuisance complaints. Reports of criminal mischief on private property comprised 7.2 percent of the requests for police service for reported incidents of property damage. Burglary and holdup alarms accounted for 5.9 percent of the alarm category.

Part I crimes involving larceny (less than \$50 and greater than \$50) account for 9.7 percent of demands for police service. Shoplifting (.9 percent) and purse snatching (.5 percent) account for 1.4 percent of the incidents of larceny less than \$50. Incidents involving Motor Vehicles account for almost half (2.5 percent) of the reported incidents of larceny greater than \$50.


Burglary is a major Part I crime and accounts for 4.9 percent of the incidents requiring police service. Forcible commercial burglaries account for 1.4 percent of the incidents, while forcible residential burglaries account for 1.1 percent of the incidents.

Demands for service arising from incidents such as assault, Disorderly Conduct, Motor Vehicle Violations, and Family Offenses account for a considerable level of police activity. Simple assault accounts for 2.5 percent of the assaults. Motor Vehicle Violations account for 4.2 percent of the demand for service. Evading responsibility accounts for 2.5 percent of the Motor Vehicle Violations. Family offenses account for 3.0 percent of incidents requiring police service. Family disputes account for 2.8 percent of Family Offenses.

B. Location

Definite location patterns can be demonstrated for the incidents requiring police action. Figure II-2 shows the location index for incidents by Car Post Sector and Grid. There are four Car Post sectors and 28 grids. Some of the grids have been broken down into different sections to further discriminate location patterns.

Table II-3 and Figure II-1 can be used to show that demand for police service is most frequent in car post sector Green (23.8 percent), Amber (16.4 percent), Blue (15.3 percent), and Red (10.9 percent). Car post sectors Green, Amber Blue account for 66.4 percent of the incidents recorded. Two special sections - Ambulance (8.2 percent) and Dog Warden (6.3 percent) account for an additional 14.5 percent of the incidents. To further demonstrate location patterns, high incident areas of the City have been plotted by police Grid locations on Figure II-3. High

Figure II-2

Location Index for Bridgeport Police Department*



Figure II-3 Grids with 50 Plus Incidents



incident Grids were determined by selecting street locations (from the street location code) which registered frequencies cf 50 or greater and matching them with the proper Grid number.

This selection was validated by comparing the adjusted frequency percents of high incident street locations from the population distribution with those from the sample. Tables II-4 and II-5 show the difference between the frequencies of identical street locations from • the population and the sample range from .2 to 1.3 percent.

Grids with street locations having the most frequent rate of incidents are: (Refer to Figure II-3 and Table II-4)

Grid	Frequency	Adj. Freq. <u>Pct. (N=2306)</u>	Street Location	
008A	395	16.5%	Father Panik Village	
009	345	14.4	1-490 731-1610	Fairfield Avenue Main Street Police Headquarters Laffayette Shopping Mall
017	246	10.3	531-2050 811-1710 341 up 851 up	Boston Avenue E. Main Street Hallet Street Pembroke Street
011	174	7.3	701–1300 725–1964 691–1195	Fairfield Avenue North Avenue State Street
016	168	7.0	1-530 1-1400	Boston Avenue Noble Avenue William Street
003	152	6.3	P. T. Bar	rnum Apartments
010	130	5.4	1-724 561-1740	North Avenue Park Avenue
027	129	5.4	Beardsle	y Terrace
008B	100	4.2	971 up	Stratford Avenue
999	88	3.8	Out of C:	ity Locations

Figure II-4

HIGH INCIDENT FREQUENCIES BY STREET LOCATION PLOTTED BY GRID



High Incident Frequencies by Street Location Plotted by Grid It must be noted that the Variable Grid frequency distribution does not represent the total distribution for all incidents occurring by Grid in the City. However, the nine highrate Grids account for approximately 76 percent of the incidents by street location. Also, 3.8 percent of the demand for police service involves out of City locations.

Table II-6 provides information on the type of location where incidents are occurring in the City. The most frequent location types are apartment dwellings (26.1 percent), private automobiles (19.6 percent), housing (12.2 percent), businesses (8.0 percent), and streets (6.0 percent). These five incident location types account for 72 percent of the incidents occurring in Bridgeport. Figure II-5 illustrates this pattern more clearly.

C. Victim - Suspect Information

Table II-7 shows the distribution of victim age, however, the results must be interpreted carefully bacause there were 6,740 missing cases. In the 1277 cases where the age information was available, it was shown that approximately 57 percent of the victims are between 19 and 45 years old. Approximately 13 percent are over 61 years old.

Table II-8 shows that 56 percent of the victims are male and approximately 65 percent of the victims are white, 26 percent black, 9-10 percent Puerto Rican. This distribution is based upon 2043 cases and should be interpreted with control.

The substantial amount of missing information for crime victims does not allow for any firm conclusions about who is most victimized in the City of Bridgeport.



D. Time

Tables II-10 and Figure II-6 indicate increased rates of incidents reported for the summer months - May thru August. Also, as indicated in Tables II-12 and II-13, there is no conclusive pattern of incidents reported by day of month. However, as shown in Table II-14, there are almost twice as many incidents in the P.M. hours (approximately 6.5 percent) than in the A.M. hours. The hours between 0300 and 0700 are the lowest frequency periods for reports. Reported incidents begin to increase at 0800 hours, hold steady at about 5 percent between 1000 and 1500 hours, then increase to 6 percent at 1600 hours and peak to 7 percent at 2100 hours (See Table II-15). This same pattern can be observed by dividing the clock into two 12 hour segments - 3 p.m. to 2 a.m., 3 a.m. to 2 p.m, Table II-16 shows that incidents occur and are reported approximately twice as frequently during the 3 p.m. to 2 a.m. segment (approximately 40 percent).

E. Conclusions

This chapter has discussed the patterns of demand according to crime-type, location, situation, and time. The great degree of missing information presents definite conclusions about victim and suspect patterns. However, several preliminary conclusions may be drawn from the location, time, and type of crime analyses. The first is that police management must supervise the reporting information collection, and computer entry process. These provisions must be developed to provide an accurate and complete data base for analysis and planning purposes. This presupposes a priority system for determining what information





is the most important. Another conclusion is that demand for police service is most frequent for: Medical Problems (10.3 percent), Motor Vehicle Theft (10.1 percent), and Nuisances (10.1 percent). Major Part I crimes account for 24.8 percent of the demand for police service. A third conclusion is that certain accounts for the major part of the calls: 008A (16.5 percent), 009 (14.4 percent); 017 (10.3 percent); 011 (7.3 percent); 016 (7 percent); 003 (6.3 percent); 010 (5.4 percent); 027 (5.4 percent); 008B (4.2 percent). A fourth conclusion is that reported incidents occur most frequently in apartments (26.1 percent); private autos (19.6 percent); houses (12.2 percent); businesses (8percent); and street (6 percent). In addition, incidents occur and are reported frequently during the summer months and between the hours of 3 p.m. and 2 a.m. (61.5 percent). These conclusions have implications for the information collection and recording system, patrol scheduling, and manpower assignments functions of Bridgeport Police Department. From the analyses presented, productivity improvements actions can be taken to measure the effectiveness of ongoing operations. Specific recommendations will be proposed in subsequent chapters of this report.

Ø

III.

BIVARIATE ANALYSIS

-



CHAPTER III

BIVARIATE ANALYSIS

The bivariate analysis was limited to the demographic variables (sex, race, age), time variables (time incident occurred, date of incident), location variables (car post sector, incident location type, grid), and the situation variable (crime code) because of data availability. The tables for Chapter Three are in Appendix D.

A. Demographic Information

When establishing the various types of crime suppression procedures to be followed, it is helpful to understand who is being victimized. This section presents a brief profile of those involved in incidents in Bridgeport in 1975.

Looking at the demographic variables (sex, race, age), one finds that victims of the incidents tended to be male and older individuals from all racial groups. Table III-1 indicates that most victims were male with the exception that a higher percentage of black females (54.6 percent) were victims. Tables III-2 and III-3 show that 36 percent of the victims were males between 46 years and over. However, most male victims (approximately 52%) were between 19 and 45 years of age. From the data available it appears that most reported victims are white males between 19 and 60 or females between 19 and 45. For the purposes of analysis, the rather extensive crime code with 50 categories of the Bridgeport Police Department was reduced into 19 major codes. The 19 categories of crime are : murder/ manslaughter; robbery, assault, burglary, larceny less than \$50, larceny greater than \$50, motor vehicle theft, disorderly conduct, family offenses, property damage, nuisances, alarms, motor vehicle violations, medical problems, miscellaneous-juvenile, other person (includes forgery, kidnapping, bribery, license violations, code violations, court violations, impeding justice, escapes, suicides, dead on arrival, missing persons, found persons, suspicion, and warrant-subpoena), other-property (includes other larceny, stolen property, weapons, arson, property investigations, and accidents), and vice (includes sex offenses, obscenity, drugs, liquor laws, drunkedness, and accidents).

Tables III-4 through III-6 show the types of crimes with which the various demographic groups were involved. The results of analyses using these 19 categories and the sex, age, and race were: higher percentages of males were victims of robberies (69.6 percent), burglaries (52.2 percent), larcenies greater than \$50 (72.1 percent), motor vehicle thefts (71.1 percent), property damage (59.3 percent), alarms (68.0 percent), and motor vehicle violations (70.8 percent). Females tended to be victims of assaults (54.4 percent), larcenies less than \$50 (52.0 percent), disorderly conducts (56.4 percent), family offenses (68.0 percent), and miscellaneous-juvenile (58.1 percent). Higher percentages of whites were victims of larcenies greater than \$50 and property damage. For the age group 19 to 30, assaults (46.2 percent) and larcenies greater than \$50 (51.2 percent)

were most prevalent; for the group 31 to 45, family offenses was the largest category of crime. As might be expected, medical problems was the largest category for the age group 61 years and over. From the previous findings, it can be seen that certain demographic groups were more likely to have crimes committed against them and that some crimes are associated with certain groups. This information may prove useful for police management in structuring crime prevention programs and crime suppression activities.

B. Location of Incident

The Bridgeport Department of Police divided the city into car post sectors - areas of patrol responsibility. The four major sectors are the red, green, amber, and blue sectors. The other sectors had relatively few cases, so the discussion will center on the four major areas.

Tables III-7 through III-9 represent the sectors where different demographic groups were involved in incidents. A higher percentage of black victims (49.1 percent) were found in the red sector. The highest percentage of whites were victimized in the amber district. For age groups, a higher percentage of 31 years through 45 years (31.5 percent) were victimized in the red sector. A higher percentage of 19 years to 30 years (36.6 percent) were victims in the amber district.

Some types of crimes appear to take place more frequently in certain sectors. Table III-10 shows that 21.8 percent of family offenses, 23.7 percent of mobberies, and 23.6 percent of burglaries occurred in the red sector; 23.5 percent of family offenses took place in the blue district.

Table III-11 represents the association between crime codes and high frequency grids (streets in Bridgeport where the number of incidents was 50 or greater). Grid 003 had a high percentage of family offenses (23.8 percent), and Grid 009 had a higher percentage of larcenies less than \$50 (38.6 percent). Medical problems occurred frequently in Grid 008A (37.9 percent).

Within the general locations of sector and grid, there are certain location types - house, apartment, store, etc. where crimes or incidents appear more likely to occur. For example, Table III-12 shows that more females (60.3 percent) were victimized in apartments. A higher percentage of males, 62.0 percent, were victimized on the streets. Table III-13 indicates that higher percentages of blacks (53.7 percent) and Pureto Ricans (49.7 percent) were victims of crimes committed in apartments. Higher percentages of whites were involved in incidents in houses (18.1 percent) and private automobiles (23.7 percent).

Table III-14 represents the association between the specific incident locations and the general sectors of the city. The conclusions that can be drawn from this table are somewhat tenuous given the small number of cases in each cell, however, the analysis indicates what information can be derived from a data base. It appears that within the red sector, incidents occur more frequently in hospitals, gas stations, sports arenas, motorcycles, sidewalks, and streets. The higher percentages of crimes in the blue sector occurred in liquor stores, theaters, churches, and commercial trucks. In the amber sector, hotels, and motels, drive-ins and banks had the higher percentages of crime occurrence.

The ability to pinpoint where certain crimes are happening frequently is important to the Police Department. While random patrol may have certain

value, police units or patrolmen cannot be everywhere. With limited resources and shrinking budgets, assignments to hich activity locations becomes more feasible. It would benefit the Bridgeport Police Department to know in detail using valid and comprehensive information where incidents are likely to occur.

C. Time of Incident

In addition to location, it is equally important to know when crimes occurred so that patrol scheduling can be completed more effectively. The two major categories used in the subsequent analysis of time are: hours of the day (am-pm, 3pm - 2am/3am - 2pm) and month of the year.

An analysis of when crimes occurred in Bridgeport in 1975 shows that the majority of incidents during the entire year occurred during the night time hours. Tables III-15 through III-17 indicate that a clear majority of incidents for each month occurred in the pm period (12:00 noon to 11:59 pm). August and December show slightly higher percentages of crimes (76.7 percent and 72.7 percent respectively) occurring in the pm period. Breaking time into two different categories (3 pm to 2:59 am and 3 am to 2:59 pm), it can be observed that the majority of incidents occurred in the late evening period. This pattern was consistent throughput the analysis. Since the majority of crimes occur late at night or early morning, it would appear to be advantageous to schedule patrol resources so that more personnel are available during these hours.

Tables III-18 through III-20 show that different crime types occurred at different times. Very high percentages of robberies (87.2 percent), and larcenies less than \$50 (78.9 percent) occurred in the PM period. Robberies were very frequent during the 3 pm to 2:59 am period. Nuisances had a slightly higher percentage (49.2 percent) than other categories in the 3am to 2:59 pm period. The analysis of specific hours for different crime types (Table III-20) reveals a higher percentage, (15.0 percent) for robberies at 2100.

Table III-21 reveals that in August, 15.0 percent of larcenies less than \$50 occurred, and in November 15.3 percent of the robberies took place.

D. Time and Location of Incident

Tables III-22 through III-24 indicate when (time) incidents were more likely to occur in certain locations around the city. For the major sectors of the City of Bridgeport (red, green, blue, amber), there was little variation; the majority of incidents occurred in the PM period. However for more specific car post sectors, tactical squad and claims car, higher percentages of incidents (41.8 percent and 41.7 percent respectively) occurred in the AM period. The grid locations (streets where the frequency of crime incidents was 50 or more), show that generally crime incidents took place in the 3pm to 2:59 am period. However, in Grid 015 and Grid 020, higher percentages (53.8 percent and 70.3 percent respectively) of crimes did take place in the 3am to 2:59 pm period. Specific incident location types show more vairation than the general locations of sector and grid. Restaurants (47.6 percent), businesses (54.2 percent) had higher percentages in the AM period than other categories.

Tables III-25 through III-27 shows the months when incidents occurred in the City. Although the major sectors (red, green, blue, amber) indicate little variation, several of the specific car post sectors did have some variation. Car post sectors with higher percentages than others included:

police sergeant (April, 22.2 percent), special services (June, 22.6 percent), park police (February and July, 28.6 percent), Board of Education (February and April 22.2 percent), and K-9 Corps (October, 27.5 percent). A few grid locations showed some variation. Slightly higher percentages of crimes occurred in Grid 026 (April, 16.7 percent), Grid 027 (July, 15.6 percent), and Grid 015 (October, 21.3 percent).

The specific incident location types also show some variation. Unfortuneately, the percentages are based on very small cell values, therefore, the conclusions are tenuous. During the early months of the year, particularly January, certain incident location types had higher percentages of crime incidents: liquor store (40.0 percent), sports rena (33.3 percent), theater 66.7 percent), private truck (35.7 percent). Sports arena and theaters had slightly higher percentages of crime incidents, 33.3 percent in April, as did private motorcycles, 60.0 percent. In June, churches (42.9 percent) and commercial taxis (50.0 percent) had higher percentages of incidents. The lack of data in these categories prohibits any firm conclusions concerning specific location types, but the analyses indicate the value of such information especially when they are related to time periods.

E. Reporting

One problem that police officers face when trying to solve crimes is the length of time an incident goes unreported. The longer an incident remains unreported, the more difficult it will be for an officer to attempt to find witnesses and evidence. In Bridgeport, the information suggests (Table III-28) that incidents are being reported without delay, but that responses by the departments are not always documented in the automated information file.



IMPACT - EFFECTIVENESS AND EXTENDED ANALYSIS

D

 t_{c}

 \mathcal{O}

IV



CHAPTER IV

IMPACT-EFFECTIVENESS AND EXTENDED ANALYSIS

The availability of response time and clearance information on the Bridgeport computer files has precluded a comprehensive approach to evaluation. However, to demonstrate what procedures and methods could be used if the data were available, certain analysis were performed. The results are presented in Tables IV-1 to IV-7D in Appendix E.

A. Response Time

Documentation of response time data and the clearance status of incidents are some prerequisites for the performance evaluation of police services. Consistent storage and retrieval methods for this type of data permits analyses to focus on the central concerns of how crimes and non-crime calls for service are actually cleared. An evaluation of this kind provides information that can be used to assess the level of effectiveness of police operations so that planning or overall management can be made more rational.

Table IV-1 presents an example of the approach and type of information which could be made available for decision-making if the elements were present in the information system. For example, Table IV-1 illustrates the comparison of response time with crime-type. This analytical technique provides the capability for close examination of the response time characteristics specific to types of crime. For instance, Table IV-1 indicates that 82 percent of the 204 incidents (of the 8,019 total) reported are responded to in one to six minutes. The distribution of response times among specific types of crime - e.g., robbery, burglary, larceny, etc., can also be determined. Coupled with data on officers dispatched and clearance status of incidents serviced (data not presently computerized by the Police Department), information of this sort would provide police management with a timely decisiontechnology to thoroughly evaluate an important element of the impact or effectiveness of police services. It would also provide a planning mechanism to permit more precise allocation of men and resources to high-incident locations to achieve the maximum impact on the prevention and apprehension facets of crime control.

B. Extended Analysis

In depth analysis of the patterns or relationships identified between crime code, location, and time variables reveal identifiable consistencies which can be used in planning patrol operations or deterrence strategies. For example, the relationship between crimetype and location of incidents throughout the city by hourly patterns can provide police management with valuable information about factors which may aid in scheduling pre-emptive patrol operations. Likewise, studying the relationship between dominant crime types and major car post sectors and/or grid locations can assist police management in devising improved strategies for handling high frequency and recurrent crime types by concentrating patrols in specific locations of the city.

To demonstrate this type of analysis, four specific relationships have been established in depth: (1) car post sectors by crime type by 12 hour segments - 3pm to 2:59 am, 3am to 2:59 pm; (2) major crime types by selected grids by time of day; (3) major crime types by car post sectors by seasons/trends; and (4) major crime types by location

status by car post sectors. As previously mentioned, large amounts of missing data inherent with the 1975 sample preclude a definitive interpretation of relationships identified. However, they do indicate that such analyses can provide practical and realistic assessments of incident patterns in the city.

Tables IV-2 and IV-3 are presented for purposes of describing the manner in which two major location variables in the Bridgeport 1975 data sample (car post sector and grid) are collapsed to provide a general explanation of location and crime type characteristics.

The four major categories of car post sectors i.e. red, green, blue, and amber included a total of 5322 cases or approximately 66 percent of a total sample of 8017 cases. Table IV-2 presents the percentage distribution of specific grid locations which contained 150 or more cases.

<u>C.</u> <u>Major Car Post Sectors By</u> Crime Type By Twelve Hour Segments

Tables IV-4A and IV-4B present the analysis of the four major car post sectors by crime type controlling for the two 12 hour segments -3pm to 2:59 am and 3am to 2:59 pm respectively.

The predominant crime types serviced by the red car post sector during the 3pm to 2:59 am time segment are motor vehicle theft (14.7 percent), alarms (12.2 percent), family offenses (7.1 percent), disorderly conduct (6.4 percent), and miscellaneous juvenile (9.9 percent). In the green sector, the predominant crime types serviced during the same time segments of the day tend to be assault (7.2 percent), motor vehicle theft (9.1 percent), property damage (9.1 percent), nuisances (6.1 percent), alarms (9.4 percent), and miscellaneous juvenile (10.8 percent). The blue sector's general crime pattern during the same hours

consists of motor vehicle theft (10.5 percent), disorderly conduct (7.8 percent), property damage (9.0 percent), alarms (9.3 percent), miscellaneous juvenile (14.2 percent). Crime types such as assault (5.8 percent), larceny less than \$50 (7.3 percent), larceny greater than \$50 (6.0 percent), motor vehicle theft (8.6 percent) are frequent in the amber car post sector during the hours of 3pm to 2:59 am.

The green car post sector tends to be the most active in terms of calls for service; 651 cases or approximately 36.0 percent of the reported cases between the hours of 3 pm to 2:59 am. The major Part I crime across all sectors during these hours tends to be motor vehicle theft. Alarm and miscellaneous juvenile crime calls also occur frequently during this period in the four car post sectors. Assault and property damage tend to comprise major percentages of the calls serviced in the blue and red sectors during 3pm to 2:59 am. In the amber car post sector, crime types such as assault, larceny, less than \$50, larceny greater than \$50, disorderly conduct and property damage comprise major percentages of the various crime calls serviced by the police during the same hours.

Table IV-4B presents the analysis of the four major car post sectors by crime type controlling for the hours of the day - 3am to 2:59 pm. The activity across the four car post sectors during these hours of the day is somewhat less than the 3pm to 2:59 am time segment. However, the green car post sector tends to be the most active in terms of calls for service with a total of 408 cases or approximately 36.4 percent of the total 1122 calls serviced by the four car post sectors during this primarily day time segment.

The predominant crime types serviced by the red car post sector during this day time segment are burglary (8.8 percent), motor vehicle theft (20.4 percent), alarms (13.3 percent), and miscellaneous juvenile

(8.8 percent). The major crime classification in the green car post sector during the same hours tend to be assault (5.9 percent), burglary (8.8 percent), larceny less than \$50 (5.6 percent), larceny greater than \$50 (6.4 percent), motor vehicle theft (11.0 percent), property damage (7.8 percent), nuisances (7.6 percent), alarms (11.3 percent) and miscellaneous juvenile (10.5 percent). The blue car post sector has frequent crime classifications such as burglary (5.5 percent) larceny greater than \$50 (6.7 percent), motor vehicle theft (17.6 percent), property damage (6.3 percent), nuisances (5.9 percent), alarms (11.3 percent), and miscellaneous juvenile (13.9 percent) during this particular time segment. Crime types such as burglary (6.1 percent), larceny less than \$50 (5.8 percent), larceny greater than \$50 (8.5 percent), motor vehicle theft (9.2 percent), property damage (9.5 percent), alarms (14.2 percent), and miscellaneous juvenile (12.2 percent) tend to be the major classifications that are serviced by the amber car post sector during the 3am to 2:59 pm time segment.

Motor vehicle theft again tends to be the most dominant Part I offense serviced across all four major car post sectors during this particular time segment. However, there is an increased tendency in the amount of burglaries that are serviced by the four car post sectors during these hours of the day as opposed to the 3pm to 2:59 am time segment. Again, larcenies and property damage crime classifications tend to be other major offensives serviced by the green and amber car post sectors.

D. <u>Major Crime Types by Selected</u> Grids Controlling Time of the Day

Table IV-5A presents the analysis of major crime types across major grid categories (see Table IV-3) controlling for the 3pm to 2:59 am time segment. Grid 008a tends to be the most active in terms of servicing calls during these hours of the day. In addition, Grid 003 has major crimes occurring such as assault (12.9 percent), motor vehicle theft (14.9 percent), family offenses (10.9 percent), and medical problems (22.8 percent) during the 3pm to 2:59 am time segment. In Grid 009 crimes such as assault (7.5 percent), larceny less than \$50 (13.6 percent), larceny greater than \$50 (8.0 percent), alarms (8.0 percent), and medical problems (18.1 percent) tend to be major classifications serviced during this evening time period. In Grid Oll there is a tendency for motor vehicle theft (7.1 percent), disorderly conduct (9.2 percent), property damage (7.1 percent), alarms (16.3 percent) to be the major crime type classification. Larceny less than \$50 (20.6 percent) and medical problems (10.8 percent) are found to be major service activities in Grid 016 during the 3pm to 2:59 am time segment. In Grid 017, assauly (8.4 percent), medical problems (9.6 percent), and miscellaneous juvenile (10.2 percent) are predominate categories. Major crime type activity in Grid 008a consists of assault (10.0 percent), medical problems (33.7 percent), and miscellaneous juvenile (14.6 percent).

The data gathered by relating crime type to major grid locations by time segments of the day can be used by police management to supplement the information derived from associating crime type to major car post sectors by time segments of the day. By using grid location data, police management is able to narrow the gross location information of car post sectors into narrower sectors throught the city. Such analysis can support police management in assessing the present patrol status and developing more precise strategies for handling crimes by location during specific time segments of the day.

Talbe IV-5B is similar to the analysis presented in Table IV-5A except that the time segment 3am to 2:59 pm is examined in this comparison. Burglary offenses tend to be predominate in Grid 003 (11.5 percent) and Grid 011 (11.1 percent) during this time period. Larceny less than \$50 is a major category for Grid 009 (13.3 percent), and Grid 016 (12.7 percent). Whereas motor vehicle theft is predominant in Grid 003 (31.4 percent), Grid 017 (16 percent), and Grid 008A (13.7 percent) during the 3am to 2:59 pm time segment. Medical problems comprise a high percentage of service activity in Grid 003 (28.6 percent), Grid 009 (17.7 percent), Grid 017 (13.3 percent), and Grid 008A (36.6 percent).

E. <u>Major Crime Type by Major Car Post Sectors</u> Controlling for Seasonal Tracts

Other interesting relationships between crime type and location can be examined in terms of seasonal trends. Table IV-6A, 6B, 6C and 6D present the analysis of crime type by major car post sector controlling for seasonal trends. It should be noted that the number of cases and percentages of crime type across major car post sectors is based upon the relative distribution of crime codes within each major car post sector. Hence, the number of burglaries in the red sector during the winter season (8.8 percent) is interpreted to mean that they comprised 8.8 percent of a total of 205 calls included in the red sector during the winter season.

Burglary offenses are major crime classifications in the red sector (8.8 percent), green (6.9 percent), amber sector (4.9 percent), and blue sector (8.2 percent) during the winter season. Motor vehicle theft comprises major percentages across all four major car post sectors in the winter months; red (17.6 percent), green (13.3 percent), blue (14.7 percent), and amber (12.8 percent). Property damage is another crime classification which has high percentages across the car post sectors; green (10.8 percent), amber (11.9 percent), and blue (10. 9 percent). Alarm calls tend to be quite high across all four car post sectors (see Table IV-6A) during the winter season.

Table IV-6B presents the analysis of crime type by the four major car post sectors controlling for the spring season. Again, motor vehicle theft is a major crime type classification across all four ca. post sectors; red (13.9 percent), green (9.6 percent), blue (11.8 percent), and amber (9.7 percent). Property damage offenses tend to comprise high percentages in the following three car post sectors; green (11.6 percent), blue (7.9 percent), and amber (9.7 percent). Alarms and miscellaneous juvenile crime classifications likewise comprise high percentage distributions across the four major car post sectors. (See Table IV-6B for a more detailed presentation of the associtaion between crime type and the four major car post sectors controlling for the spring season).

Crime type trends in the summer months depict increases in crimes such as assault, burglary, larceny less the \$50, larceny greater than \$50, disorderly conduct, alarms, and miscellaneous juvenile across the four major car post sectors. Table IV-6C presents the analysis of crime type by major car post sectors controlling for the summer months. (See Table IV-6C for a detailed presentation of the breakdown between crime classifications by the four major car post sectors for the summer months).

During the fall months, burglary, larcenies, motor vehicle thefts, disorderly conduct, property damage, alarms, and miscellaneous juvenile tend to be major crime categories across the four car post sectors. (See Table IV-6D for a more complete presentation of the association between crime type and car post sectors controlling for the fall months).

In summary, assault, larceny, disorderly conduct offenses tend to be predominate in the summer and fall months across the four major car post sectors. Miscellaneous juvenile offenses are predominate across the four car post sectors during the spring, summer, and fall months. Whereas motor vehicle theft, property damage and alarms are found to be quite consistent across the four major car post sectors controlling for the four seasons of the year.

F. <u>Crime Type By Location Status</u> Controlling for the Major Car Post Sectors

Another important relationship between crime type and major car post sectors pertains to the relative location status of the offense. That is, what crimes in what car post sector are primarily associated with residential areas, businesses, public areas, etc. Table IV-7A, 7B, 7C, and 7D present the analysis of crime type by location status controlling for the four major car post sectors.

Table IV-7a presents the analysis of crime type by location status controlling for the red car post sector. Of the 109 incidents associated with residential premises, burglaries comprise (10.1 percent), disorderly conduct (7.3 percent), family offenses (12.8 percent), medical problems (8.3 percent), and miscellaneous juvenile (17.4 percent). The major crime type associated with businesses in the red car post sector are burglary (16.4 percent), property damage (9.8 percent), and alarms

Assaults (20.0 percent) and motor vehicle thefts (20.0 percent) are the two major crime classifications associated with public areas in the red car post sector. Whereas motor vehicle theft, larceny greater than \$50 and motor vehicle violations tend to be associated with vehicles.

Table IV-7b presents the same association as presented in IV-7a except that the green car post sector is controlled sector. Of the 230 incidents commited against residential premises in the green car post sector, burglary compises (12.6 percent), family offenses (9.1 percent), property damage (9.1 percent), and miscellaneous juvenile (15.2 percent). The predominate crime classifications against businesses in the green sector are burglary, (12.7 percent), larceny less than \$50, (9.5 percent), property damage, (12.7 percent), and alarms (25.4 percent). Major crime categories associated with public areas are: assault (14.1 percent), motor vehicle juvenile (12.9 percent). Also note that property damage classification comprises 17.0 percent of the incidents against vehicles in the green car post sector.

Table IV-7C continues the analysis between crime type and location status of the call controlling for the amber car post sector. The more frequent crimes directed against residential premises in the amber car post sector are assault (8.5 percent), burglary (7.6 percent), family offenses (11.9 percent), property damage (10.2 percent), miscellaneous juvenile (13.6 percent), and other-person (10.2 percent). Burglary (9.1 percent), disorderly conduct (8.1 percent), property damage (9.1 percent), alarms (28.3 percent), and miscellaneous juvenile (16.2 percent) are the major crime categories related to businesses in the amber car post sector. Frequent public area offenses include assault (10.1 percent), larceny less than \$50 (10.1 percent), larceny greater than

\$50 (10.1 percent), motor vehicle theft (10.1 percent), and miscellaneous juvenile (10.1 percent). It is interesting to note that larceny greater than \$50 (10.2 percent) and property damage (18.2 percent) are crimes which tend to be directed against vehicles.

The blue car post sector is the controlling variable in Table IV-7D which presents the analysis between crime type and location status of the offense. Burglaries (11.2 percent), larcenies greater than \$50 (9.9 percent), disorderly conduct (9.9 percent), family offenses (15.1 percent), property damage (9.9 percent), and miscellaneous juvenile are the major crime codes associated with residential premises. The major crime codes associated with businesses in the blue car post sector are: burglary (18.4 percent), property damage (10.3 percent), alarms (27.6 percent), and miscellaneous juvenile (13.8 percent). Motor vehicle theft, larceny less than \$50, miscellaneous juvenile represent the major crime codes associated to public areas. And again, larceny greater than \$50 and property damage represent measurable percentages of the crimes associated with vehicle-type offenses.

G. Concluding Remarks

A large amount of missing data precluded the use of extensive and conclusive effectiveness analysis. This problem greatly restricted definitive interpretations or precise operational based recommendations. For example, lack of available information describing the number of officers or units dispatched, the clearance status or actions taken for the incidents, and the lack of information on response times prohibited efforts to provide a more comprehensive appraisal of the relative performance of the police department. Chapter IV presents a limited form of analysis that can be used to conduct performance analysis with the information available.

Extended indepth analysis can provide the capability for further examining important relationships identified in less involved analysis. It can also provide more precise information that can be used to derive more specific impact or effectiveness assessments. Statistical analysis can also provide a valid basis for devising scheduling strategies for men and resources in high incident locations and times.

SUMMARY AND MAJOR FINDINGS

Ŷ

 \hat{O}

V

9

đ


CHAPTER V

SUMMARY OF MAJOR FINDINGS

A. General Overview

The large amount of missing data encountered in the 1975 Bridgeport Police sample file has precluded a comprehensive and extended analysis and evaluation of the Bridgeport Police Department. While the univariate and bivariate analyses identify patterns of crime and overall demand, location, and time characteristics of calls for service occurring in the City, they are not sufficiently reliable to permit conclusive statements about the Bureau of Police. Thus, the primary emphases of the report were: document the information system issue, examine dominant crime-type, location, time characteristics of incidents, and demonstrate a general methodology for future performance analyses of police operations.

B. Frequency Analysis

The questionable nature of many of the situational variables limited the one variable analysis to demand variables (crime codes), location (car post sector, grid, incident location type), time (time incident occurred or reported, date incident occurred/reported) and victim characteristic variables (sex, age, race). It was found that the major demands for police service in the City tend to be medical problems (10.3 percent), motor vehicle theft (10.2 percent), nuisances (10.1 percent), property damage (7.5 percent), and various alarms (6.9 percent). Major Part I crime categories were burglary (4.9 percent), larceny less than \$50 (4.2 percent), larceny greater than \$50 (5.5 percent), and assault (4.6 percent). Other offenses such as disorderly conduct (4.3 percent), motor whicle violations (4.2 percent), and family offenses (3.0 percent) comprised major incidents of service demands for the police.

The four major car post sectors green, amber, blue, and red accounted for a majority of the calls for service in the City. The green car post sector was the most active with 23.8 percent of the calls, amber 16.4 percent, blue 15.3 percent, and red 10.9 percent. A location grid corresponding to police maps was created from the information presented in the sample file and it contained street locations having 50 plus incidents. Nine of the high frequency grids computed from the sample are identified in the following table. These nine grids account for approximately 76 percent of the incidents identified by street location.

Grid	Number	Percent
003	152	6.3%
008A	395	16.5
009	345	14.4
017	246	10.3
011	174	7.3
016	168	7,0
010	130	5.4
027	129	5.4
008B	100	4.2

Totals

1839

76.0%

Other specific location type information disclosed that apartments account for approximately 26 percent of the location codes. Houses account for another 12 percent, private autos (19.6 percent), businesses (8 percent), and streets (6 percent) of specific location type information.

Victim related information such as age, sex, and race provide some information but because of its low number of valid cases, the following findings are somewhat tenuous. For instance, in the sample file, it appeared that most victims were between 19 and 45 years of age (56 percent) while victims between the ages of 46 through 60 account for 19 percent. People aged 61 and over are victims in about 13 percent of the incidents. Approximately 56 percent of the victims are male, while 44 percent are female. Approximately 64 percent of the victims are white, 26 percent black, and 10 percent are Puerto Rican.

Related to time of occurrence, there is a slight increase from May through August, but no evident pattern for day of the month. However, in regard to the two 12 hour segments AM and PM it was found that there are approximately twice as many incidents occurring in the PM as in the AM. That is, 65.5 percent of the incidents occur in the PM hours while 34.4 percent of the incidents are reported to occur in the AM hours of the day. In addition, when the 24 hours of the day are collapsed into two other 12 hour segments—3pm to 2:59 am and 3am to 2:59 pm, the 3pm to 2:59 am time period contains approximately 61.5 percent of the incidents, while the 3 am to 2:59 pm period contained approximately 38.5 percent of the incidents in the sample. In terms of specific hours of the day, there tends to be a slight increase in calls for service around 1600 hours (4pm) with a peak period around 2100 hours (9pm).

C. Impact-Effectiveness and Extended Analysis

A major prerequisite for conducting performance analysis is the capability to document and maintain complete data concerning response times and the resulting clearance actions taken for incidents. Such information is necessary for the assessment of the impact of police service on the problem as well as an assessment of the effectiveness of the police in their response to demands for service. The lack of proper and complete data in the Bridgeport sample i.e. response time, clearance information, precluded a thorough and meaningful analysis of the impact and operational effectiveness of the Bridgeport police activity for 1975. However, some assessments and recommentations bearing upon the information were made.

Some indepth analysis of existing information showed that the activity across the four major car post sectors is greater during early evening, evening, and early morning time segment. The green sector tends to be the most active during both day and evening time segments. Of the Part I crimes, motor vehicle theft is found to be a major category across the four car post sectors during both time segments. Burglaries comprise a somewhat higher percentage distribution in the four car post sectors during the 3 am to 2:59 pm time segment. Person related crimes such as assault, disorderly conduct, comprise higher percentages in the four car post sectors during the 3 pm to 2:59 am time segment. Alarms of various types (fire, holdup, false) also make up a large percentage of the service calls in the four car post sectors during the 3 pm to 2:59 am period.

A

Another section of the indepth analysis focused upon crime code and selected grid locations controlling for the two time segments-3 pm to 2:59 am and 3 am to 2:59 pm. Medical problems tended to be the major

service activity performed by the police during the 3 pm to 2:59 am period. The following brief table presents grid location and major crime categories during the 3 pm to 2:59 am time segment:

Grid	Incident Type
003	Assault, motor vehicle theft and medical problems
009	Larceny less than \$50 and medical problems
011	Disorderly conduct and alarms
016	Larceny less than \$50 and medical problems
017	Motor vehicle theft, medical problems and miscellaneous juvenile
008A	Assault, medical problems and miscellaneous juvenile

During the 3 am to 2:59 pm time segment, burglary is a major offense associated with Grid 003 and Grid Oll. Larceny less than \$50 is common in Grids 009 and 016. The major Part I offense, motor vehicle theft, comprises a high frequency of occurrence in Grid 003, Grid 017, and Grid 008A. Medical problems is the major incident associated with Grids 003, 009, 017, and 008A during the 3 am to 2:59 pm time period.

D. <u>Extended Analysis of Crime Code by Location</u> <u>Controlling for Major Car Post Sectors</u>

Another finding of interest is the relationship between crime code and location status controlling for the four major car post sectors.

The following is a breakdown between location status and frequent crime codes for each of the four car post sectors:

Red Sector	Incident Type
Residential	Burglary, family offenses, medical problems, and miscellaneous juvenile
Business	Burglary, property damage and alarms
Public Areas	Assaults and motor vehicle thefts

Green Sector	Incident Type
Residential	Burglary, family offenses, property damage, and miscellaneous juvenile
Business	Burglary, larceny less than \$50, property damage and alarms
Public Areas	Assault, motor vehicle theft, nuisances, medical problems, and miscellaneous juvenile offenses
Amber Sector	Incident Type
Residential	Assault, burglary, family offenses, property damage, and miscellaneous juvenile
Business	Burglary, property damage, alarms, and miscellaneous juvenile
Public Areas	Assault, larcenies, motor vehicle theft, and miscellaneous juvenile
Blue Sector	Incident Type
Residential	Burglaries, larceny greater than \$50, disorderly conduct, family offenses, property damage, and miscellaneous juvenile
Business	Burglary, property damage, alarms, and miscellaneous juvenile
Public Areas	Motor vehicle theft, larceny less than \$50, miscellaneous juvenile

Some definite patterns have been found in the existing data which document the demand for specific police responses. However, improving the overall management of the information system can increase the quality of data available for this type of analysis and provide the capability for improving operational police performance.

PRODUCTIVITY IMPROVEMENT

VI

0

62

RECOMMENDATIONS

S.J



PRODUCTIVITY IMPROVEMENT RECOMMENDATIONS

VI

The following recommendations to improve productivity in the Bridgeport Police Department are based upon the analyses and results presented in Chapters II thru V. It is intended that they be reviewed and examined by the management and supervisory levels of the Bridgeport Police Department. The examination should result in an integration of experience i.e., police knowledge with the analytical findings presented in this report. These productivity recommendations are not designed to be implemented separately, although this strategy is a feasible alternative; rather they are intended to represent a wholistic approach that captures the management, information, patrol, and reporting functions of the Bridgeport Police Department.

A. Information System

Lack of and/or improperly coded data on the computer file was a major constraint in the use of the data for the productivity analysis. The data issue was central enough to preclude any meaningful impact/ effectiveness analysis; therefore, only a general methodology for assessment with limited data was presented. In conjunction with the major study findings, the overriding recommendation is to improve the system by which incident information is collected, coded, and transposed onto the records system. This capability can provide for valid, reliable productivity analysis and systematic analysis to improve daily operations. This process may be initiated by instituting the following changes:

- 1. Establish a priority system for ranking the most critical and necessary information required by Bridgeport Police Management.
- 2. Improve the information gathering and documentation system by supervising the complete collection code from report writing to entry on the computer records system. Emphasise the importance of complete and accurate reporting by patrolmen, especially suspect and witness information. This step may include periodic feedback to the patrolmen on how the information is being used.
- 3. Begin a formal auditing procedure to periodically review the data that is being entered from the incident file reports to the computer records file. No error in critical or necessary data reporting can be accepted.
- 4. More relevant descriptive information can be gathered by patrolmen. Information such as the number of officers and/or units responding to the calls and the resulting clearance status of the incident should be reported. This type of data should be maintained on file to enable police management to assess the operational effectiveness of police services when required.
- 5. Explore the possibility of procuring a flexible management software system capable of statistical analysis to adapt to the existing IBM hardware. The analysis in this report utilized basic elements of the SPSS system (Statistical Package for the Social Services). This system is designed to allow for relatively easy manipulation of data for statistical analysis by line personnel who would require minimal training in computer programming (estimate ten hours).

- 6. Use the expanded improved data base to periodically conduct impact and effectiveness analyses of police operations in responding to demands for service. Use the results of these analyses to improve plans for crime prevention in high incidence locations.
- 7. Continuously monitor and update the information system to keep current with changing operational demands and provide feedback to all levels of management on existing capabilities.

B. Patrol Activities

Employ the crime-type and locational patterns determined in this report for review of current patrol strategies.

- Patrol assignments should be reviewed with an emphasis on scheduling the most frequent patrols in the grid area with the highest frequency of incidents — i.e., Grids 008A, 009, 017, 011, 016, 003, 010, 027, 008B.
- 2. Calls for service increase between the hours of 3pm to 2:59 am (e.g. twice as many incidents occur during these 12 hours). Scheduling manpower resources should provide for increased patrols between 3pm and 2:59 am.
- 3. Motor vehicle theft is a major offense in all sectors of the city of Bridgeport; therefore, police management should explore the possibilities of highlighting high theft areas in the news, emphasizing common sense procedures such as locking vehicles, televising educational strategies for the community, publicizing preventative techniques, and improving information dissemination when a theft occurs.

4. The Bridgeport Police should establish periodic performance standards based upon motor vehicle thefts per area and/or dollar losses per time period to provide a monitoring mechanism for thefts.

C. Organizational Realignment

The Bridgeport Police Department may want to examine the feasibility of transferring the administration of the ambulance service from the Police Department to the Fire Department. Presently, the ERS — Emergency Reporting System — is housed in the Fire Department and it would appear to be expedient if management could consolidate the service. The Police Department would be relieved of an administrative burden which presently accounts for 10.3 percent of the demand for police service, thus freeing this time for meeting more operational requirements. The available information indicates that many of the calls for medical service are from the older citizens of Bridgeport and may not be directly crime related. However, more complete data may indicate that medical problems are associated with assaults or other crimes then the issue of ambulance placement is problematic.

APPENDIX A

۲

 \mathbb{C}

 $\langle \rangle$

(y,

DETAILED CRIME CODE CLASSIFICATION BRIDGEPORT POLICE DEPARTMENT

 $\hat{\mathcal{L}}$

õ

ß

(F):

8

 $\tilde{\mathcal{O}}$ is

æ



HOMICIDE

0101	Murder	0408	Student-Teacher Assault
0102	Misconduct With a Motor Vehicle	0409	Simple Assault
0103	Manslaughter	0410	Threatening
0104	Criminally Negligent Homicide-	0411	Reckless Endangerment
	Non-Auto	0412	Fight
0105	Killing of a Police Officer	0413	Fight-Juvenile
		0414	Bomb Scare
SEE A	LSO	0415	Attempted Assault
0401	Assault with Intent to Kill	BURGL	ARY
0402	Assault with Intent to Commit		
	Murder	0501	Residence-Forcible
		0502	Residence-Non-Forcible
RAPE		0503	Apartment—Forcible
		0504	Apartment-Non-Forcible
0201	Rape-Forcible	0505	Hotel-Forcible
0202	Rape-Without Consent	0506	Hotel-Non-Forcible
0203	Rape-Statutory-Under 14 Years	0507	Commercial—Forsible
	of Age	0508	Commercial-Non-Forcible
0204	Rane-Statutory-Under 16 Years	0500	School College Univers
0204	of App	0510	School College, Univers
0205	Bane-Without Force	0010	Forgible
0205	Attempted Rape	0511	Other-Foreible
0200	Accempted Rape	0512	Other-Non-Foraible
DUBBE	יסע	0513	With Wielense
0201	Rabberry Commondal - Fixeerry	0515	Nothing Taken
0301	Robbery-Commercial Other	0515	Attomat
0302	Nobbery-Commercial-Other	0515	Allempt Manufacture or Responde
0202	Reberry Commenced No Manager	0310	Pumples le Teole
0303	Robbery-Commercial-No weapon		Burgtar's 1001s
0304	Robbery-Residence-firearm	TADOR	
0305	Robbery-Residence-Other weapon	LARCE	NI-NON-AUTO-LESS THAN S
0300	Robbery-Residence-No weapon	0001	
0307	Robbery-Street-Firearm	0601	There
0308	Robbery-Street-Other weapon	0602	
0309	Robbery-Street-No weapon	0603	Inert-Public Building
0310	Robbery-Attempted-Firearm	0604	Thert-From Motor Venic
0311	Robbery-Attempted-Other Weapon	0.00	Non-Auto
0312	Robbery-Attempted-No Weapon	0605	Theft-From Auto
0313	Purse Snatch with Violence	0606	Theft-Auto Accessories
	For Robberv of Taxi Cab use	0607	Theft-Bicycles
	Robberv-Street 0307, 0308,	0608	Theit-From Mail
	or 0309	0609	Theft—Purse Snatching
		0610	Theft-From Person-Pick
ASSAU	LT	0611	Theft-Animal
		0612	Theft-University Reside
0401	Assault With Intent to Kill		Building
0402	Assault With Intent to Commit	0613	Theft-University Academ
0402	Murdor		Building
0403	Accault With a Firearm	0614	Theft-From Hotel
0405	Ascault With a Dangerous	0615	Theft-Of Firearm
0404	Nosauri wich a Dangerous	0616	Theft-Attempted
0405	Negaulting a Police Officer	0617	Theft-Miscellaneous
0400	Tixoman on Correctional Officer,	0618	Theft-Checks or Negotia
01.06	fileman, or correctional UITICEr		Documents
0400	UNITO ASSAULT	0619	Theft-Credit Cards
0407	KISK OF INJURY to a Child		
	а. А. А. А		
	\mathcal{N}		and the second

52

ASSAULT (Continued)

0412	Fight
0413	Fight-Juvenile
0414	Bomb Scare
0415	Attempted Assault
BURGI	JARY
0501	Residence-Forcible
0502	Residence-Non-Forcible
0503	Apartment—Forcible
0504	Apartment-Non-Forcible
0505	Hotel-Forcible
0506	HotelNon-Forcible
0507	Commercial—Forcible
0508	CommercialNon-Forcible
0509	School, College, University-Forcible
0510	School, College, University-Non-
	Forcible
0511	Other-Forcible
0512	Other-Non-Forcible
0513	With Violence
0514	Nothing Taken
0515	Attempt
0516	Manufacture or Possession of
	Burglar's Tools
LARCE	NY-NON-AUTO-LESS THAN \$50
0601	Theft-Shoplifting
0602	Theft-Residence
0603	Theft—Public Building
0604	Theft-From Motor Vehicle-
	Non-Auto
0605	Theft-From Auto
0606	Theft—Auto Accessories
0607	Theft-Bicycles
0608	Theft-From Mail
0609	Theft—Purse Snatching
0610	Theft-From Person-Pickpocket
0611	Theft-Animal
0612	Theft-University Residence
	Building
0613	Theft-University Academic
	Building
0614	Theft-From Hotel
0615	Theft-Of Firearm
0616	Theft-Attempted
0617	Theft-Miscellaneous
0618	Theft-Checks or Negotiable
	Documents
0619	TheftCredit Cards

53

SEE ALSO-1004 Confidence-Flim-Flam

LARCENY----NON-AUTO---GREATER THAN \$50

- 0701 Theft Greater-Shoplifting
- 0702 Theft Greater-Residence
- 0703 Theft Greater-Public Building
- 0704 Theft Greater-From Motor Vehicle-Non-Auto
- 0705 Theft Greater-From Auto
- 0706 Theft Greater-Auto Accessories
- 0707 Theft Greater-Bicycle
- 0708 Theft Greater-From Mail
- 0709 Theft Greater—Purse Snatching 0710 Theft Greater—From Person—Pick-
- pocket 0711 Theft Greater—Animal
- 0712 Theft Greater-University Residence Building
- 0713 Theft Greater-University Academic Building
- 0714 Theft Greater-From Hotel
- 0715 Theft Greater-Of Firearm
- 0716 Theft Greater-Attempted
- 0717 Theft Greater-Miscellaneous
- 0718 Theft Greater-Checks, Etc.
- 0719 Theft Greater-Credit Cards

SEE ALSO-

- 1004 Confidence-Flim-Flam
- MOTOR VEHICLE THEFT
- 0801 Theft of Auto
- 0802 Recovered Auto
- 0803 Theft of License Plates
- 0804 Recovered License Plates
- 0805 Theft of Other Motor Vehicle
- 0806 Recovered Other Mo or Vehicle
- 0807 Operating Motor Vehicle Without Permission of Owner
- 0808 Theft of License, Registration or Title, Attempted Theft of Motor Vehicle or Motor Vehicle Documents

SEE ALSO-

3408 Abandoned Motor Vehicle Theft of Theft Greater for Theft from Auto or Other Motor Vehicle FORGERY AND COUNTERFEITING

- 0901 Forgery of Instruments Issued by Government
- 0902 Forgery of Public Records
- 0903 Forgery-Other (Checks, Motor Vehicle Documents)
- 0904 Criminal Simulation
- 0905 Forgery of Symbols of Value
- 0906 Using Slugs
- 0907 Possession or Use of Counterfeiting Apparatus
- SEE ALSO-

1008 Issuing Bad Checks

OTHER LARCENY OFFENSES

- 1001 Embezzlement
- 1002 Fraud
- 1003 Fraud with Credit Card
- 1004 Confidence-Flim-Flam
- 1005 Acquiring Property Lost, Mislaid, or Delivered by Mistake
- 1006 Extortion
- 1007 Theft of Services
- 1008 Issuing Bad Checks
- 1009 Criminal Impersonation-Non-Police Officer
- 1010 Misapplication of Property
- 1011 Theft of a Public Record
- 1012 Theft of a Secret

STOLEN PROPERTY

1101 Receiving Stolen Property

SEE ALSO-

4203, 4202 Recovered Property

WEAPONS

- 1201 Carrying a Concealed Weapon
- 1202 Possession of Prohibited or Unlicensed Weapon
- 1203 Illegal Possession of a Weapon
- 1204 Furnish Weapon to Minor
- 1205 Illegal Manufacture of Weapon
- 1206 Illegal Sale of Weapon
- 1207 Sale of Dangerous or Prohibited Weapon
- 1208 Use of Prohibited Weapon
- 1209 Carrying of BB Gun
- 1210 Use of BB Gun
- 1211 Carrying Weapon in Motor Vehicle
- 1212 Possession of Incendiary Device

COMMERCIALIZED VICE

1301 1302 1303 1304	Prostitution Patronizing a Prostitute Promoting Prostitution Permitting Prostitution
SEX C	FFENSES
1401	Sexual Misconduct—By Guardian
1402	Sexual Misconduct-Sodomy or Without Consent
1403	Deviate Sexual Intercourse Forcible or Without Consent
1404	Deviate Sexual Intercourse—Under 14 Years of Age
1405	Deviate Sexual Intercourse—Under 16 Years of Age
1406	Sexual Contact—Forcible or Without Consent
1407	Sexual Contact—Under 11 Years of Age
1408	Sexual Contact-Under 14 Years of Age
1409	Sexual Contact—Under 16 Years of Age
1410	Adultery
1411	Bigamy
1412	Public Indecency-Indecent
	Exposure
1413	Fornication
1414	Illegal Cohabitation
1415	Incest
1416	Homosexual Assault
1417	Homosexual Soliciting
1418	Voyeurism-Peeping Tom
1419	Sexual Contact by Deception
OBSCE	NITY
1501	Promotion of or Possession to Promote Obscene Material
1502	Production of Obscene Mail
1503	Presenting Indecent Performance
1504	Participating in Indecent
1505	Performance
1202	Sale OI UDSCENE Material
T200	UDSCENITY AS TO MINORS
T201	Disseminating indecent Comics
2203	For Obscene Phone Calls or Written Notes Use Harassment

2201 For Obscene Gestures or Language in a Public Place Use Breach of Peace

DRUGS

- 1601 Possession of Narcotics-General
- 1602 Sale of Narcotics-General
- 1603 Possession of Heroin
- 1604 Sale of Heroin
- 1605 Possession of Dangerous or Controlled Drugs
- 1606 Sale of Dangerous or Controlled Drugs
- 1607 Manufacture of Dangerous or Controlled Drugs
- 1608 Possession of Marijuana
- 1609 Sale of Marijuana
- 1610 Cultivation of Marijuana
- 1611 Possession of Hashish
- 1612 Sale of Hashish
- 1613 Manufacture of Hallucinogens
- 1614 Forging Narcotic Prescriptions
- 1615 Possession of Paraphernalia
- 1616 Obtaining Drugs Under False Pretenses
- 1617 Attempted Drug Offenses-Glue Sniffing
- 1618 Narcotic Information

LIQUOR LAWS

- 1701 Serving Liquor to Minors
- 1702 Selling Liquor to Minors
- 1703 Minor Consuming Liquor
- 1704 Misrepresenting Age to Obtain Liquor
- 1705 Liquor Sale Without a License
- Selling Liquor to Drunkards 1706
- 1707 Owning and Keeping Liquor with Intent to Sale
- 1708 Manufacture of Liquor
- 1709 Sale of Liquor During Off Hours
- 1710 Other Liquor Offenses

DRUNKENESS AND VAGRANCY

- 1801 Found Intoxicated
- 1802 Common Drunkeness
- 1803 Loitering on School Grounds
- 1804 Loitering
- 1805 Vagrancy
- 1806 Begging
- Curfew Violations 1807

KIDNAPPING

- 1901 Kidnapping
- Attempted Kidnapping 1902
- Unlawful Restraint 1903
- 1904
- Custodial Interference
- n 1905 Substitution of Children

R

55

ARSON-RECKLESS BURNING

2001 Arson 2002 Reckless Burning 2003 Attempted Arson 2004 Suspicious Fires

GAMBLING

2101 Keeping Gambling Devices Machines 2102 Playing Gambling Devices 2103 Frequenting Gambling House 2104 Keeping Gambling House 2105 Pool Selling 2106 Policy Playing 2107 Gambling Offenses-Other 2108 Baming

DISORDERLY CONDUCT-BREACH OF PEACE

2201	Breach of Peace
2202	Disorderly Conduct
2203	Harassment
2204	Riot
2205	Inciting to Riot
2206	Inciting Injury to Persons
	or Property
2207	Unlawful Assembly
2208	Roving Gangs
2209	Falsely Reporting an Incident
2210	School Disturbances
2211	Landlord-Tenant Dispute
2212	Dispute-No Disturbance

BRIBERY

- 2301 Bribery 2302 Bribery of a Police Officer 2303 Attempted Bribery 2304 Bribe Receiving 2305 Commercial Bribery-Offered or Received 2306 Tampering With a Juror 2307 False Statement 2308 Soliciting or Accepting Benefit for Rigging
- 2309 Participation in a Rigged Contest

SEE ALSO-

3101 Perjury

FAMILY AND CHILDREN

- 2401 Non-Support of Family
- 2402 Desertion or Abandonment of Family
- 2403 Non-Payment of Alimony
- 2404 Neglect or Abuse of Family or Child
- 2405 Family Dispute

ABORTION

- 2501 Performaning an Abortion
- 2502 Attempted Abortion
- 2503 Having Abortion Performed

DAMAGE TO PROPERTY

- 2601 Criminal Mischief-Public Property or Public Service
- 2602 Criminal Mischief-Private Property
- 2603 Tampering With Alarm Box 2604 Window Breaking-Auto or Building
- 2605 Fire Bombing

EAVESDROPPING

- 2701 Tampering With Private Communication
- 2702 Eavesdropping
- LICENSE VIOLATING
- 2901 Practicing Medicine Without a License
- 2902 Practicing Law Without a License
- 2903 Conducting a Party Without a License
- 2904 Building Without a Permit
- 2905 Feddling Without a Permit
- 2906 Impersonating a Police Officer
- Other License Violation 2907
- 2908 Impersonation of Other Official

SEE ALSO-1705 Liquor Sale Without a License

CODES VIOLATIONS

- 3001 Housing Code Violations 3002 Building Code Violations
- 3003 Sanitation Code Violations
- 3004 Health Code Violations
- 3005 Pollution Code Violations
- 3006 Zoning Violations
- 3007 Blue Law Violations
- 3008 Other Code Violations

Ð

COURT VIOLATIONS

3101 Perjury 3102 Contempt of Court 3103 Probation Violation 3104 Injunction Violation 3105 Failure to Appear-First Degree 3106 Failure to Appear-Second Degree 3107 Failure to Appear-Juvenile Court OFFENSES AGAINST ADMINISTRATION OF JUSTICE 3201 Interfering With an Officer 3202 Failure to Assist a Police Officer Hindering Prosecution 3203 Tampering With or Fabricating 3204 Evidence SEE ALSO-2306 Tampering With a Juror 2301 Bribery 2304 Bribe Receiving DISCRIMINATION-CIVIL LIBERTIES 3301 Discrimination-Housing 3302 Discrimination-Accomodations 3303 Discrimination-Employment 3304 Discrimination-Other 3305 Depriving Civil Liberties NUISANCES 3401 Animal Nuisance 3402 Animal Bites 3403 Criminal Trespass 3404 Discharging Fireworks 3405 Manufacture of Fireworks 3406 Sale of Fireworks 3407 Cruelty to Animals 3408 Abandoned Motor Vehicle 3409 Littering Violations 3410 Other Nuisances 3413 Injured Dog SEE ALSO-2203 Harassment (for Annoying or

- 2203 Harassment (for Annoying or Obscene Phone Calls and Disorderly Conduct)
- 2202 For Unreasonable Noise

ESCAPES

- 3501 Escape From a Correctional Institution
- 3502 Escape From Custody
- 3503 Prisoner Recovered
- 2506 Traball Recovered
- 3504 Escape From a Hospital or Sanatorium
- 3505 Hospital or Sanatorium Escapee Recovered
- 3506 Reformatory Escapee Recovered
- 3507 Conveying or Possessing Unauthorized Items
- 3508 Aiding an Escape
- 3509 Possession of Weapon in a Correctional Institution
- SEE ALSO-
- 3804 Youth Missing From an Institution

SUICIDE

- 3601 Suicide 3602 Attempted
- DEAD ON ARRIVAL
- 3701 All Cases—Dead on Arrival 3702 Sudden Death—Bodies Found
- 702 Budden Beach Boures Found
- 3703 Sudden Death-Apparent Overdose

MISSING OR LOCATED PERSON OR ANIMAL

3801 Missing Child (Aged 1-6) 3802 Missing Juvenile (Aged 7-15) 3803 Missing Adult (Aged 16 and Over) 3804 Youth Missing From an Institution 3805 Missing Animal 3806 Located Child (Aged 1-6) 3807 Located Juvenile (Aged 7-15) 3808 Located Adult (Aged 16 and Over) 3809 Located Youth From an Institution 3810 Located Animal FOUND PERSON OR ANIMAL 3901 Found Child (Aged 1-6) 3902 Found Juvenile (Aged 7-15) 3903 Found Adult (Aged 16 and Over) 3904 Found Animal

3905 Found Dead Animal

Å



CONTINUED 10F3

57

SUSPICION

4001 Prowler
4002 Suspicious Person
4003 Suspicious Activity
4004 License and Registration Check

ALARMS

4101 Burglar Alarms
4102 Fire Alarms—Fire
4103 Fire Alarms—No Fire
4104 Fire—No Alarm
4105 Holdup Alarms
4106 Other Alarms

PROPERTY

4201 Lost Property
4202 Found Property
4203 Recovered Property—Owner Known
4204 Recovered Property—Owner Unknown
4205 Mysterious Loss of Money

INVESTIGATIONS

4301	Intelligence Investigation-
	Subversion
4302	Intelligence Investigation-
	Crime
4303	Internal Investigation-Brutality
4304	Internal Investigation-
	Corruption
4305	Internal Investigation-
	Background

- 4306 Internal Investigation-Other
- 4307 Background Investigation
- 4308 Stakeout
- 4309 Drug Investigation
- 4310 Gambling Investigation
- 4311 Civil Investigation
- 4312 Field Interrogation
- 4313 Other Investigation

ACCIDENTS

- 4401 Motor Vehicle Accident-Fatal
- 4402 Motor Vehicle Accident—Non-Fatal 4403 Motor Vehicle Accident—Property Damage
- 4404 Non-Motor Vehicle Accident

MOTOR VEHICLE VIOLATIONS

- 4501 Motor Vehicle Violations-Moving Vehicle
- 4502 Motor Vehicle Violations-Parked Vehicle
- 4503 Driving While Intoxicated-Alcohol
- 4504 Driving While Intoxicated-Drugs
- 4505 Evading Responsibility

MEDICAL PROBLEMS

- 4601 Medical Assistance Needed
- 4602 Medical Assistance Rendered
- 4603 Drug Overdose
- 4604 Mental Cases
- 4605 Coronaries
- 4606 Drownings
- 4607 Falls

PROCESS SERVICE

- 4701 Warrant Service 4702 Subpoena Service
- 4703 Detention Notice

INTERNAL FUNCTIONS

4801 Police Garage (Service)
4802 Chief's Office
4803 Patrol Division
4804 Detective Division
4805 Traffic Division
4806 Personal
4807 Lunch
4808 Animal Shelter
4809 Court Duty
4810 Other Internal Functions
NO INCIDENT KNOWN

4901 Incident Unfound 4902 Wrong Address

MISCELLANEOUS

- 5001 Abduction to Marry
- 5002 Admit Minor Improper Place
- 5003 AWOL-Away Without Leave
- 5004 Boating Violations
- 5005 Building Found Open
- 5006 Conspiracy
- 5007 Contributing to the Delinquency of a Minor

MISCELLANEOUS (Continued)

5008	Door Checks
5009	Escort Duty
5010	Hitchhiking
5011	Immigration Law Violation
5012	Labor Law Violation
5013	Living on Condemned Property
5014	Minor Juvenile Complaints
5015	Neighbor Trouble
5016	Notification
5017	Park Violation
5018	Person Held-Other Town
5019	Public Hazards
5020	Unwanted Person(s)
5021	Vacant House
5022	Wanted Person-BPT.
5023	Wanted Person-Other Town
5024	Other Miscellaneous
5031	Money Transfer
5033	Injured Police Officer
5034	Truancy
5035	Untaxed Cigarettes
5036	Transient Lodgings



BRIDGEPORT INCIDENT REPORT FORM

17

APPENDIX B



78.8	C	1452		ARRA	ST	CASTA	LTY	ACC	ID ENT		RVIDERAR	
DATE OF REPOR	17		•		INCID	DENT DESCRIPTION	N					
TIME: RCD.	DISP.	AT.	2.0		TIME	OF OCCURRENCE	TIME	A.M. P.M.				
LOCATION OF .	NCIDENT					<u> </u>		LOCATION CO	DE (Office Or	MJ7)		
	LASTON	857				ADDRESS		CTTY	ទា	TATE	PHONE	
Vict								1				
D.O.B.	SEX F	ACE:	OCCUPATI	ON		EDUCATION () NO H.S.	() H.3.	BUSINESS ADE	RESS OR PHC	ONE		
WITNESS LAST	/FIRST				ADDRES	S	CITY		STATE		GE PHONE	2
SUSPECT LAST	Finst					<u></u>		······				
ARRESTED SUBI	ECTS LA	ST/FIRE	T	<u></u>	ADDRES	8	CITY	······································	STATE	8TX	RACE D	
								<u></u>				
Details of Is	icident-	-										• 7 1
												÷
										·····		
	86544 - 1809 1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1											
							<u></u>		· · · · · · · · · · · · · · · · · · ·			
								<u></u>				
											CHERC	
OFFICER REPORT	TING - RJ	WE -P	ULL NAME	BAI	DGE SECTO POST	OR/ IN SERV		Det Bur 4.	Dom Rel	7. 🗂 P 8	20	
Details of In	rident-	-Who,	What, V	When,	Where, WI	by and How.	1. 11 2. 11 3. 11 Accurac	Det Bur 4. [Youth 5. [Pros 6. [cy, Clarity an] Dom Rel] Traffic] 8 Serv nd Brevity	7. 8. 9. 0. 0. 0. 0. 0. 0. 0. 0	x O W. her Form 1063	34
Details of In	rident-	-Who,	What, V	When,	Where, Wi	by and How.	1. [] 1. [] 2. [] 3. [] Accurac	Det Bur 4. [Youth 5. [Pros 6. [cy, Clarity an] Dom Rel] Traffic] S Serv ad Brevity		z O W. her Form 1063	34
Details of In	cident-	-Who,	What, V	When,	Where, Wi	by and How.	Accurac	Det Bur 4. [Youth 5. [Pros 6. [cy, Clarity an] Dom Rel] Traffic] S Serv ad Brevity	7. P 8. D 7. 9. D 7.	2 O W. her Form 1063	3
Details of In	cident-	-Who,	What, V	When,	Where, Wi	by and How.	Accurac	Det Bur 4. [Youth 5. [Pros 6. [cy, Clarity an] Dom Rel] Traffic] S Serv ad Brevity	7. P8 8. D.X 9. OU	x O W. her Form 1063	8.
Details of In	rident-	-Who,	What, V	When,	Where, W	by and How.	Accurac	Pros 6. [Pros 6. [cy, Clarity a)] Dom Rel] Traffic] S Serv ad Brevity		x O W. her Form 1063	2.
Details of In	TING - R/	-Who,	What, V	When, V	Where, W	by and How.	Accurac	Pros 6. [Cy, Clarity and Compared and Compa] Dom Rel] Traffic] S Serv ad Brevity		x O W. her Form 1063	3.
Details of In	TDM3 - RJ	-Who,	What, V	¢/hen, \	Where, Wi	by and How.	Accurac	Pros 10: Det Bur 4. [Youth 5. [Pros 6. [cy, Clarity an] Dom Rel] Traffic] S Serv ad Brevity		z O W. her Form 1063	3.
Details of In	rident-	-Who,	What, V	When,	Where, Wi	by and How.	Accurac	Pros 0: Youth 5. [Pros 6. [cy, Clarity a)] Dom Rel] Traffic] S Serv ad Brevity	7. P 6 8. D 7 9. O 00 7. 1	x O W. her Form 1063	3
Details of In	icident-	-Who,	What, V	¢hen, \	DGE SECTO POST	by and How.	Accurac	Pros 6. [cy, Clarity and Compared and Comp] Dom Rel] Traffic] S Serv ad Brevity		r O W. her Form 1063	3.
Details of In	ES OR LO	-Who,	What, V	When, V	DGE SECTO POST	PREMISES;	Accurac	IVATE VEHICLE:] Dom Rel] Traffic] S Serv ad Brevity		Form 1063	84
Details of In Details of In	ES OR LO	-Who,	What, V	47hen, 1	COMMERCIAL J Dept. SL	PREMISES:	Accurac	Pros 6. Cyouth 5. Pros 6. cy, Clarity a)] Dom Rel] Traffic] S Serv ad Brevity		Form 1063	34
Details of In Details of In TYPE OF PREMIS PRIVATE	ES OR LO	-Who,	What, V	2A.1	ECOMMERCIAL J Dept. St J Business	PREMISES:	Accurac PRI 71. [] PRI 72. [] 73. [] 73. []	IVATE VEHICLE: Automobile Motorcycle] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X 9. Oth 7. T 7. T 7. T 7. T 7. T 7. T 7. T 7. T	Form 1063 Form 1063	34
Details of In Details of In TTPE OF PREMIS PRIVATE I House I Apartn I Trailer	ES OR LO DWELLIN nent	-Who,	What, V	47hen, 1 47hen, 1 47hen, 1 41. [42. [43. [44. [COMMERCIAL J Dept. St Bank	PREMISES: Ore tore 3	Accurac 71. [] 3. [] Accurac 71. [] 72. [] 73. [] 74. []	IVATE VEHICLE: Automobile Motorcycle] Dom Rel] Traffic] S Serv ad Brevity	7. [] P & 8. [] D.X 9. [] Oth 7.] 7.] 7.] 7.] 7.] 7.] 7.] 7.]	Form 1063 Form 1063 Sidewalk Street Water	84
Details of In Details of In FIFE OF PREMIS PREVATE 1. House 2. Apartn 3. Trailer COMMEN	ES OR LO DWELLIN hent	-Who, -CALE (G)	What, V	*/hen, \ */hen, \ 4/hen, \ 41. [42. [43. [44. [45. [COMMERCIAL J Dept. St Bank Gas Sta	DR/ IN SERV TIME hy and How.	Accurac 71. [] 3. [] Accurac 71. [] 72. [] 73. [] 74. [] 75. []	IVATE VEHICLE: Automobile Bus IVATE VEHICLE: Automobile Bus Truck Roat] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X 9. Oth 7. T 7. T 9. Oth 7. T 7. T 9. Oth 7. T 7. T 9. Oth 7. T 9. Oth 7. T 7. T 9. Oth 7. T 7. T 9. Oth 7. T 7. T 7. T 7. T 7. T 7. T 7. T 7. T	Form 1063 Form 1063 Sidewalk Street Water	34
Details of In Details of In TYPE OF PREMIS PRIVATE I	ES OR LO DWELLIP nent	-Who, -Who, -CALE KG:	What, V	41. [43. [43. [43. [43. [44. [43. [44. [45. [46. [Where, Wi Where, Wi Commercial J Dept. St J Business J Bank J Car Lota	by and How. hy and How. hy and How. hy and How. hy and How. hy and How. hy and How.	Accurac 71. [] 3. [] Accurac 71. [] 72. [] 73. [] 74. [] 75. [] 75. [] 75. []	IVATE VEHICLE: Automobile Motorcycle Bus Truck Motorcal VEH] Dom Rel] Traffic] S Serv ad Brevity	7. P & & D.X 8. D.X 9. Other 7. Other 9. O	Form 1063 Form 1063 Sidewalk Street Water Hher Construction	34
Details of In Details of In TTPE OF PREMIS PRIVATE Details of In TTPE OF PREMIS PRIVATE Details of In Trailer Details Private Details of In Trailer Details Of Premis Details	ES ON LO CONV.	-Who, -Who, -CALE Gi ELLING	What,	41. [42. [43. [44. [43. [44. [45. [46. [45. [Where, Wi Where, Wi Where, Wi Where, Wi Where, Wi Where, Wi Dept. Sta Dept. Sta Dept. Sta Bank Gas Sta Car Lot, NERTAINM	A PREMISES: Lore tion /Parking Lot	Accurac 71. [] 2. [] 3. [] Accurac 73. [] 74. [] 75. [] 74. [] 75. [] 81. []	IVATE VEHICLE: Automobile Matorcycle] Dom Rel] Traffic] S Serv ad Brevity	7. [] P & 8. [] D.X 9. [] OU 7.] 7.] 9. [] OU 7.] 9. [] OU 9. [] S 91. [] S 92. [] X 99. [] O 90. [] S 92. [] X 99. [] O	Form 1063 Form 1063 Form 1063 Sidewalk Street Water Street Water Street Water Street Water	34
Details of In Details of In TTPE OF PREMIS PRIVATE I. House I. House I. Hotel 1 I. Hotel 1 I.	ES ON LO ; DWELLIP nent iCIAL DW Motel II CONV. ory: Roo UNK FRED	-Who, -Who, -CALE Gi relling alesce: oming dists:	What,	47hen, 1 47hen, 1 47hen, 1 41. [42. [43. [44. [45. [46. [251. [52. [COMMERCIAL J Dept. St Bank Gas Sta Car Lota NULTAINMI Park/Re Sports A	Arena	Accurac	IVATE VEHICLE: Automobile Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Taxicab] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X 9. Oth 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1	Form 1063 Form 1063 Form 1063 Sidewalk Siteet Water HorrEnder Chative Criend Acquaint,	3.
Details of In Details of In TPE OF PREMIS PRIVATE I. HOUSE I. HOUSE I. HOUSE I. HOUSE I. HOUSE I. HOUSE II. HOUSE III. HOUSE IIII. HOUSE III. HOUSE III. HOUSE III. HOUSE IIII. HOUSE IIII. HOUSE IIII. HOUSE IIII. HOUSE IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	ES OR LO CONVENTION CONVENTI	-Who, -Who, -CALE Gale CALE Gi etting disce:	What,	47hen, 1 47hen, 1 47hen, 1 41. [43. [44. [45. [46. [2 51. [52. [53. [COMMERCIAL DEPUTY Where, Wi Where, Wi Where, Wi Dept, Sti Dept, Sti Dept, Sti Bank Das Sta Dank Car Lot, NTERTAINM Park/Re Deputy A	by and How. hy and How. hy and How. PREMISES: Ore tore tion (Parking Lot ENT ADEA: cereation Area Arena	PRI 71. [] 72. [] 73. [] 74. [] 75. [] 74. [] 75. [] 81. [] 82. [] 83. [] 84. []	IVATE VEHICLE: Automobile Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Motorcycle Truck MMERCIAL VEHO Automobile Motorcycle Taxicab Bus] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X. 9. Oth 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1	Form 1063 Form 1063 Form 1063 Sidewalk Siteet Water Sineet Water Street Water Sine Site Sitewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk Siteet Sidewalk	
Details of In Details of In TTPE OF PREMIS PRIVATE I. House I. House I. House I. Hotel I. Hotel I. Hotel I. Hotel I. Hotel I. Dornit I. Prob Dr I. Restau I. Drive-1 I. Drive-1 I. Hotel I. Dornit I. Hotel I. Dornit I. Hotel I. Dornit I. Hotel I. Dornit I. Hotel I.	ES OR LO DWELLE CONV.	-Who, -Who, -Who, -Who, Who, P P P P P P P P P 	What,	47hen, V 47hen, V 41. [42. [43. [44. [45. [51. [52. [53. [53.]	COMMERCIAL DEP. SU DEP. SU DEP. SU DEP. SU Bank Dank Car Lot. NTERTAINMI Park/Re DEP. SULE DEP. SULE	by and How. hy and How. hy and How. PREMISES: Ore tion (Parking Lot ENT ABEA: cereation Area Arena Ding8:	Accurac Accurac 71. [] 3. [] Accurac FRI 71. [] 72. [] 73. [] 74. [] 75. [] 81. [] 81. [] 81. [] 81. [] 82. [] 83. [] 83. []	IVATE VEHICLE: Automobile Motorcycle Taxicab Bus Truck Motorcycle] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X. 9. Oth 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1	Form 1063 Form 1063 Form 1063 Sidewalk Sitreet Water Hher Hher Honship (enalive Friend Acquaint, Had Seen No Relation	
Details of In Details of In Details of In TPE of PREMIS PRIVATE I. House I. House I. Hotel 1 I. Hotel	ES OR LO DWELLE CONV. OFY: ROU OFY: ROU OFY: ROU INK FRE FRE FRE FRE Store	-Who, -Who, -Who, -Who, 	What,	aAi 47hen, \ 41. 42. 43. 44. 51. 52. 53. 61.	Where, Wi Post Where, Wi Where, Wi Where, Wi Dept. St Dept. St Dept. St Bank Car Lott, NTERTAINMI Park/Re Sports A Theater OBLIC SULL School/I	hy and How. hy and How. hy and How. PREMISES: Ore tore	Accurac Accurac 71. [] 3. [] Accurac 73. [] 74. [] 75. [] 74. [] 75. [] 81. (] 81. (] 82. [] 83. [] 84. [] 85. [] 86. []	IVATE VEHICLE: Automobile Motorcycle Motorcycle Motorcycle Taxicab Bus Truck Boat Motorcycle Taxicab Bus Truck Boat] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X. 9. Other 7. Characteristics 7. Characteristics 9. Char	Form 1063 Form 1063 Form 1063 Sidewalk Sidewalk Sitret Water Stret Water Stret Water Stret Water Stret Water Stret Nher Chalve Criend Norringen Constant, Ind Seen No Relation Juknown	
Details of In Details of In TPE OF PREMIS PRIVATE Details Details of In TPE OF PREMIS PRIVATE Details	ES OR LO DWELLE ACTAL DW Motei 11 Conv. ory: Roc UNE PRE rant in/Take Store	-Who, -Who, -Who, -Who, 	What,	Avhen, v Avhen, v 41. [42. [43. [44. [51. [52. [] 53. [61. [62. [62. []	Where, Wi Post Where, Wi Where, Wi Where, Wi Dept. St Dept. St Dept. St Bank Car Lot, NUBRAINMI Park/Re Sports A Theater OBLIC SULL School/II Governn Covernn	hy and How. hy an	Accurac Accurac 71. [] 3. [] Accurac 73. [] 74. [] 75. [] 74. [] 75. [] 81. (] 81. (] 82. [] 83. [] 84. [] 85. [] 86. [] 87. []	IVATE VEHICLE: Automobile Motorcycle Motorcycle Motorcycle Truck Motorcycle Truck Boat Motorcycle Truck Boat Truck Boat Truck Boat Truck] Dom Rel] Traffic] S Serv ad Brevity	7. P & 8. D.X. 9. Other 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1 7. 1	Form 1063 Form 1063 Form 1063 Sidewalk Sidewalk Street Water Pher Honship (Inalive Grading Constant, Find So Relation Juknown	



APPENDIX C

(Ö)

ø

CHAPTER II TABLES

is)



Table II-1

FREQUENCY PERCENTAGE COMPARING CRIME TYPE IN BRIDGEPORT POPULATION DATA AND SAMPLE DATA FOR 1975

	Populatic	on (80,966)	Sample	(8,107)	
Crime Type	Number	Percent	Number	Percent	
Murder-Manslaughter	19	0.0%	1	0.0%	
Robbery	697	0.9	59	0.7	
Assault	3851	4.8	365	4.6	
Burglary	3806	4.7	391	4.9	
Larceny LE \$50	3486	4.3	334	4.2	
Larceny GT \$50	4295	5.3	444	5.5	
Motor Vchicle Theft	7720	9.5	815	10.2	
Forgery	187	0.2	15	0.2	
Other Larceny/Fraud Embezzlement	. 419	0.5	45	0.6	
Stolen Property	31	0.0	6	0.1	
Weapons	155	0.2	18	0.2	
Vice	155	0.2	11	0.1	
Sex Offenses	138	0.2	18	0.2	
Obscenity	5	0.0	1	0.0	
Drugs	562	0.7	63	0.8	
Liquor Laws	157	0.2	20	0.2	
Drunkedness	821	1.0	88	1.1	
Kidnapping	26	0.0	1	0.0	
Arson	117	0.1	16	0.2	
Gambling	252	0.3	17	0.2	
Disorderly Conduct	3382	4.2	342	4.3	
Bribery	17	0.0	2	0.0	

(Continued)

۲

۲

6

8

.

Table II-1 (Continued)

FREQUENCY PERCENTAGE COMPARING CRIME TYPE IN BRIDGEPORT POPULATION DATA AND SAMPLE DATA FOR 1975

	Populatio	on (80,966)	Sample	(8,107)
Crime Type	Number	Percent	Number	Percent
Family Offenses	2417	3.0%	238	3.0%
Property Damage	6290	7.8	598	7.5
License Violations	29	0.0	2	0.0
Code Violations	44	0.1	5	0.1
Court Violations	116	0.1	12	0.1
Impeding Justice	228	0.3	29	0.4
Nuisances	8017	9,9	811	10.1
Escapes	68	0.1	5	0.1
Suicide	90	0.1	9	0.1
Dead on Arrival	320	0.4	38	0.5
Missing Persons	1208	1.5	115	1.4
Found Persons	110	0.1	9	0.1
Suspicion	_ 1476	1.8	140	1.7
Alarms	5738	7.1	554	6.9
Property	816	1.0	78	1.0
Investigations	276	0.3	31	0.4
Accidents	905	1.1	88	1.1
Motor Vehicle Violations	3550	4.4	340	4.2
Medical Problems	8402	10.4	824	10.3
Warrant-Subpoena	834	1.0	76	0.9
No Incident	2407	3.0	233	2.9
Miscellaneous	7279	9.0	710	8.9
Totals	80916	94.6%	8017	100.0%

Table II-2 Sample Data — 1975 CRIME CODE - COLLAPSED CATEGORIES

Crime Type	Number	Percent
Murder-Manslaughter	1	0.0%
Robbery	59	0.7
Assault	365	4.6
Burglary	391	4.9
Larceny LE \$50	334	4.2
Larceny GE \$50	444	5.5
Motor Vehicle Theft	815	10.2
Disorderly Conduct	342	4.3
Family Offenses	238	3.0
Property Damage	598	7.5
Nuisances	811	10.1
Alarms	554	6.9
Motor Vehicle Violations	340	4.2
Medical Problems	824	10.3
Miscellaneous Juvenile	710	8.9
*Other-Person	458	5.7
*Other-Property	282	3.5
*Vice	218	2.7
No Incident	233	2.9
Totals	8017	100.0%
Missing Cases	0	

']	Table	II-3	
Sam	ple D)ata —	1975
CAR	POST	SECTO	RS

Car Post Sectors	Number	Percent
A Amber	1314	16.4%
B Blue	1230	15.3
C Claims Car	22	0.3
D Detective Bureau	128	1.6
F Accident Cars	101	1.3
G Green	1906	23.8
H Headquarters, 2nd, 3rd, Precinct	271	3.4
I Board of Education	11	0.1
J Special	270	3.4
K K-9 Corps	11	0.1
L Post Number	34	0.4
M Motorcycles	13	0.2
N Ambulance	659	8.2
P Park Police	7	0.1
Q Mobile Van	10	0.1
R Red	872	10.9
S Special Services	34	0.4
T Tactical Squad	90	1.1
V FPV Specials	351	4.4
W Dog Warden	508	6.3
Y Youth Bureau	84	1.0
Z Police Sergeant	22	0.3
Totals	7947	100.0%
Missing Cases	69	

Table II-4

GRID CODE — SAMPLE DATA — 1975 STREET LOCATIONS WITH HIGH FREQUENCIES OF INCIDENTS

Grid	Number	Percent
002	50	2.1%
003	152	6.3
004	53	2.2
008	68	2.8
009	345	14.4
0010	130	5.4
0011	174	7.3
0015	50	2.1
0016	168	7.0
0017	246	10.3
0018	52	2.2
0020	64	2.7
0021	57	2.4
0026	73	3.0
0027	129	5.4
008A	395	16.5
008B	100	4.2
Outside City	88	3.7
Totals	2394	100.0%

Missing Cases
GRID CODE - POPULATION DATA - 1975 STREET LOCATIONS WITH HIGH FREQUENCIES OF INCIDENTS

6

Grid	Number	Percent	Percent
002	517	C.6%	2.1%
003	1631	2.0	6.7
004	608	0.8	2.5
008	734	0.9	3.0
009	3443	4.3	14.2
0010	1352	1.7	5.6
0011	1806	2.2	7.5
0015	430	0.5	1.8
0016	1797	2.2	7.4
0017	2352	2.9	9.7
0018	490	0.6	2.0
0020	653	0.8	2.7
0021	575	0.7	2.4
0026	654	0.8	2.7
0027	1466	1.8	6.0
008A	3671	4.5	5.2
008B	987	1.2	4.7
Outside City	1039	1.3	4.3
Totals	24205	29.8%	100.5%

Missing Cases 56751

Sample Data — 1975 INCIDENT LOCATION TYPE

6

0

Location	Number	Percent
House	285	12.2%
Apartment	610	26.1
Hotel, Motel	. 9	0.4
Hospital, Convalescence Home	20	0.9
Dormitory, Rooming House	8	0.3
Restaurant	. 22	0.9
Drive-In, Take-Out	4	0.3
Bar	39	1.7
Liquor Store	5	0.2
Market	10	0.2
Department Store	40	1.7
Small Store	69	1.7
Business	188	3.0
Band	11	0.5 [°]
Gas Station	26	0.5
Car Lot, Parking Lot	81	1.1 7 E
Park-Recreational Area	26	· 1 1
Sports Arena	3	
Theater	3	0.1
School, Library	58	2.5
Government Building	22	2.5
Church	7	0.3
Private Auto	459	19.6
Private Motorcycle	5	0.2
Private Bus	2	0.2
Private Truch	14	0.6
Community Auto	4	. 0.2
Community Taxi	4	0.2
Community Bus	. 1	0.0
Community Truck	6	0.3
Sidewalk	90	3.8
Street	141	6.0
Water	1	0.0
Other Totals Missing Cases	65 2338 3679	2.8 100.0%

Table	; II-7
Sample D	ata — 1975
AGE OF	VICTIMS

	Age	Number	Percent
l to	10 Years	38	3.0%
11 to	18 Years	114	8.9
19 to	30 Years	411	32.2
31 to	45 Years	310	24.3
46 to	60 Years	244	19.1
61 and	l Over	_160	_12.5
	Totals	1277	100.0%

Missing Cases

6740

.

۲

Table	II-8
Sample Da	ta — 1975
SEX OF	VICTIM

Sex of Victim	Number	Percent
Female	1030	44.2%
Male	1302	55.8
Totals	2332	100.0%
Missing Cases	5685	

Tab	le	II-9)
Sample	Da	ta —	1975
RACE	OF	VICT	IM

Race of Victim	Number	Percent
White	1318	64.5%
Black	520	. 25.5%
Puerto Rican	200	9.8
Other	5	0.2
Totals	2043	100.0%
Missing Cases	5974	

Table II-10 Sample Data — 1975 MONTH INCIDENT OCCURRED

Month	Number	Percent
January	717	9.2%
February	646	8.3
March	667	8.6
April	644	8.3
May	718	9.3
June	714	9.2
July	638	8.2
August	683	8.8
September	583	7.5
October	611	7.9
November	604	7.8
December	537	6.9
Totals	7762	100.0%

Missing Cases

6

Table II-11 Sample Data — 1975 MONTH INCIDENT REPORTED

Month	Number	Percent
January	722	9.0
February	659	8.2
March	. 682	8.5
April	664	8.3
Мау	734	9.2
June	739	9.2
July	653	8.2
August	712	8.9
September	605	7.6
October	646	8.1
November	617	7.7
December	572	7.1
Totals	8005	100.0%

Missing Cases

12

٩

Table II-12 Sample Data — 1975 DAY OF MONTH INCIDENT OCCURRED

Day of Month	Number	Percent
1		Kontanti ann an Anna
2	270	3.0%
3	271	3.0
4	250	3.0
5	. 272	4.0
5	261	3.0
7	283	4.0
, P	261	3,0
0	252	3.0
9	237	3.0
10	270	3.0
11	246	3.0
12	259	3.0
13	269	3.0
14	241	3.0
15	265	3.0
16	279	4.0
17	, 252	3.0
18	251	3.0
19	219	3.0
20	270	3.0
21	251	3.0
22	244	3.0
23	255	3.0
24	254	3.0
25	246	3.0
26	235	3.0
27	246	3.0
28	240	3.0
29	236	3.0
30	209	3.0
31	. 158	2.0
Totals	7752	100.0%
Missing Cases	265	

Table II-13 Sample Data — 1975 DAY OF MONTH INCIDENT REPORTED

0

Day of Month	Number	Percent
1	276	3.0
2	282	4.0
3	249	3.0
4	285	4.0
5	261	3.0
6	296	4.0
7	274	3.0
8	249	3.0
9	253	3.0
10	279	3,0
11	255	3.0
12	258	3.0
13	260	3.0
14	247	3.0
15	278	3.0
16	288	4.0
17	256	3.0
18	255	3.0
19	240	3.0
20	280	3.0
21	256	3.0
22	258	3.0
23	268	3.0
24	261	3.0
25	259	3.0
26	240	3.0
27	245	3.0
28	250	3.0
29	256	3.0
30	229	3.0
31	164	2.0
lotals	8007	100.0%
Missing Cases	10	

TIME INCIDENT OCCURRED: AM-PM

Time	Number	Percent
AM	1499	34.4%
РМ	2864	65.6
Totals	4363	100.0%
Missing Cases	3654	

TIME INCIDENT REPORTED: AM-PM

Time	Number	Percent
AM	2832	36.6%
РМ	4914	63.4
Totals	7746	100.0%

Missing Cases

0

Tabl	e II	-15	
Sample	Data	- 1	975

TIME INCIDENT REPORTED: 24 - HOUR CLOCK

Time	Number	Percent	Time	Number	Percent
0000	179	4.0%	0000	287	4.0%
0100	125	3.0	0100	212	3.0
0200	102	2.0	0200	135	2.0
0300	49	1.0	0300	100	1.0
0400	35	1.0	0400	73	1.0
0500	75	2.0	0500	136	2.0
0600	76	2.0	0600	160	2,0
0700	129	3.0	0700	278	4.0
0800	192	4.0	0800	388	5.0
0900	213	5.0	0900	374	5.0
1000	181	4.0	1000	403	5.0
1100	214	5.0	1100	303	4.0
1200	255	6.0	1200	422	5.0
1300	231	5.0	1300	374	5.0
1400	243	5.0	1400	412	5.0
1500	286	6.0	1500	481	6.0
1600	255	6.0	1600	429	6.0
1700	227	5.0	1700	428	6.0
1800	251	6.0	1800	417	5.0
1900	215	5.0	1900	402	5.0
2000	301	7.0	2000	442	6.0
2100	276	6.0	2100	420	5.0
2200	227	5.0	2200	332	4.0
2300	210	5.0	2300	358	5.0
Totals	4547	100.0%	Totals	7766	100.0%
Missing Ca	ses 3470		Missing Ca	ses 251	

Table II-16 Sample Data --- 1975 TIME INCIDENT OCCURRED: 12 HOUR SEGMENTS

Time	Number	Percent
3 PM to 2 AM	2795	61.5%
3 AM to 2 PM	1752	
Totals	4547	100.0%
Missing Cases	3470	

Ø

Ð

TIME INCIDENT REPORTED: 12 HOUR SEGMENTS

Time	Number	Percent
3 PM to 2 AM	4620	59.5%
3 AM to 2 PM	3146	_40.5
Totals	7766	100.0
Missing Cases	251	



APPENDIX D

目は

 $\int f$

CHAPTER III TABLES



VICTIM'S SEX BY VICTIM'S RACE

		Race														
Sor	Wh	ite Percent	Bla	Percent	Puerto	Rican	Other Number Percent									
	<u>Mamber</u>	<u>1010011</u>	Mamber		Humber	<u>rereene</u>	Humber	<u>i ci cent</u>								
Male	775	58.9%	236	45.4%	111	55.8%	4	80.0%								
Female	540	41.1	284	54.6	88	44.2	1	20.0								
Totals	1315	100.0%	520	100.0%	199	100.0%	5	100.0%								

۲

VICTIM'S AGE BY VICTIM'S SEX

	Sex												
	Ma	le	Female										
Years	Number	Percent	Number	Percent									
l to 10 Years	28	4.0%	9	1.6%									
11 to 18 Years	56	8.0	56	9.9									
19 to 30 Years	198	28.4	211	37.3									
31 to 45 Years	165	23.6	140	24.8									
46 to 60 Years	159	22.8	84	14.9									
6] and Over	92	13.2	65	11.5									
Totals	698	100.0%	565	100.0%									

Table I	II-3
---------	------

VICTIM'S AGE BY VICTIM'S RACE

			Касе													
			W	nite	B	lack	Puer	to Rican	Other							
	Years		Number	Percent	Number	Percent	Number	Percent	Number	Percent						
1	to	10 Years	18	2.5%	14	3.9%	4	3.0%	0	0.0%						
11	to	18 Years	54	7.5	35	9.6	23	17.4	0	0.0						
19	to	30 Years	226	31.3	118	32.5	46	34.8	1	25.0						
31	to	45 Years	152	21.1	102	28.1	39	29.5	2	50.0						
46	to	60 Years	153	21.2	67	18.5	12	9.1	1	25.0						
61	to	Highest	119	16.5	_27	7.4	8	6.1	0	0.0						
	r.	Fotals	722	100.0%	363	100.0%	132	99.9%	4	100.0%						

VICTIM'S SEX BY NEWCRIME

	Crime Code																			
	Murder Manslaughter		Robbery		Assault		Burglary		Larceny LE \$50		Larceny GE \$50		Motor Vehicle Theft		Disorderly Conduct		Family Offenses		Property Damage	
Sex	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Male	1	100.0%	16	69.6%	57	45.6%	48	52.2%	48	48.0%	106	72.1%	217	71.1%	41	43.6%	31	32.0%	108	59.3%
iemale	<u>v</u>	0.0	_7	_30.4	68	54.4	44	47.8		52.0		27.9	88	28.9	<u>53</u>	56.4	<u>66</u>	68.0	74	40.7
Total	ls 1	100.0%	23	100.0%	125	100.0%	92	100.0%	100	100.0%	147	100.0%	305	100.0%	94	100.0%	97	100.0%	182	100.0%
Continu	ued																			

80

t

, ii

Table III-4 (Continued)

0

VICTIM'S SEX BY NEWCRIME

									Crime									
Sex	Nuisances Number Percent		Alarms Number Percent		Motor Vehicle Violation Number Percent		Medical Problems Number Percent		Miscellaneous Juvenile Number Percent		Other Person Number Percent		Other Property Number Percent		Vice Number Percent		No Incident Number Percent	
Male	99	50.5%	17	68.0%	85	70.8%	190	54.8%	93	41.9%	58	52.3%	46	63.0%	22	61.1%	19	52.8%
Female	97	49.5	_3	32.0	35	29.2	<u>157</u>	45.2	<u>129</u>	58.1	53	47.7	<u>27</u>	37.0	<u>14</u>	38.9	<u>17</u>	47.2
Totals	196	100.0%	25	100.0%	120	100.0%	347	100.0%	222	100.0%	111	100.0%	73	100.0%	36	100.0%	36	100.0%

Missing Cases 5685

VICTIM'S RACE BY NEWCRIME

										Crime	Code								ور المساور وم معالي و المساطر و	<u></u>
	Mur Mansla	rder aughter	Rot	obery	As	sault	Burg	glary	Larceny	LE \$50	Larceny	/ GE \$50	Motor Tł	Vehicle neft	Dison Cor	derly duct	Fan Offe	nily enses	Prop Dan	verty nage
Race	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
White	. 1	100.0%	15	71.4%	68	58.6%	51	63.0%	73	76.0%	120	87.6%	139	68.8%	56	64.4%	37	41.6%	135	78.5%
Black	0	0.0	6	28.6	36	31.0	21	25.9	17	17.7	16	11.7	42	20.8	25	28.7	35	39,3	28	16.3
Puerto R	ican O	0.0	0	o. 0	12	10.3	9	11.1	6	6.3	1	0.7	20	9.9	6	6.9	15	16.9	9	5.2
Other	<u>0</u>	0.0	_0	0.0	0	0.0	_0	0.0	_0	0.0	0	0.0	_1	0.5	_0	0.0	_2	2.2	0	0.0
Total	.s 1	100.0%	21	100.0%	116	100.0%	81	100.0%	96	100.0%	137	100.0%	202	100.0%	87	100.0%	89	100.0%	172	100,0%
Total	.s I	100.0%	21	100.0%	116	100.0%	81	100.0%	96	100.0%	137	100.0%	202	100.0%	87	100.0%	89	100.0%	172	100,

									Crimo	Code								
	Nuis	ance	A1a	irms	Motor Viola	Vehicle tions	Mec Prol	dical blems	Miscel Juve	llancous enile	Other	-Person	Other-	Property	<u>v</u>	ice	No II	ncident
Race	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
White	119	70.8%	16	80.0%	70	76.1%	167	49.9%	119	57.2%	61	64.2%	41	62.1%	15	50.0%	15	55.6%
Black	34	20.2	2	10.0	10	10.9	115	34.3	68	32.7	25	26.3	18	27.3	11	36.7	11	40,7
Puerto Rica	n 15	8.9	2	10.0	11	12.0	52	15.5	21	10.1	9	9.5	7	10.6	4	13.3	1	3.7
Other	0	0.0	_0		_1	_1.1_		0.3	0	0.0		0.0		0.0	_0	0.0	_0	0.0
Totals	168	99.9%	20	100.0%	92	100.0%	335	100.0%	208	100.0%	95	100.01	66	100.0%	30	100.0%	27	100.0%
Missing	Cases	5974																

1	rable I	II-	5
((Contir	nuec	1)
VICTIM'S	RACE	BY	NEWCRIME

II-6	13	¢	1	ь	'n	7
II-6	11	¢	1	ь	'n	7

VICTIM'S AGE BY NEWCRIME

										Crime	Code									
	Mur Mansla	der- lughter	Rot	obery	As	sault	Burg	glary	Larcen	y LE \$50	Larceny	/ GE \$50	Motor Ti	Vehicle left	Disc Cor	orderly nduct	Fai Off	mily enses	Pro Da	perty mage
Years	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
l to 10 Years	0	0.0%	0	0.0%	1	1.3%	0	0.0%	1	1.7%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%
ll to 18 Years	0	0.0	1	5.3	15	19.2	2	3.9	6	10.3	5	6.0	4	2,9	2	4.8	5	8,9	1	1.1
19 to 30 Years	0	0.0	5	31.3	36	46.2	22	45.1	15	25,9	43	51.2	47	34.1	19	45.2	19	33.9	27	31.0
31 to 45 Years	1	100.0	5	31.3	16	20.5	13	25.5	16	27.6	18	21.4	35	25.4	9	21.4	20	35.7	23	25.4
46 to 60 Years	0	0.0	5	31.3	10	12.8	7	13.7	13	22.4	11	13.1	38	27.5	8	19.0	9	16.1	23	26.4
6l to Highest	<u>0</u>	0.0		0.0	0	0.0	6	11.8	7	12.1		8.3	14	10.1	4	9.5	3	5.4	<u>13</u>	14.9
Total	ls 1	100.0%	16	100.0%	78	100.0%	51	100.0%	58	100.0%	84	100.0%	138	100.0%	42	99.5%	56	100.0%	87	99.8%

Continued

.

									Crim	e Code								
	Nui:	sance	A1;	arms	Motor Viol	Vehicle ations	Mec Prol	dical blems	Misce Juv	llaneous enile	Other	-Person	Other-	Property	<u> </u>	ice	No II	ncident
Years	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
1 to 10 Years	11	16.9%	0	0.0%	1	2.0%	18	6.2%	2	1.7%	0	0.0%	3	5.5%	0	0.0%	1	6.3%
11 to 18 Years	14	21.5	. 0	0.0	1	2.0	36	12.5	8	6.8	5	11.1	7	12.7	1	6.3	1	6.3
19 to 30 Years	15	23.1	5	41.7	18	35.3	58	20.1	35	29.9	14	31.1	23	41.8	7	43.8	2	12.5
31 to 45 Years	12	18.5	3	25.0	13	25.5	61	21.1	35	29.9	12	26.7	13	23.6	1	6.3	4	25.0
46 to 60 Years	8	12.3	2	16.7	13	25.5	55	19.0	· 21	17.9	8	17.8	6	10.9	4	25.0	3	18.8
61 to Highest	_5	7.7	_2	16.7	5	9.8	61	21.1	_16	13.7	_6	13.3	3	5.5	_3	18.8	_5	31.3
Totals	65	100.0%	12	100.1%	51	100.1%	289	100.1%	117	`99 . 9%	45	100.0%	55	100.0%	16	100.2%	16	100.2%
Missir	ng Cases	6470																

Table III-6 (Continued) VICTIM'S AGE BY NEWCRIME

VICTIM'S SEX BY CAR POST SECTOR

				•					Car Post	t Sector	r								
Po Ser	lice géant	Youth	Bureau	Dog	Warden	FPV S	pecials	Tactica	1 Squad	Spe Serv	ecial vices	<u> </u>	Red	Amb	ulance	Motor	rcycles	Post	Number
Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
6	66.7%	7	50.0%	63	52.5%	40	34.8%	10	90.9%	2	50.0%	137	54.6%	150	57.3%	2	66.7%	4	100.0%
3		_7	50.0	57	47.5	_75	65.2	_1	9.1	2	50.0	<u>114</u>	45.4	<u>112</u>	42.7	<u>1</u>	33,3	<u>0</u>	0.0
ls 9	100.0%	14	100.0%	120	100.0%	115	100.0%	11	100.0%	4	100.0%	251	100.0%	262	100.0%	3	100.0%	4	100.0%
	Po <u>Ser</u> <u>Number</u> 6 <u>3</u> 1s 9	Police Sergeant <u>Number Percent</u> 6 66.7% <u>3 33.3</u> 1s 9 100.0%	Police Sergeant Youth Number Percent Number 6 66.7% 7 3 33.3 7 1s 9 100.0% 14	Police Sergeant Youth Bureau Number Percent Number 6 66.7% 7 50.0% 3 33.3 7 50.0 1s 9 100.0% 14 100.0%	Police Sergeant Youth Bureau Dog Number Percent Number Percent Number 6 66.7% 7 50.0% 63 <u>3</u> <u>33.3</u> 7 <u>50.0</u> <u>57</u> 1s 9 100.0% 14 100.0% 120	Police Sergeant Youth Bureau Dog Warden Number Percent Number Percent 6 66.7% 7 50.0% 63 52.5% 3 33.3 7 50.0 57 47.5 1s 9 100.0% 14 100.0% 120 100.0%	Police Sergeant Youth Bureau Dog Warden FPV Sp Number Percent Number Percent Number 6 66.7% 7 50.0% 63 52.5% 40 3 33.3 7 50.0 57 47.5 75 1s 9 100.0% 14 100.0% 120 100.0% 115	Police Sergeant Youth Bureau Dog Warden FPV Specials Number Percent Number Percent Number Percent 6 66.7% 7 50.0% 63 52.5% 40 34.8% 3 33.3 7 50.0 57 47.5 75 65.2 1s 9 100.0% 14 100.0% 120 100.0% 115 100.0%	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Number Percent Number Percent Number Percent Number 6 66.7% 7 50.0% 63 52.5% 40 34.8% 10 <u>3</u> <u>33.3</u> <u>7</u> <u>50.0</u> <u>57</u> <u>47.5</u> <u>75</u> <u>65.2</u> <u>1</u> 1s 9 100.0% 14 100.0% 120 100.0% 115 100.0% 11	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Number Percent Number Secondary Number Secondary Number Number Percent Number Number Number Number Number Secondary Number Secondary	Police Special Youth Bureau Dog Warden FPV Specials Tactical Squad Special Number Percent <	Police Special Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Services Number Percent Number Percent	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Services I Number Percent Number Percent <td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Services Red Number Percent Number Number Percent Number Number Number Percent Number Percent Number</td> <td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Services Red Ambinities Number Percent Number</td> <td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Services Red Ambulance Number Percent Numbe</td> <td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Red Ambulance Motor Number Percent Number Number Percent Number</td> <td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Red Ambulance Motorcycles Number Percent Number Percent<td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Services Red Ambulance Motorcycles Post Number Percent Num</td></td>	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Services Red Number Percent Number Number Percent Number Number Number Percent Number Percent Number	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Services Red Ambinities Number Percent Number	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Services Red Ambulance Number Percent Numbe	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Red Ambulance Motor Number Percent Number Number Percent Number	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Red Ambulance Motorcycles Number Percent Number Percent <td>Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Services Red Ambulance Motorcycles Post Number Percent Num</td>	Police Sergeant Youth Bureau Dog Warden FPV Specials Tactical Squad Special Services Red Ambulance Motorcycles Post Number Percent Num

Continued

Table III-7 (Continued)

VICTIM'S SEX BY CAR POST SECTOR

K-9 Corps			Spe	ecial	Headqua <u>3rd Pr</u>	rter,2nd, recinct	G1	reen	Accide	nt Cars	Dete Bur	ctive cau	Clain	ns Car	BI	ue	пу	ber	
Sex	Numi	ber	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Male		1	100.0%	43	53.1%	67	69.8%	323	55.4%	26	72.2%	4	40.0%	1	100.0%	189	54.2%	219	59.3%
Female		<u>0</u>	0.0	38	46.9	<u>29</u>	30.2	260	_44.6	<u>10</u>	27.8	_6	60.0	<u>0</u>		<u>160</u>	45.8	150	40.7
Tota Miss	1s ing	l Case	100.0%	81 5698	100.0%	96	100.0%	583	100.0%	36	100.0%	10	100.0%	1	100.0%	349	100.0%	369	100.0%

VICTIM'S RACE BY CAR POST SECTOR

	Po Ser	lice geant	Youth	Bureau	Dog I	Warden	FPV S	pecials	Taci Sc	ical Juad	Spo Serv	ecial /ices		Red	Ambı	lance	Post	Number
Race	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
White	6	66.7%	6	46.2%	79	76.0%	6	5.8%	2	22.2%	0	0.0%	82	38.7%	144	56.0%	4	100.0%
Black	3	33.3	3	23.1	16	15.4	70	68.0	6	66.7	2	100.0	104	49.1	75	29.2	0	0.0
Puerto Rican	0	0,0	4	30.8	9	8.7	26	25.2	1	11.1	0	0.0	26	12.3	38	14.8	0	0.0
Other	<u>0</u>	0.0	_0	0.0	0	0.0		1.0	<u>o</u>	0.0	<u>0</u>	0.0	0	0.0	0	0.0	<u>0</u>	0.0
Totals	9	100.0%	13	100.1%	104	100.0%	103	100.0%	9	100.0%	2	100.0%	212	100.1%	257	100.0%	4	100.0%

Continued

Table III-8 (Continued)

VICTIM'S RACE BY CAR POST SECTOR

									Crim	e Code								
	K-9	Corps	Sp	ecial	Headqua 3rd Pi	rter, 2nd recinct	G	reen	Accide	ent Cars	Dete But	ective reau	Clai	ms Car	<u> </u>	lue	A	mber
Race	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
White	0	0.0%	24	34.8%	63	77.8%	383	75.1%	16	72.7%	5	71.4%	1	100.0%	217	74.1%	276	82.1%
Black	0	0.0	34	49.3	15	18.5	97	19.0	4	18.2	2	28.6	0	0.0	47	16.0	38	11.3
Puerto Rican	1 1	100.0	11	15.9	2	2.5	29	5.7	2	9.1	0	0.0	0	0.0	28	9.6	21	6.3
Other	<u>0</u>	0.0	0	0.0	<u> </u>	1.2	1	0.2	_0	0.0	<u>o</u>	0.0	<u>0</u>	0.0		0.3	_1	0.3
Totals	1	100.0%	69	100.0%	81	100.0%	510	100.0%	22	100.0%	7	100.0%	1	100.0%	293	100.0%	336	100.0%
Missin	g Cases	5984																

VICTIM'S AGE BY CAR POST SECTOR

		استدروا الرجاب وجاري ساليواني						ورسار بالاردار والم	Car Pos	st Sector							*****	
	Po Ser	lice geant	Youth	Bureau	Dog \	larden	FPV S	pecials	Tac	tical quad	Spo Serv	ecial vices	1	Red	Anib	ulance	Post	Number
Years	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
l to 10 Years	0	0.0%	0	0.0%	3	15.8%	0	0.0%	0	0.0%	0	0.0%	3	2.1%	14	5.9%	0	0.0%
ll to 18 Years	0	0.0	0	0.0	5	26.3	5	5.6	0	0.0	0	0.0	14	9.6	16	6.7	0	0.0
19 to 30 Years	1	33.3	1	100.0	6	31.6	29	32.2	1	100.0	2	100.0	38	26.0	53	22.2	1	50.0
31 to 45 Years	0	0.0	0	0.0	1	5.3	36	40.0	0	0.0	0	0.0	46	31,5	43	18.0	0	0.0
45 to 60 Years	2	66.7	0	0.0	3	15.8	14	15.6	0	0.0	0	0.0	28	19,2	56	23.4	0	0.0
6l to Highest	<u>0</u>	0.0	<u>0</u>	0.0	_1		_6	6.7	<u>0</u>	0.0	<u>o</u>	0.0	_17_	11.6	_57	23.8	1	50.0
Total	. s 3	100.0%	1	100.0%	19	100.1%	90	100.1%	1	100.0%	2	100.07	146	100.0%	239	100.0%	2	100.0%

Continued

Table III-9 (Continued)

Ð

VICTIM'S AGE BY	CAR	POST	SECTOR
-----------------	-----	------	--------

								Car Pos	t Sector							
	Special		Headquarter, 2nd 3rd Precinct		Green		Accident Cars		Detective Bureau		Claim	is Car	B.	lue	Am	ber
Years	Number	Percent	Number Percent		<u>Number</u>	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
l to 10 Years	0	0.0%	0	0.0%	9	3.2%	1	5.9%	0	0.0%	0	0.0%	3	2.0%	5	2.7%
ll to 18 Years	7	11.7	3	4.3	32	11.3	1	5.9	0	0.0	0	0.0	11	7.2	18	9.8
19 to 30 Years	27	45.0	30	43.5	91	32.3	8	47.1	4	80.0	1	100.0	49	32.0	67	36.6
31 to 45 Years	10	16.7	14	20.3	69	24.5	2	11.8	1	20.0	0	0.0	46	30.1	42	23.0
46 to 60 Years	9	15.0	14	20.3	54	19.1	3	17.6	0	0.0	0	0.0	30	19.6	31	16.9
61 to Highest	<u>_7</u>	11.7	_8	11.6	27	9.6	_2	11.8	<u>0</u>	0.0	<u>0</u>	_0.0	14	9.2	_20	<u>10.9</u>
Total	s 60	100.1%	69	100.0%	282	100.0%	17	100.1%	5	100.02	1	100.02	153	100.0%	183	99.9%
Missin	ng Cases	6744														

CAR POST SECTOR BY NEWCRIME

	M	rdor								Crime (Code									
Car Post Sector	Mansla	aughter	Ro	bbery	Assault		Burglary		Larceny LE \$50		Larcen	y GE \$50	Motor T	Vehicle heft	Diso: Co:	rderly nduct	Fa Off	mily enses	Prop Dam	erty age
Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Police Sergeant	0	0.0%	0	0.0%	1	0.3%	2	0.5%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	3	0.5%
Youth Bureau	0	0.0	0	0.0	2	0.6	0	0.0	18	5.4	7	1.6	0	0.0	1	0,3	3	1.3	3	0.5
Dog Warde	n 0	0.0	0	0.0	0	0.0	0	0.0	2	0.6	1	0.2	0	0.0	0	0.0	0	0.0	0	0.0
FPV Specials	0	0.0	5	8.5	29	8.1	7	1.8	11	3.3	9	2.0	29	3.6	14	4.1	14	5.9	16	2.7
Tactical Squad	0	0.0	1	1.7	2	0.6	1	0.3	2	0.6	0	0.0	0	0.0	12	3.5	0	0.0	0	0.0
Special Services	0	0.0	0	0.0	0	0.0	. 0	0.0	0	0.0	0	0.0	0	0.0	1	0.3	1	0.4	0	0.0
Red	0	0.0	5	8.5	42	11.7	68	17.6	36	10.9	38	8.6	136	16.9	43	12.6	52	21.8	57	9.6
Mobile Van	0	0.0	0	0.0	2	0.6	3	0.5	0	0.0	0	0.0	0	0.0	3	0.9	0	0.0	1	0.2
Park Police	0	0.0	0	0.0	1	0.3	0	0.0	0	0.0	0	0.0	0	0.0	2	0.6	0	0.0	1	0.2
Ambulance	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	ů	0.0	0	0.0
Motor- cycles	0	0.0	0	0.0	1	0.3	0	0.0	0	0.0	2	0.5	1	0.1	1	0.3	0	0.0	0	0.0
Continue	ed																			

CAR POST SECTOR BY NEWCRIME (Continued)

									Crime	Code								
a	Nuisance		A1.	arms	Motor Vehicle Violations		Medical Problems		Miscellaneous Juvenile		Other-	Person	<u>Other-P</u>	roperty	Vi	ce	No I	ncident
Sector	<u>Number</u>	Percent	Number	Percent	Number	Percent	<u>Number</u>	Percent	Number	Percent	<u>Number</u>	Percent	Number	Percent	Number	Percent	Number	Percent
Police Sergeant	1	0.1%	0	0.0%	1	0.3%	0	0.0%	3	0.4%	4	0.9%	4	1.4%	3	1.4%	0	0.0%
Youth Bureau	0	0.0	0	0.0	0	0.0	0	0.0	5	0.7	43	9.5	2	0.7 '	. 0	0.0	0	0.0
Dog Warden	502	62.0	0	0.0	0	0.0	0	0.0	2	0.3	1	0.2	0	0.0	0	0.0	0	0.0
FPV Special	12	1.5	6	1.1	4	1.2	111	13.6	46	6.5	18	4.0	5	1.8	9	4.2	6	2,6
Tactical Squad	1	0,1	0	0.0	2	0.6	0	0.0	0	0.0	15	3.3	6	2.2	48	22.5	0	0,0
Special Services	0	0.0	0	0.0	0	0.0	0	0.0	0	0,0	15	3.3	2	0.7	15	7.0	0	0.0
Red	29	3.6	100	18.2	36	10.8	28	3.4	93	13.2	45	9.9	16	5.8	9	4.2	39	16.8
Mobile Van	0	0.0	0	0.0	0	0.0	0	0.0	1	0.1	• 0	0.0	0	0.0	1	0.5	0	0.0
Park Police	0	0.0	2	0.4	0	0.0	0	0.0	1	0.1	1	0.2	0	0.0	0	0.0	0	0.0
Ambulance	1	0.1	1	0.2	0	0.0	489	59.9	4	0.6	39	8.6	75	27.0	12	5.6	38	16.4
Motorcycles	2	0.2	0	0.0	6	1.8	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0

Continued

CAR	POST	SECTOR	BY	NEWCRIME
		(Contin	ued)

A	a 1
Crime	Code

	Murder <u>Manslaughter</u>		Robbery		As:	Assault		Burglary		Larcent LE \$50		0 Larceny GE \$50		Vehicle heft	Diso: Co	rderly nduct	Fa Off	mily enses	Pro Da	perty mage
Car Post Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	<u>Percent</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Post Number	0	0.0%	1	1.7%	3	0.8%	0	0.0%	2	0.6%	3	0.7%	3	0.4%	4	1.2%	0	0.0%	1	0.2%
K-9 Corps	0	0.0	0	0.0	4	1.1	0	0.0	0	0.0	0	0.0	1	0.1	0	0.0	0	0.0	0	0.0
Special	0	0.0	0	0.0	18	5.0	14	3.6	10	3.0	23	5.2	43	5.3	10	2.9	15	6.3	28	4.7
Board of Education	1 O	0.0	0	0.0	0	0.0	2	0.5	1	0.3	0	0.0	0	0.0	0	0.0	0	0.0	2	0.3
Headquart 2nd, 3rd, Precinct	er, O	0.0	3	5.1	7	1.9	2	0.5	7	2.1	45	10.2	87	10.8	7	2.1	0	0.0	31	5.2
Green	1	100.0	19	32.2	110	30.6	127	32.9	109	32.9	117	26.5	199	24.7	92	27.1	56	23.5	204	34.2
Accident Cars	0	0.0	0	0.0	1	0.3	0	0.0	0	0.0	1	0.2	1	0.1	1	0.3	0	0.0	1	0.2
Detective Bureau	• 0	0.0	0	0.0	17	4.7	4	1.0	4	1,2	7	1.6	12	1.5	7	2.1	3	1.3	3	0.5
Claims Car	0	0.0	1	1.7	0	0.0	0	0.0	0	0.0	0	0.0	2	0.2	1	0.3	0	0.0	0	0.0
Blue	0	0.0	14	23.7	58	16.1	91	23.6	46	13.9	90	20.4	159	19.7	71	20.9	56	23.5	116	19.5
Amber	<u>0</u>	0.0	<u>10</u>	16.9	62	17.2	66	17.1	83	25.1	_98	22.2	134	16.6	70	20.6	38	16.0	<u>130</u>	_21.8
Totals	1	100.07	59	100.0%	360	100.2%	386	99.9%	331	99.9%	441	99.9%	807	100.0%	340	100.1%	238	100.0%	596	100.1%
Continu	hoi				9 **															

CAR POST SECTOR BY NEWCRIME (Continued)

			.,		Motor	Vehicle	Med	lical	Crime Miscel	<u>Code</u> laneous								
	Nuis	sance	A1	arms	Viola	tions	Prob	lems	Juve	nile	Other	-Person	Other-H	roperty	Vi	ce	No In	ncident
Car Post Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Post Number	2	0.2%	l	0.2%	0	0,0%	0	0.0%	1	0.1%	3	0.7%	5	1.8%	5	2.3%	0	0.0%
K-9 Corps	0	0.0	1	0.2	0	0.0	0	0.0	1	0.1	2	0.4	0	0.0	2	0.9	0	0.0
Special	6	0.7	4	0.7	1	0.3	31	3.8	25	3.5	15	3.3	20	7.2	5	2.3	2	0.9
Board of Education	1	0.1	3	0.5	0	0.0	0	0,0	2	0.3	0	0.0	0	0.0	0	0.0	0	0.0
Headquarter, 2nd, 3rd, Precinct	3	0.4	1	0.2	23	6.9	3	0.4	10	1.4	17	3.8	20	7.2	4	1.9	1	0.4
Green	118	14.6	174	31.8	73	21.9	85	10.4	205	29.1	87	19.2	39	14.0	29	13.6	62	26.7
Accident Car	s O	0.0	1	0.2	89	26.7	5	0.6	0	0.0	0	0.0	1	0.4	0	0.0	0	0.0
Detective Bureau	0	0.0	0	0.0	0	0.0	0	0.0	11	1.6	35	7.7	16	5.8	9	4.2	0	0.0
Claims Car	. 0	0.0	0	0.0	Ö	0.0	0	0.0	1	0.1	2	0.4	13	4.7	2	0.9	0	0.0
Blue	68	8.4	112	20.4	46	13.8	24	2.9	136	19.3	54	11.9	30	10.8	18	8.5	41	17.7
Amber	64	7.9	<u>142</u>	25.9	_52	15.6	_41	5.0	158	22.4	57	12.6	_24	8.6	42	19.7	_43	18.5
Totals	810	99.9%	548	99.8%	333	99.9%	817	100.0%	905	99.8%	453	99.9%	278	100.1%	213	99.7%	232	100.0%

GRID BY CRIME CODE

			ii						Crime	Code								
	Rob	bery	Ass	ault	Burg	lary	Larceny	LE \$50	Larceny	GE \$50	Motor Th	Vehicle neft	Disor Cor	derly duct	Far Off	nily enses	Prop Dar	perty mage
<u>Grid</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
002	1	4.8%	2	1.5%	5	4.7%	0	0.0%	1	1.0%	4	2.1%	3	3.3%	1	1.6%	4	3.3%
003	1	4.8	15	11.2	7	6.6	1	0.8	4	4.1	29	15.2	4	4.3	15	23.8	6	5.0
004	0	0.0	2	1.5	3	2.8	0	0.0	0	0.0	· 11	5.8	4	4.3	2	3.2	9	7.4
008	0	0.0	9	6.7	5	4.7	0	0.0	3	3.1	5	2.6	2	2.2	6	9.5	3	2.5
009	0	0.0	20	14.9	9	8.5	49	38.6	25	25.5	10	5.2	24	26.1	1	1.6	6	5.0
0010	2	9.5	11	8.2	7	6.6	8	6.3	11	11.2	9	4.7	3	3.3	1	1.6	10	8.3
0011	2	9.5	6	4.5	10	9.4	4	3.1	7	7.1	10	5.2	12	13.0	4	6.3	16	13.2
0015	0	0.0	1	0.7	2	1.9	1	0.8	4	4.1	1	0,5	0	0.0	0	0.0	1	0.8
0016	1	4.8	9	6.7	7	6.6	30	23.6	5	5.1	7	3.7	6	6.5	1	1.6	10	8.3
0017	6	28.6	14	10.4	9	8.5	10	7.9	10	10.2	33	17.3	7	7.6	7	11.1	19	15.7
0018	0	0.0	1	0.7	2	1.9	2	1.6	0	0.0	5	2.6	6	6.5	1	1.6	6	5.0
0 020	0	0.0	0	0.0	3	2.8	1	0.8	1	1.0	0	0.0	0	0.0	0	0.0	0	0.0
0021	0	0.0	1	0.7	5	4.7	3	2.4	2	2.0	. 0	0.0	3	3.3	0	0.0	3	2.5
0026	1	4.8	3	2.2	3	2.8	4	3.1	7	7.1	7	3.7	2	2.2	1	1.6	5	4.1
0027	0	0.0	7	5.2	6	5.7	2	1.6	3	3.1	20	10.5	2	2.2	5	7.9	6	5.0
008A	6	28.6	30	22.4	8	7.5	8	6.3	9	9.2	30	15.7	12	13.0	16	25.4	12	9.9
008B	_1	4.8	<u>3</u>	2.2	<u>15</u>	14.2	4	3.1	_6	6.1	10	5.2	_2	2.2	2	3.2	5	4.1
To	tals 21	100.2%	134	99.7%	106	99.9%	127	100.0%	98	99.9%	191	100.0%	92	100.0%	63	100.0%	121	100.1%
Co	ntinued																	

ø

Table III-11 (Continued)

GRID BY CRIME CODE

	Nuisances		Alarms		Motor Vehicle Violations		Medical Problems		Miscellaneous Juvenile		Other-Person		<u>Other-F</u>	roperty	V1	ce	No Ir	cident
<u>Grid</u>	Number	Percent	Number	Percent	Number	Percent	<u>Number</u>	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
002	3	1.8%	11	6.3%	3	3,8%	1	0.3%	5	z.4%	1	0.7%	1	1.1%	2	3.0%	2	3.4%
0003	4	2.4	l	0.6	1	1.2	37	10.3	11	5.3	9	6.3	5	5.3	1	1.5	1	1.7
0004	2	1.2	0	0.0	1	1.2	6	1.7	7	3.4	5	3.5	0	0.0	1	1,5	0	0.0
8000	3	1.8	8	4.6	1	1.2	2	0.6	6	2.9	2	1,4	4	4.2	3	4.5	6	10.2
0009	6	3.6	25	14.4	12	15.0	56	15.6	21	10.1	39	27.1	25	26.3	15	22.7	2	3.4
0010	10	5.9	20	11.5	5	6.2	5	1.4	11	5.3	2	1.4	11	11.6	3	4.5	1	1.7
0011	1.2	7.1	26	14.9	9	11.2	11	3.1	10	4.8	14	9.7	8	8.4	7	10.6	6	10.2
0015	1	0.6	3	1.7	5	6.2	9	2.5	8	3.9	5	3.5	3	3.2	1	1.5	5	8.5
0016	22	13.0	14	8.0	5	6.2	17	4.7	12	5.8	10	6.9	7	7.4	3	4.5	2	3.4
0017	12	7.1	16	9.2	8	10.0	27	7.5	24	11.6	15	10.4	10	10.5	10	15.2	9	15.3
0018	3	1.8	9	5.2	5	6.2	2	0.6	4	1.9	3	2,1	1	1.1	0	0.0	2	3.4
0020	54	32.0	2	1.1	0	0.0	0	0.0	2	1.0	0	0,0	0	0.0	1	1.5	0	0.0
0021	2	1.2	8	4.6	7	8.8	2	0.6	9	4.3	3	2.1	3	3.2	2	3.0	4	6.8
0026	2	1.2	7	4.0	7	8.8	6	1.7	7	3.4	6	4.2	. 4	4.2	1	1.5	0	0.0
0027	4	2.4	5	2.9	0	0.0	37	10.3	12	5.8	9	6.3	7	7.4	2	3.0	2	3.4
008A	23	13.6	8	4.6	7	8.8	136	37.9	50	24.2	17	11.8	6	6.3	10	15.2	7	11.9
008B	6	3.6		6.3	_4	5.0	5	1.4	8	3.9	4		_0	0.0	_4	6.1	<u>10</u>	16.9
Total	ls 169	100.3%	174	99.9%	80	99.8%	359	100.2%	207	100.0%	144	100.27	95	100.2%	66	99.8%	59	100.2%
Missi	ng Cases	5711																

97

•

 \bigcirc
.

2.

VICTIM'S SEX BY INCIDENT LOCATION TYPE

											Locati	on Type										
	H	ouse	Apar	tment	<u>llotel,</u>	Motel	Hosp <u>Conval</u>	ital, escence	Dormit Roomin	ory, g House	Resta	urant	В	ar	Mar	ket	Depa: Sto	rtment ore	Small	Store	Busi	ness
Sex	Numbe	r Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Male	124	52.8%	207	39.7%	5	100.02	3	27.3%	3	50.0%	2	50.0%	17	89.5%],	50.02	3	33.3%	10	52.6%	31	86.1%
Female	<u>111</u>	47.2	<u>314</u>	60.3	<u>0</u>	0.0	_8	72.7	<u>3</u>	50.0	<u>2</u>	50.0	_2	10.5	<u>1</u>	50.0	<u>6</u>	66.7	9	47.4	_5	13.9
Tota	18 23	100.0%	521	100.0Z	5	100.02	11	100.0%	6	100.0%	4	100.07	19	100.02	2	100.02	9	100.02	19	100.02	36	100.02

Continued

Ø

	********			·····						Locati	on Type	1								
	B	ank	Gas	Station	Car Pa	Lot, rking	Park Re	creation rea	Sport	s Arena	Th	eater	Sc Li	hool, brary	Gove	rnment lding		unah		
<u>Sex</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	<u>Percent</u>
Male	1	100.0%	8	80.0%	36	64.3%	6	50.0%	2	100.0%	1	100.0%	15	62.5%	9	75.0%	3	100.0%	242	66.5%
Female	<u>o</u>	0.0	_2	_20.0	<u>20</u>	35.7	_6	_50.0	<u>0</u>	0.0	<u>0</u>	0.0	9	37.5	_3	_25.0	<u>0</u>	0.0	122	33.5
Total	ls 1	100.0%	10	100.0%	56	100.0%	12	100.0%	2	100.0%	1	100.0%	24	100.0%	12	100.0%	3	100.0%	364	100.0%
Conti	nued																5	100.0%	204	100.0%

Table III-12 (Continued) VICTIM'S SEX BY INCIDENT LOCATION TYPE

Table III-12 (Continued)

VICTIM'S SEX BY INCIDENT LOCATION TYPE

	•								······	Locatio	on Type					<u> </u>				
	Pri Motos	lvate cycle	Priva	ite Bus	Pr T	ivate ruck	Com At	minity ito	Com <u> </u>	nunity Ixi	Com T	munity ruck	Side	ewalk	Sti	reet	Wa	ater	0	ther
Sex	Number	Percent	Number	Percent	Number	Percent	Number	Percent	<u>Number</u>	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Male	3	75.0%	2	100.0%	10	100.0%	1	190.0%	2	100.0%	5	100.0%	34	54.0%	57	62.0%	1	100.0%	39	81.3%
Female	<u>1</u>	25.0	<u>o</u>	0.0	_0	0.0	<u>0</u>	0.0	<u>0</u>	0.0	<u>0</u>	0.0	<u>29</u>	46.0	35		<u>0</u>	0.0	9	18.8
Totals	4	100.0%	2	100.0%	10	100.0%	1	100.0%	2	100.0%	5	100.0%	63	100.0%	92	100.0%	1	100.0%	48	100.1%
Missin	g Cases	6457																		



Table III-13 (Continued)

INCIDENT LOCATION TYPE, BY VICTIM'S RACE

	WI	nite	B	Lack	Puert	o Rican	0	ther
Location Type	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Community Taxi	1	0.1%	0	0.0%	0	0.0%	0	0.0%
Community Truck	4	0.5	1	0.3	0	0.0	0	0.0
Sidewalk	44	5.0	14	3.8	3	2.1	0	0.0
Street	46	5.2	20	5.4	7	4.9	0	0.0
Water	0	0.0	0	0.0	1	0.7	0	0.0
Other	28	3.2	9	2.4	7	4.9	<u>0</u>	0.0
Totals	878	100.0%	369	100.1%	143	100.1%	3	99.9%

									Locati	on Type		- <u></u>						
			•		W - 4 - 1	Mar 1	Hos	pital	Dorm	itory,	D 4		Dri	ve-In,		-		
	<u></u> <u>H</u>	ouse	Apa	rtment	Hotel	Motel	<u>Conva</u>	Lescence	Roomi	ng House	Rest	aurant	Tak	e-Out		Bar	Liquo	r Store
Car Post Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Police Sergeant	1	0.4%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%
Youth Bureau	7	2.5	5	0.8	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
FPV Specials	2	0.7	82	13.5	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Tactical Squad	1	0.4	1	0.2	0	0.0	0	0.0	0	0.0	1	4.5	0	0.0	1	2.6	0	0.0
Special Services	3	1.1	1	0.2	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Red	25	8.0	84	13.8	0	0.0	1	5.0	0	0.0	1	4.5	0	0.0	6	15.4	0	0.0
Mobile Van	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	Û	0.0	0	0.0	0	0.0
Park Police	0	0.0	0	0.0	0	0.0	· 0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Ambulance	56	19,9	82	13.5	2	22.2	1	5.0	2	25.0	0	0.0	0	0.0	1	2.6	0	0.0
Motorcycles	0	0.0	· 0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Post Number	0	0.0	1	0.2	0	0.0	0	0.0	0	0.0	1	4.5	0	Ö.0	0	0.0	0	0.0
K-9 Corps	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Special	1	0.4	32	5.3	0	0.0	0	0.0	4	50.0	0	0.0	0	0.0	0	0.0	0	0.0
Board of Education	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Headquarter, 2nd, 3rd, Precinct	2	0.7	1	0.2	0	0.0	0	0.0	1	12.5	1	4.5	0	0.0	1	2.6	0	0.0
Green	91	32.3	139	22.9	0	0.0	14	70.0	0	0.0	7	31.8	1	25,0	12	30.8	1	20.0
Accident Cars	1	0.4	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Detective Bureau	0	0.0	2	0.3	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Claims Car	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	Ó	0.0	0	0.0	0	0.0
Blue	53	18.8	99	16.3	1	11.1	3	15.0	0	0.0	4	18.2	0	0.0	11	28.2	2	40.0
Amber	39_	13.8	78	12.9	<u>6</u>	66.7	1	5.0	<u>1</u>	12.5		31.8	<u>. 3</u>	75.0	<u> </u>	17.9	<u>2</u>	40.0
Totals	282	99.4%	607	100.12	9	100.07	20	100.07	8	100.0%	22	99.8%	4	100.02	39	100.17	5	100.0%
Continued																		

		·····				*****		Locati	on Type									
	Mai	rket	Depai St	tment	Small	Store	Bus	iness	B	ank	Gas	Station	Car Pa:	Lot, rking	Park Re	creation rea	Sport	s Arena
Car Post Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Police Sergeant	0	0.0%	0	0.0%	0	0.0%	0	0.0%	υ	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%
Youth Bureau	0	0.0	1	2.5	0	0.0	1	0.5	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
FPV Specials	0	0,0	0	0.0	0	0.0	1	0.5	0	0.0	0	0.0	5	6.2	2	7.7	0	0.0
Tactical Squad	0	0.0	1	2.5	9	13.2	1	0.5	0	0,0	0	0.0	0	0.0	0	0.0	0	0.0
Special Services	0	0.0	0	0.0	3	4.4	1	0.5	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Red	2	20.0	4	10.0	6	8.8	37	19.7	2	18.2	3	11.5	4	4.9	0	0.0	0	0.0
Mobile Van	Ũ	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0,0	0	0.0	0	0.0
Park Police	0	0.0	0	0.0	1	1.5	0	0.0	0	0.0	0	0.0	0	0.0	1	3.8	0	0.0
Ambulance	1	10.0	0	0.0	0	0.0	4	2.1	0	0.0	0	0.0	1	1,2	1	3.8	0	0.0
Motorcycles	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Post Number	0	0.0	1	2,5	0	0.0	0	0.0	0	0.0	0	0.0	1	1,2	2	7.7	0	0.0
K-9 Corps	0	0.0	0	0.0	1	1.5	1	0.5	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Special	0	0.0	1	2.5	0	0.0	1	0.5	0	0.0	0	0.0	12	14.8	4	15.4	0	0.0
Board of Education	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Headquarter, 2nd, 3rd, Precinct	0	0.0	0	0.0	3	4.4	2	1.1	1	9.1	1	3.8	4	4.9	0	0.0	0	0.0
Green	4	40.0	15	37.5	14	20.6	59	31.4	2	18.2	11	42.3	17	21.0	7	26.9	2	66.7
Accident Cars	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	Ö	0.0	0	0.0
Detective Bureau	0	0.0	1	2.5	1	1.5	2	1.1	1	9.1	0	0.0	0	0.0	0	0.0	0	0.0
Claims Car	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Blue	2	20.0	4	10.0	11	16.2	44	23.4	1	9.1	7	26.9	11	13.6	3	11.5	0	0.0
Amber	1	10.0	<u>12</u>	30.0	<u>19</u>	27.9	34	18.1		36.4	4	<u>15.4</u>	<u>26</u>	32.1	<u> </u>	23.1	. 1	33.3
Totals	10	100.07	: 40	100.07	68	100.07	188	99.9%	11	100.12	26	99.92	81	99.9%	26	99.97	3	100.0%

Table III-14 (Continued) CAR POST SECTOR BY INCIDENT LOCATION TYPE

Continued

Table III-14 (Continued) CAR POST SECTOR BY INCIDENT LOCATION TYPE

								Location	i Type							
	The	eater	School	, Library	Gove Bui	rnment lding	Ch	urch	Priva	të Auto	Pr Moto	ivate rcycle	Priva	ate Bus	Priva	te Truck
Car Post Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Police Sergeant	0	0.0%	0	0.0%	0	0.0%	0	0.0%	2	0.4%	0	0.0%	0	0.0%	0	0.0%
Youth Bureau	0	0.0	0	0.0	0	0.0	0	0.0	1	0.2	0	0.0	0	0.0	0	0.0
FPV Specials	0	0.0	0	0.0	0	0.0	0	0.0	8	1.7	0	0.0	0	0.0	0	0.0
Tactical Squad	0	0.0	0	0.0	1	4.5	0	0.0	4	0.9	0	0.0	0	0.0	1	7.1
Special Services	0	0.0	0	0.0	0	0.0	0	0.0	2	0.4	0	0.0	0	0.0	0	0,0
Red	0	0.0	5	8,6	2	9.1	1	14.3	69	15.0	0	0.0	0	0.0	3	21.4
Mobile Van	0	0.0	0	0.0	0	0.0	Ö	0.0	2	0.4	0	0.0	0	0.0	0	0.0
Park Police	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Ambulance	1	33.3	5	8.6	10	45.5	1	14.3	16	3.5	0	0.0	0	0.0	0	0.0
Motorcycles	0	0.0	0	0.0	0	0.0	0	0.0	1	0.2	0	0.0	0	0.0	0	0.0
Post Number	0	0.0	0	0.0	1	4.5	0	0.0	2	0.4	0	0.0	0	0.0	0	0.0
K-9 Corps	0	0.0	0	0.0	0	0.0	Ö	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Special	0	0.0	2	3.4	0	0.0	0	0.0	12	2.6	0	0.0	1	50.0	0	0.0
Board of Education	0	0.0	6	10.3	0	0.0	0	0.0	0	(`م	0	0.0	0	0.0	0	0.0
Headquarter, 2nd, 3rd, Precinct	0	0.0	l	1.7	0	0.0	0	0.0	36	7.8	0	0.0	0	0.0	2	14.3
Green	0	0.0	18	31.0	2	9.1	1	14.3	127	27.7	3	60.0	1	50,0	1	7.1
Accident Cars	0	0.0	1	1.7	0	0.0	0	0.0	13	2.8	1	20.0	0	0.0	0	0.0
Detective Bureau	· 0	0.0	0	0.0	0	0.0	0	0.0	4	0.9	0	0.0	0	0.0	0	0.0
Claims Car	0	0.0	0	0.0	0	0.0	0	0.0	1	0.2	0	0.0	0	0.0	0	0.0
Blue	1	33.3	8	13.8	2	9.1	3	42.9	79	17.2	0	0.0	0	0.0	4	28.6
Amber	1	33.3	<u>12</u>	20.7	_4	18.2	1	14.3	80	17.4	1	20.0	<u>0</u>	0.0	3	21.4
Totals	3	99.9%	58	99.8%	22	100.0%	7	100.1%	459	99.7%	5	100.07	2	100.0%	14	99.9%

Continued

Table III-14 (Continued)

CAR POST SECTOR BY INCIDENT LOCATION TYPE

							Location	п Туре								
	Com A	munity uto	Com T	munity axi	Com	nunity Bus	Com T	munity ruck	Side	ewalk	St	reet	W	ater	0	ther
Car Post Sector	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Police Sergeant	0	0.0%	1	25.0%	0	0.0%	0	0.0%	0	0.0%	1	0.7%	0	0.0%	0	0.0%
Youth Bureau	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	` 0.0	0	0.0	1	1.5
FPV Specials	0	0.0	0	0.0	0	0.0	0	0.0	1	1.1	10	7.2	0	0.0	2	3.1
Tactical Squad	0	0.0	0	0.0	0	0.0	0	0.0	5	5.6	1	0.7	0	0.0	1	1.6
Special Services	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Red	1	25.0	1	25.0	0	0.0	0	0.0	8	8.9	18	12.9	0	0.0	. 5	7.8
Mobile Van	0	0.0	Ó	0.0	0	0.0	0	0.0	1	1.1	0	0.0	0	0.0	0	0.0
Park Police	Ó	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Ambulance	0	0.0	0	0.0	0	0.0	0	0.0	11	12.2	11	7.9	1	100.0	16	25.0
Motorcycles	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	1	0.7	0	0.0	0	0.0
Post Number	0	0.0	0	0.0	0	0.0	0	0.0	2	2.2	1	0.7	0	0.0	1	1.6
K-9 Corps	0	0.0	Ö	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	1	1.6
Special	0	0.0	0	0.0	0	0.0	0	0.0	2	2.2	0	0.0	0	0.0	3	4.7
Board of Education	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	1	1.6
Headquarter, 2nd 3rd, Precinct	0	0.0	0	0.0	0	0.0	o	0.0	1	1.1	6	4.3	0	0.0	1	1.6
Green	1	25.0	0	0.0	0	0.0	2	33.3	24	26.7	35	25.2	0	0.0	9	14.1
Accident Cars	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	1	0.7	0	0.0	0	0.0
Detective Bureau	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	1	0.7	0	0.0	0	0.0
Claims Car	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Blue	1	25.0	1	25.0	1	100.0	2	33.3	13	14.4	29	20.9	0	0.0	9	14.1
Amber	1	25.0	<u>1</u>	_25.0	<u>0</u>	0.0	<u>2</u>	33.3	22	24.4	_24	17.3	<u>0</u>	0.0	<u>14</u>	21.9
Totals	4	100.0%	4	100.0%	1	100.0%	6	99.9%	90	99.9%	139	99.9%	1	100.0%	64	100.3%
Missing Cases	, 5687															

TIME INCIDENT OCCURRED (AM-PM) BY DATE INCIDENT OCCURRED (Month)

			······			Mor	nth					
	Jar	nuary	Febr	cuary	Ma	arch	Ap	oril	1	lay	Ju	ine
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
РМ	356	66.7%	308	62.9%	281	67.7%	232	60 . 7%	249	62.3%	238	64.3%
AM	<u>173</u>	33.3	182		<u>134</u>	32.3	<u>150</u>	39.3	<u>151</u>	37.8	<u>132</u>	35.7
Totals	534	100.0%	490	100.0%	415	100.0%	382	100.0%	400	100.1%	370	100.0%
						Мог	nth					
	Jı	uly	Au	gust	Sep	tember	001	tober	Nove	ember	De.c.e	ember
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
PM	189	68.7%	227	76.7%	201	68.6%	232	60.3%	206	63.6%	141	72.7%
AM	_86	31.3	69	23.3	92	31.4	153	39.7	<u>118</u>	36.4	_53	27.3
Totals	275	100.0%	296	100.0%	293	100.0%	385	100.0%	324	100.0%	194	100.0%

TIME INCIDENT OCCURRED (3 PM/2:59 AM-3 AM/2:59 PM) BY DATE INCIDENT OCCURRED (MONTH)

.

						Mon	nth					
	Jai	nuary	Febi	uary	M	arch	Aj	pril]	lay	J	une
Time	<u>Number</u>	<u>Percent</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
3 IM to 2 AM	1 350	61.5%	332	61.7%	266	61.7%	232	58.9%	251	61.8%	241	64.1%
3 AM to 2 AM	<u>219</u>	38.5	206	38.3	<u>165</u>	38.3	162	41.1	155	38.2	135	35.9
Totals	569	100.0%	538	100.0%	431	100.0%	394	100.0%	406	100.0%	376	100.0%

						Mot	nth					
	J	uly	Au	gust	Sep	tember	<u> 0c</u> 1	tober	Nove	ember	Dec	ember
Time	Number	Percent	Number	<u>Percent</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent
3 PM to 2 AM	181	64.4%	196	61.6%	187	60.5%	233	59.3%	197	60.1%	126	63.3%
3 AM to 2 AM	1 <u>100</u>	35.6	122	38.4	122	39.5	160	<u>\40.7</u>	<u>131</u>	39.9	73	36.7
Totals	281	100.0%	318	100.0%	309	100.0%	393	100.0%	328	100.0%	199	100.0%

				the second s				Но	urs							
Month		ae]	<u>[wo</u>	Th	ree	· Fo	ur	Fi	ve	S	ix	Se	ven	E	ight
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	21	11.8%	19	15.2%	16	15.7%	6	12.2%	2	5.7%	7	9.3%	9	11.8%	17	13.2%
February	26	14.6	23	18.4	9	8.8	5	10,2	2	5.7	5	6.7	11	14.5	15	11.6
March	16	9.0	13	10.4	12	11.8	6	12.2	4	11.4	7	9,3	8	10.5	11	8.5
April	17	9.6	12	9.6	8	7.8	8	16.3	3	8.6	11	14.7	12	15.8	18	14.0
May	21	11.8	14	10.4	4	3.9	3	6.1	1	2.9	9	12.0	6	7.9	11	8.5
June	19	10.7	7	5.6	10	9,8	3	6.1	4	11.4	3	4.0	8	10.5	11	8.5
July	9	5.1	8	6.4	9	8.8	3	6.1	2	5.7	8	10.7	4	5.3	4	3.1
August	6	3.4	2	1.6	4	3.9	2	4.1	1	2.9	4	5.3	4	5.3	9	7.0
September	8	4.5	4	3,2	12	11.8	2	4.1	6	17.1	4	5.3	4	5.3	5	3.9
October	19	10.7	10	8.0	8	7.8	4	8.2	1	2.9	7	9.3	5	6.6	16	12.4
November	10	5.6	12	9.6	6	5.9	5	10.2	7	20.0	7	9.3	3	3.9	9	7.0
December	6	3.4	2	1.6	4	3.9	2	4.1	_2	5.7	3	_4.0	_2	2.6	3	3
Totals	s 178	100.2%	125	100.0%	102	99.9%	49	99.9%	35	100.0%	75	99.9%	76	100.0%	129	100.07
Continued																

DATE INCIDENT OCCURRED (MONTH) BY TIME INCIDENT OCCURRED (24 HOUR CLOCK)

					·			Нс	ours							
	<u>N1</u>	ne	T	en	E1	even	Twe	lve	Thi	teen	Four	teen	Fi	fteen	Sixt	teen
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	<u>Percent</u>	Number	Percent	Number	Percent	Number	Percent
January	21	10.9%	33	15.5%	18	9.9%	25	11.7%	37	14.6%	28	12.2%	28	11.5%	32	11.2%
February	24	12.5	32	15.0	21	11.6	28	13.1	25	9.8	29	12.6	25	10.3	45	15.7
March	15	7.8	16	7.5	17	9.4	18	8.4	27	10.6	24	10.4	25	10.3	26	9.1
April	19	9.9	16	7.5	11	6.1	19	8.9	22	8.7	15	6.5	17	7.0	23	8.0
May	19	9.9	25	11.7	12	6.6	30	14.0	14	5.5	21	9.1	20	8,2	27	9.4
June	15	7.8	13	6.1	19	10.5	20	9,3	14	5.5	15	6.5	11	4.5	25	8.7
July	17	8.9	4	1.9	11	6.1	9	4.2	17	6.7	12	5.2	15	6.2	19	6.6
August	. 9	4.7	13	6.1	17	9.4	4	1.9	29	1,1.4	26	11.3	26	10.7	23	8.0
September	12	6.3	. 16	7.5	J,S	8.3	11	5.1	17	6.7	18	7.8	23	9.5	18	6.3
October	16	8.3	23	10.8	23	12.7	22	10.3	21	8.3	14	6.1	16	6.6	16	5.6
November	17	8.9	16	7.5	12	6.6	15	7.9	. 19	7.5	15	6.5	21	8.6	20	7.0
December	8	4.2	6	2.8	5	2.8	_13	<u>_6.1</u>	_12	4.7	_13	5.7	_16	6.6	_12	4.2
Totals	192	100.1%	213	99.9%	181	100.0%	214	100.0%	254	100.0%	230	99.9%	243	100.0%	286	99.8%

Table III-17 (Continued) DATE INCIDENT OCCURRED (MONTH) BY TIME INCIDENT OCCURRED (24 HOUR CLOCK)

Continued

ſ)

Table	III-17
(Conti	nued)

DATE INCIDENT OCCURRED (MONTH) BY TIME INCIDENT OCCURRED (24 HOUR CLOCK)

				<u> </u>				Нс	ours							
	Seve	enteen	Eigh	teen	Nine	teen	Twe	enty	Twent	y-one	Twent	y-two	Twenty	-three	Twent	y-four
Month	Number	Percent	Number	Percent	Number	Percent	<u>Number</u>	Percent	Number	<u>Percent</u>	<u>Number</u>	Percent	Number	Percent	<u>Number</u>	Percent
January	37	14.5%	33	14.5%	35	13.9%	29	13.5%	35	11.6%	33	12.0%	25	11.0%	23	11.0%
February	28	11.0	19	8.4	28	11.2	20	9.3	34	11.3	30	10.9	28	12.3	26	12.4
March	24	9.4	21	9.3	15	6.0	28	13.0	30	10.0	24	8.7	20	8.8	24	11.5
April	18	7.1	14	6.2	17	6.8	1.0	4.7	32	10.6	32	11.6	11	4,8	29	13.9
Мау	20	7.8	20	8.8	18	7.2	19	8.8	26	8.6	23	8.4	29	12.8	15	7.2
June	18	7.1	22	9,7	23	9.2	24	11,2	21	7.0	36	13.1	22	9.7	13	6.2
July	13	5.1	22	9.7	24	9.6	15	7.0	13	4.3	19	6.9	17	7.5	7	3.3
August	25	9.8	17	7.5	24	9.6	7	3.3	17	5.6	15	5.5	18	7.9	16	7.7
September	19	7.5	22	9.7	12	4.8	18	8.4	22	7.3	10	3.6	14	6.2	17	8.1
October	19	7.5	13	5.7	22	8.8	22	10.2	35	11.6	24	8.7	15	6.6	22	10.5
November	18	7.1	13	5.7	21	8.4	14	6.5	30	6.6	20	7.3	20	8.8	8	3.8
December	16	6.3	<u>_11</u>	4.8	12	4.8	9	4.2	10	5.3	9	3.3	8	3.5	9	4.3
Totals	255	100.2%	227	100.0%	251	100.3%	215	100.1%	301	45.8%	275	100.0%	227	99.9%	209	99.9%

Missing Cases 3475

TIME INCIDENT OCCURRED (A.M.-P.M.) BY NEWCRIME

									Crime	Code								
	Rot	obery	Ass	ault	Burg	lary	Larceny	/ LE \$50	Larceny	GE \$50	Motor Tl	Vehicle neft	Diso: Co	rderly nduct	Fan Offe	nily enses	Prope Dam	erty age
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
рм	54	87.2%	142	68.6%	95	56.5%	142	78.9%	133	65.5%	232	54.5%	143	71.1%	105	74.5%	167	64.7%
AM	34 87,2% 5 12,3		.65	31.4	73	43.5	38	21.1	70	34.5	194	45.5	58	28.9	36	25.5	91	35.3
			<u></u>		******													
Totals	39	100.0%	207	100.0%	168	100.0%	180	100.0%	203	100.0%	426	100.0%	201	100.0%	141	100.0%	258	100.0%
Continued																		

 \mathcal{L}^{γ}

 $\mathbb{Q}^{\mathbb{Z}}$



CONTINUED 20F3

Table III-18 (Continued)

TIME INCIDENT OCCURRED (A.M.-P.M.) BY NEWCRIME

								······································		<u>Crime</u>	<u>Code</u>								·
		Nuis	sances	A1	arms	Motor Viola	Vehicle ations	Me Pro	dical blems	Misce Juv	llaneous enile	Other	-Person	Other-	Property	<u> </u>	ice	No In	cident
Time		Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
PM		264	65.8%	181	53.4%	129	69,4%	330	62.0%	296	72.5%	171	66.3%	103	66.5%	90	71.4%	107	79.3%
AM		<u>137</u>	34.2	<u>158</u>	46.6	57	30.6	202	38.0	<u>112</u>	27.5	87	33.7	52	33.5	36	28.6	28	20.7
	Totals	401	100.0%	339	100.0%	186	100.0%	532	100.0%	408	100.0%	258	100.0%	155	100.0%	126	100.0%	135	100.0%
	Missing	Cases	3654																

TIME INCIDENT OCCURRED (3 P.M./2 A.M.-3 A.M.-2 P.M.) BY NEWCRIME

	······	• • • • • • • • • • • • • • • • • • •							Crime	Code				- <u> </u>			** **********************************	
	Robbery Number Percent		Ass	ault	Burg	lary	Larceny	/ LE \$50	Larceny	GE \$50	Motor Tł	Vehicle neft	Diso: Cor	rderly nduct	Far Off	nily enses	Prop Dai	perty nage
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
3 PM to 2 AM	31	77.5%	145	69.4%	92	51.4%	118	63.8%	120	56.6%	236	54.4%	148	72.5%	104	74.3%	173	65.0%
3 AM to 2 PM	9	22.5	64	30.6	87	48.6	67	36.2		43.4	<u>198</u>	45.6	56	27.5	36	25.7	93	35.0
Totals	40	100.0%	209	100.0%	179	100.0%	185	100.0%	212	100.0%	434	100.0%	204	100.0%	140	100.0%	266	100.0%

									Crime	Code '							·	
	Nui	sances	A1a	arms	Motor Viola	Vehicle ations	Mec Prol	lical plems	Misce Juve	laneous enile	Other	-Person	Other-	Property	V	ice	No II	ncident
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
3 PM to 2 AM	262	50.8%	197	57.9%	120	64.2%	317	59.3%	266	64.9%	177	66.0%	96	60.8%	91	70.0%	102	76.1%
3 AM to 2 PM	254	49.2	143	42.1	67	35.8	218	40.7	<u>144</u>	35.1		34.0	62	39.2	39	30.0	32	23.9
Totals	516	100.0%	340	100.0%	187	100.0%	535	100.0%	410	100.0%	268	100.0%	158	100.0%	130	100.0%	134	100.0%
Missing	Cases	3470																

TIME INCIDENT OCCURRED (24 HOUR CLOCK) BY NEWCRIME

	, <u></u>	<u> </u>			· · · · · · · · · · · · · · · · · · ·		<u></u>		Crime	Code	Motor	Vehicle	Disor	derly	Fam	ily	Prop	erty
	Rob	bery	Ass	ault	Bur	glary	Larceny	LE \$50	Larceny	GE \$50	Th	eft	Con	luct	Offe	nses	Dan	age
Hours	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
1	1	2.5%	13	6.2%	4	2.2%	5	2.7%	8	3.8%	21	4.8%	6	2.9%	8	5.7%	9	3.4%
2	0	0.0	8	3.8	6	3.4	3	1.6	2	0.9	13	3.0	10	4.9	2	1.4	9	3.4
3	1	2.5	9	4.3	9	5.0	1	0.5	7	3.3	12	2.8	4	2.0	1	0.7	7	2.6
4	0	0.0	2	1.0	5	2.8	0	0.0	3	1.4	4	0.9	2	1.0	2	1.4	5	1.9
5	. 0	0.0	1	0.5	4	2.2	1	0.5	4	1.9	2	0.5	0	0.0	2	1.4	. 0	0.0
6	0	0.0	3	1.4	4	2.2	2	1.1	4	1.9	13	3.0	1	0.5	4	2.9	2	0.8
7	0	0.0	0	0.0	8	4.5	2	1.1	5	2.4	12	2.8	2	1.0	0	0.0	3	1.1
8	1	2.5	2	1.0	9	5.0	2	1.1	9	4.2	21	4.8	2	1.0	1	0.7	9	3.4
9	0	0.0	1	0.5	6	3.4	3	1.6	4	1.9	27	6.2	2	1.0	3	2.1	14	5.3
10	0	0.0	7	3.3	8	4.5	9	4.9	6	2.8	25	5.8	6	2.9	1	0.7	12	4.5
11	1	2.5	8	3.8	7	3.9	6	3,2	12	5.7	19	4.4	8	3.9	1	0.7	5	1.9
12	2	5.0	11	5.3	6	3.4	7	3.8	12	5.7	32	7.4	16	7.8	11	7.9	17	6.4
13	3	7.5	13	6.2	7	3.9	17	9,2	12	5.7	12	2,8 °	7	3.4	4	2.9	9	3.4
14	. 1	2.5	7	3.3	14	7.8	17	9.2	14	6.6	19	4.4	6	2.9	6	4.3	10	3.8
15	4	10.0	13	6.2	4	2.2	11	5.9	17	8.0	16	3.7	16	7.8	5	3.6	. 7	2.6
16	3	7.5	7	3.3	7	3.9	20	10.8	16	7.5	23	5.3	16	7.8	12	8.6	14	5.3
17	1	2.5	12	5.7	12	6.7	14	7.6	8	3.8	24	5.5	7	3.4	11	7.9	21	7.9
18	4	10.0	17	8.1	4	2.2	11	5,9	14	6.6	18	4.1	16	7.8	5	3.6	14	5.3
19	2	5.0	<u>i</u> (6.7	12	6.7	15	8.1	5	2.4	16	3.7	15	7.4	9	6.4	13	4.9
20	4	10.0	12	5.7	6	3.4	6	3.2	10	4.7	10	2.3	13	6.4	6	4.3	24	9.0
21	6	15.0	10	4.8	14	7.8	8	4.3	17	8.0	28	6.5	18	8.8	15	10.7	25	9.4
22	3	7.5	15	7.2	6	3.4	7	3.8	7	3.3	26	6.0	20	9.8	13	9.3	14	5.3
23	3	7.5	17	8.1	8	4.5	4	2.2	4	1.9	25	5.8	7	3.4	13	9.3	11	4.1
24		0.0	7	3.3	9	5.0	14	7.6		5.7	_16	3.7	4	2.0	5	3.6	12	4.5
Tota	als 40	100.0%	209	99.7%	179	100.0%	185	99.9%	212	100.1%	434	100.2%	204	99.8%	140	99.8%	265	100.2%

Continued

Table III-20 (Continued)

TIME INCIDENT OCCURRED (24 HOUR CLOCK) BY NEWCRIME

									Crime	Code								
	Nuis	ances	Alaı	:ms	Motor Viola	Vehicle tions	Med Prob	ical lems	Miscel Juve	laneous nile	Other-	Person	Other-P	roperty	Vi	ce	No In	cident
Hours	Number	Percent	Number H	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
1	5	1.0%	17	5.0%	4	2.1%	23	4.3%	13	3.2%	19	7.1%	9	5.7%	10	7.7%	4	3,0%
2	9	1.7	14	4.1	6	3.2	17	3.2	6	1.5	7	2.6	5	3.2	5	3.8	3	2.2
3	8	1.6	11	3.2	0	0.0	11	2.1	3	0.7	8	3.0	6	3.8	2	1.5	2	1.5
4	0	0.0	6	1.8	1	0.5	9	1.7	4	1.0	5	1.9	0	0.0	0	0.0	1	0.7
5	2	0.4	10	2.9	0	0.0	8	1.5	1	0.2	0	0.0	0	0.0	0	0.0	0	0.0
6	1	0.2	16	4.7	6	3,2	9	1.7	3	0.7	3	1.1	2	1.3	2	1.5	0	0.0
7	3	0.6	14	4.1	1	0.5	17	3.2	3	0.7	7	1.1	3	1.9	0	0.0	0	0.0
8	17	3.3	8	2.4	5	2.7	21	3.9	7	1.7	6	2.2	7	4.4	1	0.8	1	0
9	53	10.3	14	A.1	8	4.3	19	3.6	18	4.4	8	3.0	5	3.2	4	3.1	3	2.2
10	34	6.6	11	3.2	11	5.9	29	5.4	26	6.3	16	6.0	5	3.2	3	2.3	4	3,0
11	37	7.2	10	2.9	5	2.7	26	4.9	18	4.4	7	2.6	7	4.4	2	1.5	2	1.5
12	6	1.2	28	8.2	10	5.3	15	2.8	11	2.7	10	3.7	4	2.5	9	6.9	7	5.2
13	53	10.3	7	2.1	14	7.5	27	5.0	24	5.9	13	4.9	15	9.5	10	7.7	8	6.0
14	40	7.8	8	2.4	6	3.2	27	5.0	26	6.3	12	4.5	8	5.1	6	4.6	4	3.0
15	47	9.1	14	4.1	10	5.3	18	3.4	22	5.4	19	7.1	9	5.7	5	3.8	6	4.5
16	29	5.6	14	4.1	13	7.0	30	5.6	36	8.8	13	4.9	13	8.2	7	5.4	13	9.7
17	23	4.5	16	4.7	5	2.7	33	6.2 -	33	8.0	16	6.0	5	3.2	6	4.6	8	6.0
18	14	2.7	20	5.9	9	4,8	31	5.8	23	5.6	10	3.7	5	3.2	3	2.3	9	6.7
19	23	4.5	16	4.7	, 10	5.3	37	6.9	32	7.8	12	4.5	7	4.4	6	4.6	7	5.2
20	13	2.5	20	5.9	11	5.9	22	4.1	25	6.1	12	4.5	6	3.8	7	5.4	8	6.0
21	20	3.9	17	5.0	17	9.1	29	5.4	25	6.1	16	6.0	11	7.0	14	10.8	11	8.2
22	28	5.4	22	6.5	14	7.5	31	5.8	20	4.9	19	7.1	10	6.3	10	7.7	11	8.2
23	16	3.1	17	5.0	13	7.0	22	4.1	10	2,4	24	9.0	7	4.4	10	7.7	16	11.9
24	35	6.8	10	2.9	8	4.3	24	4.5		5.1	10	3.7	9	5.7	8	6.2	6	4.5
Tot	als 516	100.3%	340	99.9%	187	100.0%	535	100.1%	410	99.9%	268	100.2%	158	100.1%	130	99.9%	134	99.9%
								100										

Missing Cases 3470

115

 $\langle \hat{\gamma} \rangle$

DATE INCIDENT OCCURRED (MONTH) BY NEWCRIME

	<u></u>		···							Crime (Code	•••••								·
	Mur Mansla	rder- aughter	Rol	obery	As	sault	Burg	glary	Larcen	/ LE \$50	Larceny	<u>GE \$50</u>	Motor T	Vehicle heft	Diso Co	rderly nduct	Far Offe	nily enses	Ргор Dan	erty age
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	. 1	100.0%	10	16.9%	27	7.6%	38	10.3%	33	10.3%	16	3.8%	83	10.7%	23	6.9%	20	8.6%	55	9.7%
February	0	0.0	2	3.4	14	4.0	24	6.3	23	7.2	36	8.5	65	8.4	26	7.8	22	9.4	60	10.5
March	0	0.0	5	8.5	35	9.9	34	8.9	24	7.5	23	5.4	63	8.1	32	9.6	28	12.0	45	7.9
April	0	0.0	3	5.1	18	5,1	26	6.8	28	8.8	32	7.5	57	7.3	32	9.6	22	9,4	56	9,8
May	0	0.0	4	6.8	35	9.9	30	7.9	26	8.1	38	8.9	70	9.0	33	9,9	21	9.0	45	7.9
June	0	0.0	4	6.8	38	10.8	33	8.7	27	8.4	37	8.7	58	7.5	32	9.6	20	8.6	43	7.6
July	0	0.0	6	10.2	35	9.9	26	6.8	17	5.3	43	10.1	46	5.9	38	11.4	15	6.4	37	6.5
August	0	0.0	3	5.1	32	9.1	25	6.6	48	15.0	44	10.4	56	7.2	.21	6.3	18	7,7	49	8.6
September	0	0.0	5	8.5	33	9.3	34	8.9	31	9.7	49	11.5	54	7.0	20	6.0	19	8.2	42	7.4
October	0	0.0	4	6.8	29	8.2	28	7.4	24	7.5	37	8.7	77	9,9	35	10.5	16	6.9	57	10.0
November	0	0.0	9	15.3	33	9.3	47	12.4	25	7.8	36	8.5	64	8.2	24	7,2	12	5.2	43	7.6
December	<u>o</u>	0.0	4	6.8	24	6.8	35	9.2	14	4.4	34	8.0	83	<u>10.7</u>		5.1	20	. 8.6		6.5
Totals	1	100.0%	59	100.2%	353	99.9%	380	99.9%	320	100.0%	425	100.0%	776	99,9%	333	99.9%	233	100.0%	569	100.0%

Continued

Table III-21 (Continued)

DATE INCIDENT OCCURRED (MONTH) BY NEWCRIME

									Crime	<u>Code</u>								
	<u> </u>	sances	A1	arms	Motor Viol	Vehicle ations	Me Pro	dical blems	Miscel Juve	llaneous enile	Other	Person	Other-I	roperty	<u> </u>	ice	No I	ncident_
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	68	8.7%	44	8.0%	33	10.2%	76	9、3%	69	10.0%	46	10.5%	26	9.6%	21	9.9%	28	12.3%
February	77	9.8	58	10.6	30	9.2	64	7.9	42	6.1	47	10.7	27	9.9	16	7,5	13	5.7
March	88	11.2	47	8.6	31	9.5	57	7.0	58	8.4	40	9.1	16	5.9	24	11.3	17	7.5
April	62	7.9	40	7.3	26	8,0	73	9.0	72	10.5	27	6.2	23	8.5	22	10.3	25	11.0
May	71	9.1	44	8.0	24	7.4	78	9.6	81	11.8	38	8.7	31	11.4	17	8.0	32	14.0
June	73	9.3	47	8.6	33	10.2	81	9.9	74	10.8	49	11.2	21	7.7	15	7.0	29	12.7
July	69	8.8	47	8.6	31	9.5	73	9.0	59	8.6	31	7.1	22	8.1	19	8.9	24	10.5
August	80	10.2	57	10,4	23	7.1	68	8.3	64	9.3	35	8.0	27	9.9	16	7.5	17	7.5
September	59	7,5	41	7.5	20	6.2	63	7.7	45	6.5	27	6.2	16	5.9	20	9.4	5	2.2
October	53	6.8	33	6.0	28	8.6	64	7.9	48	7.0	28	6.4	27	9.9	15	7.0	8	3.5
November	49	6.3	50	9.1	24	7.4	58	7.1	41	6.0	30	6.8	27	9.9	16	7.5	16	7,0
December	35_	4.5	41	7.5	22	6.8	60	7.4	35		41	9.3	9	3.3	12	5.6	<u> 14 </u>	6.1
Totals	784	100.1%	549	100.2%	325	100.1%	815	100.1%	688	100.1%	439	100.2%	272	100.0%	213	99.9%	228	100.0%
Missing	Cases	638																

TIME INCIDENT OCCURRED (A.M.-P.M.) BY CAR POST SECTOR

					<u> </u>					<u> </u>	ar Pos	t Sector	r									
-	Pol	ice geant	Yo Bur	uth eau	Dog W	arden	FPV Sp	ecials	Tact Sq	ical uad	Spe Serv	cial ices	R	ed	Mobil	e Van	Park	Police	Атьц	lance	Motor	cycles
Time M	lumber	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
PM	2	25.0%	35	79.5%	150	66.4%	147	71.7%	32	58.2%	14	63,6%	301	61.4%	8	100.0%	2	66.7%	266	61.4%	9	100.0%
AM	<u>6</u>	75.0	_9	20.5	76	33.6	58	28.3	23	41.8	8	36.4	189	38.6	<u>0</u>	0.0	1	33.3	167	38.6	<u>0</u>	0.0
Tota	n1s 8	100.0%	44	100.0%	226	100.0%	205	100.0%	55	100.0%	22	100.0%	490	100.0%	8	100.0%	3	100.0%	433	100.0%	9	100.0%

						······································				<u> </u>	ar Pos	t Sector	<u>.</u>									
	Post	Number	K-9	Corps	Spe	cial	Boar Educ	d of dation	Headqua 3rd Pr	rter, 211, ecinct	Gr	een	Accider	nt Cars	Dete Bur	ctive eau	Claim	s Car	B1	ue	Am	ber
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
PM	14	70.0%	7	77.8%	116	77.3%	8	100.0%	81	66.4%	664	64.0%	52	94.5%	37	82.2%	7	58.3%	431	67.4%	466	63.3%
AM	_6	30.0	2	22.2	34	22.7	<u>0</u>	0.0		33.6	374	36.0		5.5	8	17.8	_5	41.7	208	32.6	270	36.7
Ъt	als 20	100.0%	9	100.0%	150	100.0%	8	100.0%	122	100.0%	1038	100.0%	55	100.0%	45	100.0%	12	100.0%	639	100.0%	736	100.0%
Mis	sing C	ases	3680																			

118

									Gr	id				******				
*		002		003 `		004		008	,	009	0	010	0	01]	0	015	0	016
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
3 PM to 2 AM	23	82.1%	59	75.6%	26	78.8%	22	59.5%	139	64.7%	44	62.0%	57	63.3%	12	46.2%	59	59.0%
3 AM to 2 PM	_5	17.9	<u>19</u>	24.4		21.2	<u>15</u>	40.5	76	35.3	<u>27</u>	38.0	<u>33</u>	36.7	14	53.8		41.0
Totals	28	100.0%	78	100.0%	33	100.0%	37	100.0%	215	100.0%	71	100.0%	90	100.0%	26	100.0%	100	100.0%
								منحصير من	Gr	id								
	0	017		0018		0020		002	1	00	26		0027		008A		008	В
Time	Number	Percent	Numb	er Percef	t Nu	mber Perc	ent	Number P	ercent	Number.	Percent	Numbe	er Perce	nt Nur	nber Per	cent	Number P	ercent
3 PM to 2 AM	90	63.4%	2	0 66.7	18	11 29	0.7%	25	78.1%	21	52.5%	4	5 59.	2%	157 6	5.7%	34	63.0%
3 AM to 2 PM	52	36.6	1	0 33.3	5	<u>26 7(</u>	0.3	7	21.9	<u>19</u>	47.5	3	<u>40.</u>	8	82 3	4.3	20	37.0
Totals	142	100.0%	3	io 100.0)%	37 100).0%	32	100.0%	40	100.0%	5 7 (5 100.	0%	239 10	0.0%	54	100.0%
missing	uases	6689																

TIME INCIDENT OCCURRED (3 P.M./2 A.M.-3 A.M./2 P.M.) BY GRID

									Locatio	on Type								
	H	ouse	Apa	rtment	Hotel,	Motel	Ho sj Conva	pital, lescence	Dormi Roomii	tory ng House	Resta	urant	Driv Take	ve-In, e-Out	I	Bar	Liquo	r Store
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
РМ	175	72.0%	354	68.2%	6	75.0%	8	50.0%	6	100.0%	11	52,4%	3	75.0%	23	65.7%	4	100.0%
AM	68	28.0	165	31.8	2	25.0	8	50.0	<u>0</u>	0.0	<u>10</u>	47.6	<u>1</u>	25.0	<u>12</u>	34.3	<u>0</u>	0.0
Tota	ls 243	100.0%	519	100.0%	8	100.0%	16	100.0%	6	100.0%	21	100.0%	4	100.0%	35	100.0%	4	100.0%
				•					Lodati	on /Type			1					
	Ma	rket	Depa S	rtment tore	Small	l Store	Bus	iness	В	ank	Gas_S	Station	Car Pa	Lot, rking	Recreat	nrk <u>ion Area</u>	Spor	ts Arena
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
РМ	7	87.5%	29	78.4%	36	62.1%	70	45.8%	4	44.4%	14	70.0%	38	66.7%	19	82.6%	1	100.0%
AM	<u>1</u>	12.5	_8	21.6	22	37.9	83	54.2	5	55.6	6	30.0	19	33.3	4	17.4	<u>0</u>	
Tota	ls 8	100.0%	37	100.0%	58	100.0%	153	100.0%	9	100.0%	20	100.0%	57	100.0%	23	100.0%	1	100.0%

TIME INCIDENT OCCURRED (A.M.-P.M.) BY INCIDENT LOCATION TYPE

Continued

Table	III-24
(Cont	inued)

TIME INCIDENT OCCURRED (A.M. P.M.) BY INCIDENT LOCATION TYPE

					<u> </u>			Location	Туре							
	The	ater	School,	Library	Gover Buil	nment ding	Cł	urch	Privat	e Auto	Pri Motor	vate cycle	Priva	te Bus	Privat	e Truck
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
РМ	3	100.0%	23	52.3%	12	57.1%	7	100.0%	216	59.3%	3	60.0%	2	100.0%	5	62.5%
АМ	<u>0</u>	0.0	21	47.7	9	42.9	<u>0</u>	0.0	148	40.7	2	40.0	<u>0</u>	0.0	<u>3</u>	37.5
Total	s 3	100.0%	44	100.0%	21	100.0%	7	100.0%	364	100.0%	5	100.0%	2	100.0%	8	100.0%
								Location	Туре							
	Communi	ty Auto	Communi	ty Taxi	Commun	ity Bus	Communi	ity Truck	Side	ewalk	St	reet	Wa	iter	01	ther
Time	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
РМ	1	33.3%	2	50.0%	0	0.0%	O	0.0%	54	67.5%	88	75.9%	1	100.0%	41	67.2%
AM	2	66.7	<u>2</u>	50.0	1	100.0%	<u>2</u>	<u>100.0%</u>	26	32.5	_28	_24.1	<u>0</u>	0.0	20	32.8
Total	s 3	100.0%	4	100.0%	1	100.0%	2	100.0%	80	100.0%	116	100.0%	1	100.0%	61	100.0%
Missi	ng Case	s 6073														

	-							Car Pos	t Sector							
	Pol Serg	ice eant	Youth	Bureau	Dog W	arden	FPV Sp	ecials	Tact Sc	ical	Spe Serv	cial ices		Red	Mobi	le Van
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	1	5.6%	5	6.6%	48	9.8%	26	7.4%	10	11.6%	1	3.2%	76	8.9%	. 0	0.0%
February	6	33.3	5	6,6	48	9.8	31	8.9	4	4.7	4	12.9	62	7.3	0	0.0
March	0	0.0	6	7.9	56	11.4	22	6.3	4	4.7	5	16.1	77	9.1	1	10.0
April	4	22.2	4	5.3	37	7.5	28	8.0	4	4.7	2	6.5	53	6,2	0	0.0
May	0	0.0	8	10.5	. 38	7.7	47	13.4	12	14.0	1	3.2	74	8.7	1	10.0
June	3	16.7	8	10.5	47	9.6	28	8.0	10	11.6	7	22.6	74	8.7	1	10.0
July	0	0.0	4	5.3	46	9.3	37	10.6	7	8.1	1	3.2	75	8.8	0	0.0
August	1	5.6	11	14.5	43	8.7	30	8.6	7	8,1	2	6.5	71	8,4	1	10.0
September	0	0.0	7	9.2	40	8.1	33	9.4	9	10.5	2	6.5	73	8.6	. 2	20.0
October	1	5.6	. 9	11.8	31	6.3	26	7.4	10	11.6	3	9.7	73	8.6	2	20.0
November	1	5.6	6	7.9	38	7.7	26	7.4	7	8.1	1	3.2	79	9.3	0	0.0
December	_1	5.6		3.9	_20	4.1	_16	4.6		2.3	_2	6.5	63		_2	20.0
Totals	18	100.2%	76	100.0%	492	100.0%	350	100.0%	86	100.0%	31	100.1%	850	100.0%	10	100.0%

DATE INCIDENT OCCURRED (MONTH) BY CAR POST SECTOR

Continued

							Car Post	Sector	•	- <i>4</i>				
	Park	Police	Ambu	lance	Motor	cycles	Post_	Number	<u> </u>	Corps	Spe	cial	Boa Edu	rd of cation
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	0	0.0%	58	8.9%	0	0.0%	1	3.0%	1	9.1%	12	4.7%	1	11.1%
February	2	28.6	50	7.7	0	0.0	4	12.1	i	9.1	22	8.7	2	22.2
March	0	0.0	53	8.1	1	8.3	2	6.1	0	0.0	19	7.5	0	0.0
April	0	0.0	54	8.3	0	0.0	5	15,2	0	0.0	25	9.9	2	22.2
May	1	14.3	60	9.2	0	0.0	3	9.1	0	0.0	12	4.7	1	11.1
June	1	14.3	60	9.2	2	16.7	0	0.0	2	18.2	25	9.9	1	11.1
July	2	28.6	54	8.3	2	16.7	3	9.1	0	0.0	21	8.3	1	11.1
August	0	0.0	58	8.9	1	8.3	5	15.2	1	9.1	19	7.5	0	0.0
September	0	0.0	47	7.2	1	8.3	1	3.0	-0	0,0	16	6.3	0	0.0
October	0	0.0	59	9.0	2	16.7	4	12.1	3	27.3	29	11.5	, 1	11.1
November	1	14.3	49	7.5	1	8.3	1	3.0	2	18.2	33	13.0	0	0.0
December	<u>0</u>	0.0	50	7.7		16.7	_4	12.1	_1	9.1	20	7.9	<u>0</u>	0.0
Totals	7	100.1%	652	100.0%	12	100.0%	33	100.0%	11	100.1%	253	99.9%	9	99.9%

Table III-25 (Continued) DATE INCIDENT OCCURRED (MONTH) BY CAR POST SECTOR

Continued

	to a state of the						Car Post	Sector					والمستقام مستعينات	
	Headquar 3rd, P	ters, 2d, recinct	Gr	een	Accido	ent Cars	Dete Bur	ective reau	<u>Claim</u>	s Car	B1	ue	Ar	nber
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	28	10.9%	163	8.9%	11	11.1%	2	1,9%	0	0.0%	140	11.7%	128	9.9"
February	31	12,1	152	8.3	7	7.1	2	1.9	0	0.0	97	8.1	113	8.7
March	23	9.0	159	8.6	9	9.1	11	10.2	1	5.0	102	8.5	110	8.5
April	16	6.3	172	9.4	8	8,1	11	10.2	1	5.0	94	7.9	121	9.4
Мау	17	6.6	162	8.8	4	4.0	10	9.3	0	0.0	125	10.5	135	10.4
June	19	7.4	181	9.8	8	8,1	10	9.3	0	0.0	108	9.0	110	8.5
July	14	5.5	152	8.3	11	11.1	12	11.1	0	0.0	96	8.0	92	7.1
August	22	8.6	· 159	8.6	7	7.1	12	11.1	0	0.0	101	8.5	127	9.8
September	20	7.8	149	8.1	7	7.1	7	6.5	0	0.0	73	6.1	93	7.2
October	18	7.0	143	7.8	11	11.1	10	9,3	5	25,0	83	6.9	85	6.0
November	30	11.7	118	6.4	. 10	10.1	12	11,1	9	45.0	80	6.7	94	7.3
December	18	7.0	129	7.0	_6	6.1	9	8.3	4	20.0	96	8.0	86	<u>. 0.6</u>
Totals	256	99.9%	1839	100.0%	99	100.0%	108	100.2%	20	100.0%	1195	99.9%	1294	100.0%
Missing	Cases	376												

		Table 1	11-25				
		(Conti:	nued)				
DATE	INCIDENT	OCCURRED	(MONTH)	BY	CAR	POST	SECTOR

10010 277	Tab	1e	III-	26
-----------	-----	----	------	----

DATE INCIDENT OCCURRED (MONTH) BY GRID

									G	rid								*****
	(002	(003		004		008	(009	0	010	0	011	0	015	0	016
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	4	8.2%	13	9.0%	4	7.8%	6	9.0%	26	8.0%	14	11.2%	18	10.6%	4	8.5%	17	10.4%
February	6	12.2	7	4.9	5	9.8	4	6.0	23	7.1	8	6.4	10	5,9	8	17.0	14	8.5
March	5	10.2	11	7.6	6	11.8	7	10.4	21	6.5	12	9.6	16	9.4	5	10.6	14	8.5
April	2	4.1	13	9,0	3	5.9	4	6.0	26	8.0	13	10.4	12	7.1	4	10.6	12	7,3
May	3	6.1	6	4.2	5	9.8	8	11.9	29	8.9	10	8.0	14	8.2	2	4.3	11	6.7
June	3	6.1	13	9.0	2	3.9	4	6.0	29	8.9	14	11.2	21	12.4	0	0.0	17	10,4
July	6	12.2	14	9,7	2	3.9	8	11.9	13	4.0	5	4.0	14	8.2	0	0.0	15	9,1
August	1	2.0	9	6.3	6	11.8	5	7.5	41	12.6	13	10.4	12	7.1	2	4.3	16	9.8
September	5	10.2	9	6.3	3	5.9	4	6.0	29	8.9	6	4.8	15	8.8	3	6.4	13	7.9
October	5	10.2	18	12.5	7	13.7	7	10.4	33	10.2	12	9.6	10	5.9	10	21.3	11	6,7
November ·	5	10.2	18	12.5	6	11.8	8	11.9	27	8.3	13	10.4	12	7.1	4	8,5	12	7.3
December	_4	8.2	<u>13</u> ,	9.0	_2	3.9	_2		28	8.6	5	4.0	_16	9.4	_4	8.5	12	7.3
Totals	s 49	99.7%	144	100.0%	51	100.0%	67	100.0%	325	100.0%	125	100.0%	170	100.1%	47	00.0%	164	99.9%

Continued

Table III-26 (Continued)

DATE INCIDENT OCCURRED (MONTH) BY GRID

								Gr	id				·			
	0	017	0	018	0	020	0	021	0	026	0)27	0)8A	<u> </u>	08 B
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	19	7.9%	4	7.8%	4	6.6%	5	8,9%	6	8.3%	8	6.6 [%]	24	ó.1%	12	12.45
February	20	8.4	1	2.0	2	3.3	7	12.5	3	4.2	11	9.0	37	9.5	6	6.2
March	22	9.2	6	11.8	0	0.0	3	5.4	9	12.5	11	9.0	28	7.2	11	11,3
April	27	11.3	5	9.8	2	3.3	4	7.1	12	16.7	9	7.4	31	7.9	5	5,2
May	21	8.8	3	5.9	2	3.3	5	8.9	9	12.5	13	10.7	45	11.5	6	6.2
June	18	7.5	3	5.9	6	9.8	5	8.9	4	5.6	11	9.0	37	9.5	8	3.2
July	19	7.9	6	11.8	9	14.8	7	12.5	5	4.2	19	15.6	35	9.0	7	7,2
August	17	7,1	7	13.7	7	11.5	3	5.4	5	6.9	7	5.7	32	8.2	10	10.3
September	20	8.4	3	5.9	6	9.8	4	7.1	2	2.8	11	9.0	34	8.7	7	7.2
October	24	10.0	4	7.8	7	11.5	3	5.4	7	9.7	9	7.4	30	7.7	10	10.3
November	12	5.0	4	7.8	7	11.5	7	12.5	8	11.1	7	5.7	35	9.0	7	7.2
December	_20	8.4	_5	9.8	_9	14.8	3	5.4	_4	5.6	6	4.9	_23	5.9	_8	3.2
Totals	239	99.9%	51	100.0%	61	100.2%	56	100.0%	72	100.1%	122	100.0%	391	100.2%	97	99,9%
Missing	, Cases	5786														

DATE INCIDENT OCCURRED	(MONTH)	BY	INCIDENT	LOCATION	TVDE
------------------------	---------	----	----------	----------	------

									Locat	ion Type																
	House		House		Apartment		Apartment		Apartment		Apartment		Hote1	, Motel	Hos Conva	pital, lescence	Dorm Roomi	itory, ng House	Rest	aurant	Dri Tak	ve-In, e-Out				
Month	Number	Percent	Number	Percent	Number	Percent	Number Percent		Numbon Deserved		N			Tuxe-out		Dar		Liquor Store								
January	67	24.0%	125	20.6%	1	12 5%			MUNDEL	rercent	Number	Percent	Number	Percent	Number	Percent	Number	Percent								
February	59	21 1	114	10.0		16.33	4	20.0%	0	0,0%	5	22.7%	1	25.0%	6	15.4%	2	40.0%								
Mamah			114	18.8	1	12.5	6	30.0	1	12.5	4	18.2	0	0.0	8	20.5	0	0 0								
March	25	9.0	97	16.0	2	25.0	2	10.0	2	25.0	1	4.5	,	25.0				0.0								
April	25	9.0	69	11.4	1	12.5	3	15.0	z	77 F			1	23.0	/	17.9	1	20.0								
May	24	8.6	77	12.7	,	12 5				37.5	4	18.2	1	25.0	4	10.3	1	20.0								
June	36	ם כו	50		•	12.5	3	15.0	1	12.5	0	0.0	1	25.0	5	12.8	1	20.0								
7		12,5	29	9.7	1	12.5	2	10.0	.0	0.0	4	18.2	0	0.0	3	77	0	0.0								
JULY	33	11.8	52	8.6	1	12.5	0	0.0	0	0.0	4	18 2	0	7.0			0	0.0								
August	10	3.6	12	2.0	0	0.0	0	0.0	,	10 -		10.2	U	0.0	6	15.4	0	0.0								
November	0	0.0	n	0 0	0	• •		0.0	1	12.5	0	0.0	0	0.0	0	0.0	0	0.0								
December	0	0.0	•	0.0	Ū	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0								
December		0.0	2	0.3	<u>0</u>	0.0	_0	0.0	<u>0</u>	0.0	0	0.0	0	0.0	0	0.0	-									
Totals	270	100.0%	607	100 10								No. of Concession, Name	-				<u>0</u>									
	2,5	100.05	007	100.1%	8	100.0%	20	100.0%	. 8	100.0%	22	100.0%	4	100.0%	39	100.0%	Ę	100 0%								
																	5	100.0%								

Continued

	Location Type																	
	Market		Depa:	Department Store		1 Store	Bus	iness	Ba	ink	Gas (Station	Car Parl	Lot, ting	Pa <u>Recreat</u>	irk ion Area	Sport	s Arena
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	0	0.0%	6	15.0%	22	31.9%	50	27.2%	3	27.3%	4	15.4%	19	25.0%	1	4.0%	1	33.3%
February	4	50.0	9	22.5	14	20.3	36	19.6	4	36.4	5	19.2	20	26.3	2	8.0	0	0.0
March	0	0.0	4	10.0	9	13.0	25	13.6	0	0.0	2	7.7	10	13.2	2	8.0	0	0.0
April	0	0.0	5	12.5	10	14.5	23	12.5	0	0.0	4	15.4	9	11.8	6	24.0	1	33.3
May	2	25.0	9	22.5	6	8.7	22	12.0	1	9.1	6	23.1	3	3.9	8	32.0	0	0.0
June	1	12,5	4	10.0	6	8.7	15	8.2	1	9.1	3	11.5	5	6.6	1	4.0	0	0.0
July	1	12.5	2	5.0	2	2.9	9	4.9	1	9.1	0	0.0	9	11.8	5	20.0	1	33.3
August	0	0.0	1	2.5	0	0.0	2	1.1	0	0.0	1	3.8	1	1.3	0	0.0	0	0.0
November	0	0.0	0	0.0	0	0.0	2	1,1	1	9.1	0	0.0	0	0.0	0	0.0	0	0.0
December	<u>0</u>	0.0	_0	0.0		0.0	0	0.0	_0	0.0	<u> </u>	3.8	_0	0.0	_0	0.0	<u>0</u>	0.0
Totals	s 8	100.0%	40	100.0%	69	100.0%	184	100.2%	11	100.1%	26	99,9%	76	99.9%	25	100.0%	3	99.9%

Table III-27 (Continued) DATE INCIDENT OCCURRED (MONTH) BY INCIDENT LOCATION TYPE

Continued

	Location Type														ii	
	Theater Number Percent		School, Library Number Percent		Gover Buil	vernment uilding Church Privat		e Auto	Private e Auto Motorcycle			ate Bus	Private Truck			
Month					Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	2	66.7%	10	18.2%	5	22.7%	0	0.0%	106	23.8%	0	0.0%	0	0.0%	5	35.7%
February	0	0.0	17	30.9	5	22.7	1	14.3	104	23.3	0	0.0	2	100.0	2	14.3
March	0	0.0	6	10.9	4	18.2	1	14.3	55	12.3	0	0.0	0	0.0	2	14.3
April	1	33.3	3	5.5	3	13.6	0	0.0	44	9.9	3	60.0	0	0.0	0	0.0
May	0	0.0	5	9.1	2	9.1	. 1	14.3	44	9.9	0	0.0	0	0.0	3	21.4
June	0	0.0	10	18.2	0	0.0	3	42.9	46	10.3	2	40.0	0	0.0	1	7.1
July	0	0.0	3	5.5	2	9.1	1	14.3	38	8.5	0	0.0	0	0.0	0	0.0
August	0	0.0	1	1.8	1	4.5	0	0.0	9	2.0	0	0.0	0	0.0	1	7.1
November	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	. 0	0.0
December	<u>0</u>	0.0	_0	0.0	0	0.0	<u>o</u>	0.0	0	0.0	<u>0</u>	0.0	<u>0</u>	0.0	_0	0.0
Totals	3	100.0%	55	100.1%	22	99.9%	7	100.1%	446	100.0%	5	100.0%	2	100.0%	14	99.9%

Table III-27 (Continued) DATE INCIDENT OCCURRED (MONTH) BY INCIDENT LOCATION TYPE

Continued

								Locati	on Type							
Month	Community Auto		<u>Community Taxi</u> <u>Number Percent</u>		Community Bus Number Percent		Community Truck Number Percent		Sidewalk Number Percent		Street Number Percent		Water Number Percent		Other Number Percen	
February	0	0.0	0	0.0	0	0.0	3	50.0	13	15.1	29	21.5	0	0.0	10	15.4
March	0	0.0	0	0.0	1	100.0	0	0.0	10	11.6	17	12.6	0	0.0	6	9.2
April	0	0.0	0	0.0	0	0.0	0	0.0	10	11.6	14	10.4	0	0.0	11	16.9
May	2	50.0	0	0.0	0	0.0	1	16.7	15	17,4	17	12.6	0	0 ~ 0	9	13.8
June	0	0.0	2	50.0	0	0.0	1	16.7	8	9.3	19	14.1	0	0.0	13	20.0
July	2	50.0	0	0.0	0	0.0	0	0.0	12	14.0	7	5.2	0	0.0	4	6.2
August	0	0.0	0	0.0	0	0.0	0	0.0	3	3.5	0	0.0	1	100.0	0	0.0
November	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
December	<u>0</u>	0.0	<u>0</u>	0.0	<u>0</u>	0.0	<u>0</u>	0.0	_0	0.0	0	0.0	<u>0</u>	0.0	_0	0.0
Totals	4	100.0%	. 4	100.0%	1	100.0%	6	100.1%	86	99.9%	135	100.1%	1	100.0%	65	100.0%
Missin	g Cases	5727														

Table III-27 (Continued) DATE INCIDENT OCCURRED (MONTH) BY INCIDENT LOCATION TYPE

\$


Table III-28

TIME INCIDENT OCCURRED (AM-PM) BY TIME INCIDENT REPORTED (AM-PM)

		****	Time Rep	orted	
m.t.	P	<u>M</u>		A	M
Occurred	Number	Percent		Number	Percent
PM	2643	96.5%		102	7.0%
AM	95	3.5		1365	93.0
Totals	2738	100.0%		1467	100.0%

Table III-29

DATE INCIDENT OCCURRED (MONTH) BY DATE INCIDENT REPORTED (MONTH)

	Date Reported													
Date <u>Dccurred</u>	Ja	nuary	Feb	ruary	M	arch	A	pril]	May	J	une		
Date <u>Occurred</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent		
January	705	99.3%	6	0.9%	1	0.2%	0	0.0%	0	0.0%	0	0.0%		
February	0	0.0	633	98.3	7	1.1	1	0.2	0	0.0	l	0.1		
March	0	0.0	2	0.3	653	98.2	5	0.8	1	0.1	1	0.1		
April	0	0.0	0	0.0	1	0.2	635	98.8	6	0.8	1	0.1		
May	0	0.0	0	0.0	i	0.2	1	0.2	710	99.0	6	0.8		
June	0	0.0	0	0.0	1	0.2	0	0.0	0	0.0	708	98.7		
July	0	0.0	0	0.0	0	0.0	1	0.2	0	0.0	0	0,0		
August	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0		
September	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0		
October	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0		
November	1	0.1	3	0.5	0	0.0	0	0.0	0	0.0	0	0.0		
December	4	0.6	0	0.0	1	0.2	0	0.0	0	_0.0_	0	0.0		
Totals	710	99.9%	644	99 . 7%	665	100.3%	643	100.2%	717	99.9%	717	99.8%		

٠

Continued

.

Table III-29 (Continued)

DATE INCIDENT OCCURRED (MONTH) BY DATE INCIDENT REPORTED (MONTH)

		<u></u>					L	ocation I	уре	· · ·		
	Jı	ily	Au	gust	Sep	tember	- 0c	tober	Nov	ember	Dec	ember
Month	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
January	2	0.3%	0	0.0%	0	0.0%	0	0.0%	2	0.3	0	0.0%
February	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
March	1	0.2	. 0	0.0	0	0.0	1	0,2	0	0.0	0	0.0
April	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
May	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
June	. 4	0.6	0	0.0	0	0,0	0	0,0	0	0.0		0.0
July	630	98.6	3	0.4	2	0.3	1	0.2	1	0.2	0	0.0
August	0	0.0	673	99.6	9	1.5	0	0.0	0	0.0	0	0.0
September	1	0.2	0	0.0	578	98.1	4	0.7	0	0.0	0	0.0
October	0	0.0	0	0.0	0	0.0	603	98.9	6	1.0	2	0.4
November	1	0.2	0	0.0	0	0.0	0	0.0	594	98.5	5	0.9
December	0	0.0	0	0.0	0	0.0	1	0.2	0	0.0	531	98.7
Totals	639	99.8%	676	100.0%	589	99.9%	610	100.2%	603	100.0%	538	100.0%



CHAPTER IV TABLES

APPENDIX E

Table IV-1

RESPONSE TIME BY CRIME CODE

										Crime	Туре									
Response	Mun Mansla	rder- aughter Percent	Rol	Dery	As	sault	Burg	glary	Larcer Thar	ny Less n \$50 Percent	Larceny That	Greater n \$50	Motor Tl	Vehicle heft	Disor Cor	nderly nduct	Far Offe	nily enses	Proj Dai	perty nage
111110	number	rercent	Hunder	rercent	Munder	rercent	Number	rercent	Number	rercent	namber	rercent	number	rercent	Mumber	reicent	number	rercent	Number	recent
1- 3 Minutes	1	100.0%	1	100.0%	2	50.0%	10	55.6%	5	38.5%	1	16.7%	9	50.0%	5	62.5%	6	66.7%	8	34.8%
4- 6 Minutes	. 0	0.0	0	0.0	1	25.0	5	27.8	7	53,8	4	66.7	4	2,2.2	3	37.5	3	33.3	10	43.5
7-10 Minutes	0	0.0	0	0.0	1	25.0	3	16.7	1	7.7	1	16.7	3	16.7	0	0.0	0	0.0	3	13.0
11 Minutes and Above	<u>0</u>	0.0	<u>0</u>	0.0	<u>0</u>	0.0	0	0.0	_0	0.0	<u> </u>	0.0	_2	11.1	<u>0</u>	0.0	<u>0</u>	0.0	_2	8.7
Totals	1	100.0%	1	100.0%	4	100.0%	18	100.0%	13	100.0	\$6	100.0%	1.8	100.0%	8	100.0%	9	100.0%	23	100.0%

Continued

Table IV-1 (Continued) RESPONSE TIME BY CRIME CODE

	<u> </u>				<u></u>				Crime	Туре							·····	
Time	Nuis	sances	Ala	irms	Motor Viola	Vehicle tions	Me Pro	dical blems	Misce Juve	llaneous enile	Other	-Person_	<u>Other-I</u>	Propertý	V:	Lce	No L	icident_
Response	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
1- 3 Minutes	6	54.5%	18	85.7%	4	66.7%	6	54.5%	14	50.0%	6	54.5%	1	25.0%	0	0.0%	3	30.0%
4- 6 Minutes	3	27.3	` 3	14,3	2	33.3	1	9.1	7	25.0	3	27.3	2	50.0	1	100.0	4	40.0
7-10 Minutes	0	0.0	0	0.0	0	0.0	. 4	36.4	6	21.4	0	0.0	0	0.0	0	0.0	3	30.0
ll Minutes and Above	2	18.2	_0	0.0	<u>0</u>	0.0	_0	0.0	_1	3.6	_2	18.2	<u>1</u>	25.0	<u>0</u>	0.0	_0	0.0
Totals	11	100.0%	21	100.0%	6	100.0%	11	100.0%	28	100.07	11	100.0%	4	1.00.0%	1	100.0%	10	100.0%
Mie	ssing Ca	865	7813															



Table IV-2

FREQUENCY DISTRIBUTION OF THE FOUR MAJOR CAR POST SECTORS

Car Post Sectors	Number	Percent
Amber	1314	24.7%
Blue	1230	23.1
Green	1906	35.8
Red	872	16.4
Totals	5322	100.0%

Missing Cases

۲

Table IV-3

۵

2)

FREQUENCY DISTRIBUTION OF MAJOR GRID LOCATIONS

Grid	Number	Percent
003	152	10.3%
009	345	23.3
0011	174	11.7
0016	168	11.4
0017	246	16.6
008a	395	26.7
Totals	1480	100.0%

Table IV-4A

62

6

MAJOR CAR POST SECTORS BY CRIME CODE CONTROLLING FOR THE 3 P.M. TO 2:59 A.M. TIME SEGMENT

	Car Post Sector											
	R	ed	Gr	een	B1	ue	Am	ber				
<u>Crime</u> Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent				
Robbery	4	1.3%	12	1.8%	5	1.2%	5	1.1%				
Assault	16	5.1	47	7.2	15	3.7	26	5,8				
Burglary	14	4.5	34	5.2	16	3.9	11	2.4				
Larceny Less Than \$50	14	4.5	38	5.8	16	3.9	33	7.3				
Larceny Greater Than \$50	12	3.8	30	4.6	25	6.1	27	6.0				
Motor Vehicle Theft	46	14.7	59	9.1	43	10.5	39	8.6				
Disorderly Conduct	20	6.4	35	5.4	32	7.8	32	7.1				
Family Offenses	22	7.1	23	3.5	23	5.6	18	4.0				
Property Damage	1	4.2	59	9.1	37	9.0	36	8.0				
Nuisances	9	2.9	40	6.1	17	4.2	24	5.3				
Alarms	38	12.2	61	9.4	38	9.3	46	10.2				
Motor Vehicle Violations	16	5.1	22	3.4	17	4.2	12	2.7				
Medical Problems	10	3.2	31	4.8	10	2.4	18	4.0				
Miscellaneous Juvenile	31.	9.9	70	10.8	58	14.2	56	12.4				
Other-Person	21	6.7	40	6.1	25	6.1	24	5.3				
Other Property	5	1.6	12	1.8	7	1.7	8	1.8				
Vice	4	1.3	11	1.7	9	2.2	15	3.3				
No Incident	<u>17</u>	5.4	<u>27</u>	4.1	<u>16</u>	<u>3.9</u>	21	4.7				
Totals	312	100.0%	651	100.0%	409	100.0%	451	100.0%				

Table IV-4B

MAJOR CAR POST SECTORS BY CRIME CODE CONTROLLING FOR THE 3 A.M. TO 2:59 P.M. TIME SEGMENT

	Car Post Sector											
	R	ed	Gr	een	<u>B1</u>	ue	An	ber				
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent				
Robbery	0	0.0%	3	0.7%	3	1.3%	1	0.3%				
Assault	8	4.4	24	5.9	9	3.8	14	4.7				
Burglary	16	8.8	36	8.8	13	5.5	1.8	6.1				
Larceny Less Than \$50	7	3.9	23	5.6	7	2.9	17	5.8				
Larceny Greater Than \$50	8	4.4	26	6.4	16	6.7	25	8.5				
Motor Vehicle Theft	37	20.4	45	11.0	42	17.6	27	9.2				
Disorderly Conduct	: 8	4.4	11	2.7	5	2.1	5	1.7				
Property Damage	6	3.3	32	7.8	15	6.3	28	9.5				
Nuisances	12	6.6	31	7.6	14	5.9	14	4.7				
Alarms	24	13.3	46	11.3	27	11.3	42	14.2				
Motor Vehicle Violations	3	1.7	21	5.1	9	3.8	. 14	4.7				
Medical Problems	8	4.4	19	4.7	6	2.5	10	3.4 .				
Miscellaneous Juvenile	16	8.8	43	10.5	33	13.9	36	12.2				
Other-Person	8	4.4	18	4.4	10	4.2	7	2.4				
Other-Property	3	1.7	7	1.7	10	4.2	7	2.4				
Vice	2	1.1	5	1.2	4	1.7	10	3.4				
No Incident	<u>3</u>	<u>1.7</u>	<u>7</u>	<u>1.7</u>	<u>6</u>	2.5	<u>7</u>	2.4				
Totals	181	100.0%	408	106.0%	238	100.0%	295	100.0%				



Tante to ho	1	'a	Ь	16	ġ.	ľ	V-V	5A
-------------	---	----	---	----	----	---	-----	----

MAJOR GRID CATEGORIES BY CRIME CODE CONTROLLING FOR THE 3PM TO 2:59AM TIME SEGMENT

	<u></u>			·		Gr	id					
		003	(009	(011		016		017	0	08a
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Robbery	1	1.0%	0	0.0%	2	2.0%	1	1.0%	4	2.4%	3	1.1%
Assault	13	12.9	15	7.5	4	4.1	7	6.9	14	8.4	26	10.0
Burglary	3	3.0	4	2.0	2	2.0	3	2.9`	4	2.4	6	2.3
Larceny Less Than \$50	0	0.0	27	13.6	0	0.0	21	20.6	6	3,6	5	1.9
Larceny Greater Than \$50	3	3.0	16	8.0	4	4.1	2	2.0	6	3.6	7	2.7
Motor Vehicle Theft	15	14.9	7	3.5	7	7.1	5	4.9	20	12.0	11	4.2
Disorderly Conduct	3	3.0	12	6.0	9	9.2	6	5.9	6	3.6	8	3.1
Family Offenses	11	10.9	1	0.5	2	2.0	1	1.0	5	3.0	11	4.2
Property Damage	4	4.0	6	3.0	7	7.1	4	3.9	14	8.4	7	2.7
Nuisances	4	4.0	4	2.0	6	6.1	10	9.8	7	4.2	13	5.0
Alarms	1	1.0	16	8.0	16	16.3	7	6.9	9	5.4	7	2.7
Motor Vehicle Violations	0	0.0	10	5.0	7	7.1	3	2.9	7	4.2	4	1.5
Medical Problems	23	22.8	36	18.1	6	6.1	· 11	10.8	16	9.6	88	33.7
Miscellaneous Juvenile	8	7.9	9	4.5	4	4.1	6	5.9	17	10.2	38	14.6
Other-Person	7	6.9	17	8.5	9	9.2	5	4.9	11	5.6	12	4.6
Other-Property	3	3.0	7	3.5	5	5.1	7	6.9	6	3.6	3	1.1
Vice	1.	1.0	10	5.0	5	5.1	2	2.0	7	4.2	7	2.7
No Incídent	1	1.0	2	1.0	3	3.1	1	1.0	8	4.8	5	1.9
Totals	101	100.0%	199	100.0%	98	100.0%	102	100.0%	167	100.0%	261	100.0%

Table IV-5B

MAJOR GRID CATEGORIES BY CRIME CODE CONTROLLING FOR THE 3AM TO 2:59 PM TIME SEGMENT

						Gr	id		·····			
	(003		009		011		016	(017	0	08a
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Robbery	0	0.0%	0	0.0%	0	0.0%	0	0.0%	2	2.7%	2	1.5%
Assault	2	5.7	5	4.4	0	0.0	2	3.2	0	0.0	4	3.1
Burglary	4	11.4	5	4.4	8	11.1	4	6.3	5	6.7	2	1.5
Larceny Less Than \$50	1	2,9	15	13.3	4	5.6	8	12.7	4	5.3	3	2.3
Larceny Greater Than \$50	1	2.9	8	7.1	3	4.2	3	4.8	4	5.3	2	1.5
Motor Vehicle Theft	11	31.4	3	2.7	3	4.2	2	3.2	12	16.0	.18	13.7
Disorderly Conduct	1	2.9	7	6.2	3	4.2	0	0.0	1	1.3	4	3.1
Family Offenses	2	5.7	0	0.0	2	2.8	0	0.0	2	2.7	5	3.8
Property Damage	1	2.9	0	0.0	9	12.5	6	9.5	5	6.7	5	3.8
Nuisances	0	0.0	2	1.8	5	6.9	12	19.0	5	6.7	10	7.6
Alarms	0	0.0	9	8.0	10	13.9	7	11.1	7	9.3	1	0.8
Motor Vehicle Violations	1	2.9	2	1.8	2	2.8	2	3.2	1	1.3	3	2.3
Medical Problems	10	28.6	20	17.7	5	6.9	6	9.5	10	13.3	48	36.6
Miscellaneous Juvenile	1	2.9	10	8.8	6	8.3	6	9.5	7	9.3	12	9.2
Other-Person	0	0.0	15	13,3	5	6.9	4	6.3	4	5.3	4	3.1
Other-Property	0	0.0	9	8.0	3	4.2	0	0.0	2	2,7	3	2.3
Vice	0	0.0	3	2.7	1	1.4	Ó	0.0	3	4.0	3	2.3
No Incident	_0	0.0	0	0.0	. <u>3</u>	4.2		1.6	<u> </u>		2	1.5
Totals	35	100.0%	113	100.0%	72	100.0%	63	100.0%	75	100.0%	131	100.0%



Table IV-6A

CAR POST SECTOR BY CRIME CODE CONTROLLING FOR THE WINTER SEASON (DEC - JAN - FEB)

	Car Post Sector										
]	Red	G	reen	A	mber	Blue				
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent			
Murder-Manslaughter	• 0	0.0%	1	0.2%	0	0.0%	0	0.0%			
Robbery	2	1.0	3	0.7	4	1.2	4	1.2			
Assault	8	3.9	26	5.8	11	3.4	13	3.8			
Burglary	18	8.8	31	6.9	16	4.9	28	8.2			
Larceny Less Than \$50	7	3.4	25	5.5	20	6.1	14	4.1			
Larceny Greater Than \$50	7	3.4	19	4.2	19	5.8	19	5.6			
Motor Vehicle Theft	36	17.6	60	13.3	42	12.8	50	14.7			
Disorderly Conduct	7	3.4	13	2.9	14	4.3	19	5.6			
Family Offenses	14	6.8	14	3.1	10	3.0	18	5.3			
Property Damage	12	5.9	49	10.8	39	11.9	37	10.9			
Nuisances	9	4.4	23	5.1	7	2.1	15	4.4			
Alarms	28	13.7	48	10.6	34	10.4	31	9.1			
Motor Vehicle Violations	12	5.9	22	4.9	15	4.6	13	3.8			
Medical Problems	8	3.9	15	3.3	16	4.9	2	0.6			
Miscellaneous Juvenile	15	7.3	48	10.6	33	10.1	34	10.0			
Other-Person	9	4,4	28	6.2	13	4.0	20	5.9			
Other-Property	3	1.5	10	2.2	6	1.8	12	3.5			
Vice	3	1.5	4	0.9	15	4.6	1	0.3			
No Incident	7	3.4	_13	2.9	_14	4.3	_11	3.2			
Totals	205	100.0%	452	100.0%	328	100.0%	341	100.0%			

Table IV-6B

CAR POST SECTOR BY CRIME CODE CONTROLLING FOR THE SPRING SEASON (MAR - APR - MAY)

	Car Post Sector											
]	Red	G	reen	<u>B</u>	lue	A	nber				
Crime Code	Number	Percent	Number	Pezcent	Number	Percent	Number	Percent				
Robbery	0	0.0%	7	1.4%	2	0.6%	2	0.5%				
Assault	9	4.3	26	5.1	14	4.2	18	4.8				
Burglary	16	7.7	30	5.9	19	5.7	14	3.8				
Larceny Less Than \$50	7	3.3	25	4.9	10	3.0	24	6.5				
Larceny Greater Than \$50	7	3.3	31	6.1	23	6.9	24	6.5				
Motor Vehicle Theft	29	13.9	49	9.6	39	11.8	36	9.7				
Disorderly Conduct	7	3.3	29	5.7	20	6.0	27	7.3				
Family Offenses	14	6.7	13	2.6	19	5.7	15	4.0				
Property Damage	11	5.3	59	11.6	26	7.9	36	9,7				
Nuisances	7	3.3	34	6.7	19	5.7	28	7.5				
Alarms	21	10.0	42	8.3	27	8.2	33	8.9				
Motor Vehicle Violations	8	3.8	14	2.8	18	5.4	15	4.0				
Medical Problems	8	3.8	19	3.7	10	3.0	7	1.9				
Miscellaneous Juvenile	27	12.9	63	12.4	45	13.6	50	13.4				
Other-Person	16	7.7	21	4.1	13	3.9	17	4.6				
Other-Property	6	2.9	10	2.0	8	2.4	8	2.2				
Vice	2	1.0	11	2.2	8	2.4	7	1.9				
No Incident	_14	6.7		4.9	_11	3.3	_11	3.0				
Totals	209	100.0%	508	100.0%	331	100.0%	372	100.0%				

Table IV-6C

CAR POST SECTOR BY CRIME CODE CONTROLLING FOR THE SUMMER SEASON (JUNE — JULY — AUG)

	Car Post Sector											
	<u></u>	Red	G	reen	B	lue	A	nber				
Crime Type	Number	Percent	Number	Percent	Number	Percent	Number	Percent				
Robbery	0	0.0%	5	1.0%	2	0.6%	1	0.3%				
Assault	12	5.4	31	6.0	19	6.1	18	5.4				
Burglary	11	4.9	28	5.4	26	8.3	12	3.6				
Larceny Less Than \$50	10	4.5	33	6.3	14	4.5	19	5.7				
Larceny Greater Than \$50	12	5.4	32	6.2	26	8.3	28	8.4				
Motor Vehicle Theft	26	11.7	41	7.9	32	10.3	35	10.5				
Disorderly Conduct	15	6.7	32	5.2	17	5.4	11	3.3				
Family Offenses	11	4.9	17	3.3	12	3.8	8	2.4				
Property Damage	13	5.8	48	9.2	28	9.0	25	7.5				
Nuisances	7	3.1	41	7.9	18	5.8	18	5.4				
Alarms	31	13.9	45	8.7	27	8.7	42	12.7				
Motor Vehicle Violations	8	3.6	23	4.4	7	2.2	14	4.2				
Medical Problems	7	3.1	34	6.5	8	2.6	8	2.4				
Miscellaneous Juvenile	31	13.9	51	9.8	39	12.5	45	13.6				
Other-Person	9	4.0	21	4.0	15	4.8	20	6.0				
Other-Property	5	2.2	12	2.3	5	1.6	5	1.5				
Vice	2	0.9	8	1.5	3	1.0	8	2.4				
No Incident	_13	5.8	18	3.5	_14	4.5		4.5				
Totals	223	100.0%	520	100.0%	312	100.0%	332	100.0%				

Table IV-6D

CAR POST SECTOR BY CRIME CODE CONTROLLING FOR THE FALL SEASON (SEPT - OCT - NOV)

	Car Post Sector											
]	Red	G	reen	B	lue	A1	nber				
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent				
Robbery	3	1.3%	4	0.9%	Б	2.4%	3	1.1%				
Assault	13	5.6	27	6.4	12	4.9	14	5.0				
Burglary	23	9.9	37	8.7	18	7.3	24	8.6				
Larceny Less Than \$50	12	5.2	26	6.1	8	3.3	20	7.2				
Larceny Greater Than \$50	12	5.2	35	8.3	22	9.0	27	9.7				
Motor Vehicle Theft	45	19.3	49	11.6	37	15.1	21	7.5				
Disorderly Conduct	14	6.6	18	4.2	15	6.1	17	6.1				
Family Offenses	12	5.2	12	2,8	7	2.9	5	1.8				
Property Damage	20	8.6	48	11.3	25	10.2	30	.0.8				
Nuisances	6	2.6	19	4.5	16	6.5	11	3.9				
Alarms	20	8.6	39	9.2	27	11.0	32	11.5				
Motor Vehicle Violations	8	3.4	14	3.3	8	3.3	8	2.9				
Medical Problems	5	2.1	17	4.0	4	1.6	10	3.6				
Miscellaneous Juvenile	20	8.6	43	10.1	18	7.3	30	10.8				
Other-Person	11	4.7	17	4.0	6	2.4	7	2.5				
Other-Property	2	0.9	7	1.7	5	2.0	5	1.8				
Vice	2	0.9	6	1.4	6	2.4	12	4.3				
No Incident	5		6	1.4	5	2.0	3					
Totals	233	100.0%	424	100.0%	245	100.0%	279	100.0%				



Table IV-7A

đ)

CRIME CODE BY LOCATION STATUS CONTROLLING FOR THE RED CAR POST SECTOR

	Location Status												
	Resi	dential	Bus	iness	Publ	ic Area	Bui	ldings	A11 V	ehicles	0	ther	
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	
Robbery	0	0.0%	2	3.3%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	
Assault	7	6.4	1	1.6	6	20.0	1	11.1	1	1.4	0	0.0	
Burglary	11	10.1	10	16.4	0	0.0	0	0.0	0	0.0	0	0.0	
Larceny Less Than \$50	4	3.7	4	6.6	1	3.3	0	0.0	1	1.4	0	0.0	
Larceny Greater Than \$50	5	4.6	1	1.6	0	0.0	0	0.0	6	8,1	0	0.0	
Motor Vehicle Theft	2	1.8	0	0.0	6	20.0	0	0.0	40	54.1	1	20.0	
Disorderly Conduct	8	7.3	3	4.9	1	3.3	0	0.0	0	0.0	0	0.0	
Family Offenses	14	12.8	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0	
Property Damage	2	1.8	6	9.8	1	3.3	1	11.1	4	5.4	1	20.0	
Nuisances	5	4.6	0	0.0	2	6.7	1	11.1	3	4.1	1	20.0	
Alarms	3	2.8	27	44.3	1,	3.3	4	44.4	1	1.4	1	20.0	
Motor Vehicle Violations	1	0.9	0	0.0	2	6.7	0	0.0	11	14.9	0	0.0	
Medical Problems	9	8.3	0	0.0	0	0.0	2	22.2	0	0.0	0	0.0	
Miscellaneous Juvenile	19	17.4	3	4.9	3	10.0	0	0.0	1	1.4	1	20.0	
Other-Person	10	9.2	3	4.9	4	13.3	0	0.0	1	1.4	0	0.0	
Other-Property	0	0.0	0	0.0	1	3.3	0	0.0	4	5.4	0	0.0	
Vice	2	1.8	0	0.0	1	3.3	0	0.0	1	1.4	0	0.0	
No Incident	7	6.4	_1	1.6		3.3	<u>0</u>	0.0	_0	0.0	<u>0</u>	0.0	
Totals	109	100.0%	61	100.0%	30	100.0%	9	100.0%	74	100.0%	5	100.0%	

Table IV-7B

CRIME CODE BY LOCATION STATUS CONTROLLING FOR THE GREEN CAR POST SECTOR

		······································				Location	n Statu	3				
	Resi	dential	Bus	iness	Pub1:	ic Area	Bui	ldings	A11 V	ehicles	0	ther
<u>Crime Code</u>	Number	Percent	Number	<u>Percent</u>	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Murder-Manslaughter	1	0.4%	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%
Robbery	0	0.0	7	5.6	1	1.2	0	0.0	0	0.0	0	0.0
Assault	13	5.7	7	5.6	12	14.1	5	14.3	1	0.7	1	11.1
Burglary	29	12.6	16	12.7	0	0.0	0	0.0	0	0.0	1	11.1
Larceny Less Than \$50	10	4.3	12	9.5	3	3.5	2	5.7	8	5.9	0	0.0
Larceny Greater Than \$50	13	5.7	1	0.8	3	3.5	1	2.9	6	4.4	1	11.1
Motor Vehicle Theft	7	3.0	4	3.2	8	9.4	1	2.9	52	38.5	0	0.0
Disorderly Conduct	18	7.8	6	4.8	3	3.5	3	8.6	0	0.0	0	0.0
Family Offenses	21	9.1	1	0.8	0	0.0	0	0.0	0	0.0	0	0.0
Property Damage	21	9.1	16	12.7	3	3.5	6	17.1	23	17.0	2	22.2
Nuisances	10	4.3	4	3.2	9	10.6	2	5.7	13	9.6	1	11.1
Alarms	8	3.5	32	25.4	3	3.5	3	8.6	1	0.7	0	0.0
Motor Vehicle Violations	· 2	0.9	3	2.4	5	5.9	0	0.0	19	14.1	0	0.0
Medical Problems	10	4.3	1	0.8	9	10.6	4	11.4	0	0.0	1	11.1
Miscellaneous Juvenile	35	15.2	7	5.6	11	12.9	5	14.3	1	0.7	1	11.1
Other-Person	17	7.4	3	2.4	5	5.9	2	5.7	3	2.2	1	11.1
Other-Property	5	2.2	4	3.2	3	3.5	0	0.0	3	2.2	0	0.0
Vice	1	0.4	1	0.8	2	2.4	1	2.9	1	0.7	0	0.0
No Incident	9	3.9	1	0.8		5.9		0.0	4	3.0	<u>o</u>	0.0
Totals	230	100.0%	126	100.0%	85	100.0%	35	100.0%	135	100.0%	9	100.0%

Table IV-7C

CRIME CODE BY LOCATION STATUS CONTROLLING FOR THE AMBER CAR POST SECTOR

	<u></u>			··· <u>··································</u>		Location	n Statu	<u>S</u>				
	Residential		Bus:	Business		ic Area	Buildings		<u>A11 V</u>	ehicles	0	ther
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Robbery	1	0.8%	1	1.0%	3	3.8%	0	0.0%	0	0.0%	0	0.0%
Assault	10	8.5	5	5.1	8	10.1	1	5.3	1	1.1	0	0.0
Burglary	9	7.6	9	9.1	0	0.0	0	0.0	0	0.0	0	0.0
Larceny Less Than \$50	2	1.7	6	6.1	7	8.9	0	0.0	6	6.8	1	7.1
Larceny Greater Than \$50	8	6.8	3	3.0	8	10.1	3	15.8	9	10.2	2	14.3
Motor Vehicle Theft	1	0.8	1	1.0	8	10.1	0	0.0	34	38.6	1	7.1
Disorderly Conduct	. 7	5.9	8	8.1	6	7.6	1	5.3	0	0.0	0	0.0
Family Offenses	14	11.9	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Property Damage	12	10.2	9	9.1	6	7.6	2	10.5	16	18.2	2	14.3
Nuisances	2	1.7	1	1.0	3	3.8	0	0.0	4	4.5	2	14.3
Alarms	3	2.5	28	28.3	4	5.1	4	21.1	1	1.1	0	0.0
Motor Vehicle Violations	0	0.0	0	0.0	2	2.5	1	5.3	10	11.4	1	7.1
Medical Problems	5	4.2	2	2.0	3	3.8	1	5.3	0	0.0	2	14.3
Miscellaneous Juvenile	16	13.6	16	16.2	8	10.1	3	15.8	3	3.4	2	14.3
Other-Person	12	10.2	2	2.0	3	3.8	0	0.0	0	0.0	1	7.1
Other-Property	2	1.7	2	2.0	1	1.3	1	5.3	2	2.3	0	0.0
Vice	3	2.5	3	3.0	4	5.1	2	10.5	1	1.1	0	0.0
No Incident		9_3_	3	3.0		6.3	0	0.0	1		0	0.0
Totals	113	100.0%	99	100.0%	79	100.0%	19	100.0%	88	100.0%	14	100.0%

Table IV-7D

CRIME CODE BY LOCATION STATUS CONTROLLING FOR THE BLUE CAR POST SECTOR

						Location	n Statu	3				
	Residential		Bus	iness	Publ:	ic Area	Bui	ldings	All Vehicles		0	ther
Crime Code	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Robbery	0	0.0%	1	1.1%	1	1.8%	0	0.0%	0	0.0%	1	11.1%
Assault	5	3.3	4	4.6	1	1.8	1	5.9	0	0.0	1	11.1
Burglary	17	11.2	16	18.4	0	0.0	0	0.0	0	0.0	0	0.0
Larceny Less Than \$50	7	4.6	2	2.3	6	10.7	1	5.9	2	2.3	1	11.1
Larceny Greater Than \$50	15	9.9	2	2.3	4	7.1	1	5.9	9	10.2	0	0.0
Motor Vehicle Theft	2	1.3	0	0.0	10	17.9	0	0.0	36	40.9	1	11.1
Disorderly Conduct	15	9.9	б	6.9	2	3.6	0	0.0	0	0.0	0	0.0
Family Offenses	23	15.1	0	0.0	0	0.0	0	0.0	0	0.0	0	0.0
Property Damage	15	9.9	9	10.3	2	3.6	2	11.8	10	11.4	0	0.0
Nuisances	4	2.6	2	2.3	2	3.6	1	5.9	9	10.2	0	0.0
Alarms	1	0.7	24	27.6	0	0.0	5	29.4	0	0.0	2	22.2
Motor Vehicle Violations	· 2	1.3	0	0.0	4	7.1	1	5.9	10	11.4	1	11.1
Medical Problems	3	2.0	- 0	0.0	5	8.9	0	0.0	0	0.0	0	0.0
Miscellaneous Juvenile	26	17.1	12	13.8	11	19.6	2	11.8	4	4.5	0	0.0
Other-Person	4	2.6	3	3.4	3	5.4	0	0.0	5	5.7	0	0.0
Other-Property	3	2.0	2	2.3	1	1.8	1	5.9	3	3.4	1	11.1
Vice	3	2.0	1	1.1	2	3.6	1	5.9	0	0.0	0	0.0
No Incident	7	4.6	. <u>3</u>	3.4	2	3.6		5.9	0	0.0	<u>1</u>	_11.1_
Totals	152	100.0%	87	100.0%	56	100.0%	17	100.0%	88	100.0%	9	100.0%

BIBLIOGRAPHY FOR POLICE PRODUCTIVITY STUDIES

- Angell, John and Fontaine Hagedorn. "A Municipal System for Improving Local Police." Journal of Police Science and Administration, Vol. 4, No. 2 (1976), pp. 217-222.
- Anonymous. <u>Burglary in San Jose</u>. Washington, D.C.: LEAA, NILE/CJ, Report No. LEAA NI 70-023B; PB211 789, February, 1972.
- Anonymous. <u>Law Enforcement Manpower Resource Allocation System</u> (LEMRAS) <u>Application Description Manual</u>. White Plains, N.Y.: IBM Corp., September 1969.
- Benton, John B. <u>Managing the Organizational Decision Process</u>. Lexington, Mass.: Lexington Books, 1973.
- Biderman, Albert D. "Surveys of Population Samples for Estimating Crime Incidence." <u>The Annals of the American Academy of Political and</u> <u>Social Science</u>, November 1967.
- Bloch, Peter. <u>Equality of Distribution of Policy Services--A Case</u> <u>Study of Washington</u>. Washington; The Urban Institute, 1974.
- Bloch, P. B. and D. I. Specht. <u>Neighborhood Team Policing</u>. Washington; The Urban Institute, 1973.
- Bloch, P. B. and D. I. Specht. <u>Evaluation of Operation Neighborhood</u>. Washington: The Urban Institute, 1973.
- Buren, Michael. "A Police Management Training Program: Effective Use of Manpower and Money." Journal of Police Science and Administration, Vol. No. 1, 1973, pp. 294-302.
- Burgin, A. Lad. "Organization Development: A Strategy for Improving Police Effectiveness. O. D. Activities within the Columbus, Ohio, Division of Police." Police Chief, May 1977.
- Cann, William. "4/40 Basic Team Concept." Law and Order. Vol. 21, No. 5, May 1973. pp. 16-22+.
- Clawson, Calvin. "A Theoretical Approach to the Allocation of Police Preventive Patrol." Police Chief, July 1973, pp. 53-59.
- Cohen, Bernard and Jan M. Charken. <u>Police Background Characteristics</u> and Performance. Lexington Books. 1973.

Commonwealth of Pennsylvania. Department of Community Affairs, Bureau of Policy Planning. <u>Effective Municipal Police Services</u>. 1974. BIBLIOGRAPHY FOR POLICE PRODUCTIVITY STUDIES

 $\hat{\mathcal{V}}$

Ó



- Cox, David L. and Warren L. Maddox, "The Police Budget and Indicess of Productivity." Police Chief, Vol. XLIII, No. 4, April 1976 pp. 44-47.
- Crabtree, Charles. <u>Investigation-Apprehension Control and Management</u> <u>System</u>. Sunnyvale, Cal.: Department of Public Safety, City of Sunnyvale, October, 1973.
- Crowther, R. F. <u>The Use of a Computer System for Police Manpower</u> <u>Allocation in St. Louis Missouri: Part I, Manpower Requirements</u> <u>for Call Answering Services</u>. Bloomington: Indiana University, 1964.
- Dart, Richard, Val Lubano and Ralph Showalter. <u>A Plan to Increase Police</u> <u>Productivity, A Report on the Reorganization of the East Hartford</u> <u>Police Department.</u> Social Development Corporation. July, 1974.
- Elliot, J. F. "Random Patrol." Syracuse, N.Y.: General Electric, Electronics Laboratory, September 1967. R67E1S-77.
- Fisk, Donald M. <u>The Indianapolis Police Fleet Plan: An Example of</u> <u>Program Evaluation for Local Government</u>. The Urban Institute. 1970.
- Gass, Saul. "On the Division of Police Districts into Patrol Beats." A Preliminary Systems and Allocation Study of the Cleveland Police Department. February 1970.
- Gass, Saul and John M. Dawson. <u>An Evaluation of Policy Related Research:</u> <u>Reviews and Critical Discussions of Policy Research in the Field</u> <u>of Police Protection</u>. Washington: U.S. Department of Commerce, October 1974.
- Greenwood, Peter. An Analysis of the Apprehension Activities of the <u>New York City Police Department</u>. New York: The New York City Rand Institute, September 1970.
- Gylys, Julius A. "Application of a Production Function to Police Patrol Activity." <u>Police Chief</u>. Volume XLI, Number 7. July, 1974. pp.70-72.
- Hamilton, E. K. <u>Productivity Bargaining and the Police</u>. Guidelines and Papers from the National Symposium on Police Labor Relations. Police Foundation. Washington. D.C. 1974
- Harrison, Frank. "Perceptions of Role Performance and Organization Effectiveness." Police Chief, April 1974, pp. 56-58.
- Hirsch, Gary B. and Lucius J. Riccio. "Measuring and Improving the Productivity of Police Patrol." <u>Journal of Police Science and</u> Administration. Vol. 2, No. 2 1974. pp.169-184.
- Hoffman, Richard B. "Perfermance Measurements in Crime Control." Journal of Research in Crime and Delinquency. Vol. 8, No. 2. July, 1971. pp. 165-174.



- Holzer, Marc. "Police Productivity: A Conceptual Framework for Measurement and Improvement." Journal of Police Science and Administration. Vol. 1, No. 4. 1973. pp. 459-467.
- Improving Police Productivity: A Special Report from the Police Foundation. <u>Nations Cities</u>. Vol. 13, March 1975. pp. 17-32.
- Indiana University School of Public and Environmental Affairs. "Protective Services in the Public Sector: Part I." <u>Midwest Monitor</u>. January/February, 1977, pp. 1-8.
- International Association of Chiefs of Police. "Productivity and Priorities." <u>The Police Yearbook</u>. September 14-18 1975.
- Kobrow, Ernst. "Measuring Police Efficiency." <u>Police Chief</u>, April 1976, p. 56.

Larson, Richard. Urban Police Patrol Analysis. Cambridge: MIT, 1972.

- Lind, Robert C. and John P. Lipsky. "The Measurement of Police Output: Conceptual Issues and Alternative Approaches." Law and Contemporary Problems, Vol. 36, No. 4. Autumn 1971. p. 569.
- Lineberry, Robert L. "Who Gets What: Measuring the Distribution of Urban Police Services." Social Science Quarterly, Vol. 54, No. 4, March, 1974, pp. 700-712.
- McDavid, James C. and David N. Allen. "Perceived Opportunities for Improving Productivity in the Harrisburg Bureau of Police." The Institute of Public Administration, The Pennsylvania State University, University Park, Pa.: August, 1976.
- Manion, Patrick. "Productivity Study--Phoenix Police Patrol Activities." American Society for Public Administration Newsletter. <u>Management</u> Science Section, March 1977.
- Maxr, Gary T. "On the Need for Alternative Measures of Police Performance." <u>A Paper Commissioned by the Police Foundation</u>. Washington, D.C. March 1973.
- Mitchell, Phillip S. Patrol Manpower Allocation and Distribution. San Diego, Cal.: San Diego County Sheriff, 1972.
- Moore, Harry W. <u>Critical Issues in Law Enforcement</u>. Cincinatti: W. A. Anderson Co., 1972.
- Moschos, D. M. "Management in the Police Labor Relations: Where Art Thou." <u>Public Management</u>, April 1977, pp. 8-9.
- Morgan, J. M. and Scott R. Fosler. "Police Productivity." <u>The Police</u> <u>Chief.</u> July, 1974. pp. 28-30.



- The National Commission on Productivity and Work Quality. <u>The Challenge</u> Of Productivity Diversity Improving Local Government Productivity <u>Measurement and Evaluation, Part III, Measuring Police-Crime Con</u>trol Productivity. Prepared by the Urban Institute. June 1972.
- The National Commission on Productivity and Work Quality. <u>Improving</u> Police Productivity: A Brief for Elected Officials.
- The National Commission on Productivity and Work Quality. <u>Imporving</u> <u>Police Productivity: More for Your Law Enforcement Dollar. A</u> <u>Brief for Elected Officials.</u> 1975
- The National Commission on Productivity and Work Quality. <u>Opportunities</u> for Improving Productivity in Police Services, Report of the Advisory Group on Productivity in Law Enforcement. 1973.
- Ostrom, Elinor. "Institutional Arrangements and the Measurement of Policy Consequences: Applications to Evaluating Police Performance." Urban Affairs Quarterly. June, 1971. pp.447-475.
- Ostrom, Elinor. "Multi-Mode Measures: From Potholes to Police." <u>Public</u> Productivity Review, March 1967, p. 57.
- Ostrom, Elinor. "On the Meaning and Measurement of Output and Efficiency in the Provision of the Urban Police Service." Journal of Criminal Justice, Vol. 1, 1973, pp. 93-112.
- Ostrom, Elinor and Robert B. Parks. "Suburban Police Departments: Too Many and Too Small?" The Urbanization of the Suburbs, Masotti, Louis H. and Jeffrey Hadden, eds., <u>Urban Affairs Annual Reviews</u>, Vol. 7. Beverly Hills, Cal.: Sage Publications, 1973.
- Ostrom, Elinor, Robert B. Parks and G. P. Whitaker, "Do We Really Want to Consolidate Urban Police Forces? A Reappraisal of Some Old Assertions." <u>Public Administration Review</u>, September-October 1973, pp. 423-432.
- Page, Louis and David Belson. "A System to Account for Costs and Effectiveness." <u>Police Chief</u>. September, 1975, pp. 42-43
- Peck, John and Manohar Nasta. "Simulating the Operation of an Urban Police Department." Police Chief, October 1975, pp. 66 ff
- Police Foundation. "Improving Police Productivity." <u>Nation's Cities</u>, March 1975, pp. 18-32.
- Police Foundation, <u>National Conference on Productivity in Policing</u>. In cooperation with the National Commission on Productivity and Work Quality.



Police Productivity. Intellect, Vol. 103, December, 1974. pp. 145-146.

- "Police Productivity Means Grief to Lawbreakers, Relief to Taxpayers." <u>Commerce Today</u>. Vol. 4, April 1, 1974. pp. 8-10.
- Presiden's Commission on Law Enforcement and Administration of Justice. The Challenge of Crime in a Free Society. New York: Avon, 1967.
- Raab, Selwyn. "Patrol Car Controversy is Just Part of Debate." <u>New</u> <u>York Times</u>, June 26, 1977.
- Reams, Ronald, Jack Kuykendall, and David Burns. "Police Management: Systems: What is an Appropriate Model?" <u>Journal of Police Science</u> <u>and Administration</u>, Vol. 3, No. 4, 1975, pp. 475-481.
- Repetto, Thomas. The Influence of Police Organization Style on Crime Control Effectiveness." Journal Of Police Science and Administration, Vol. 3, No. 3, 1975, pp. 274-279.
- Riccio, Lucius J. "Direct Deterrence: An Analysis of the Effectiveness of Police and Other Crime Prevention Technologies." Journal of Criminal Justice, Vol. 2, 1974, pp. 207-217.
- Ross, John and Jesse Burkhead. <u>Productivity in the Local Government</u> Sector. Lexington, Mass.: Lexington Books, 1974.
- Shepherd, Donald. "A Systems Approach for the Reduction of False Burglary and Robbery Alarms." <u>Police Chief</u>, June 1977, pp. 75-76.
- Skogan, Wesley G. "Efficency and Effectiveness in Big City Police Departments." <u>Public Administration Review</u>, No. 3, May-June 1976, pp. 278-285.
- Smith, Dean and David A. Espie. "Guiderines for Police Services on Controlled Access Roadways." International Association of Chiefs of Police, Gaithersburg, Md.: April 1968.
- Staudohar, Paul D. "An Experiment in Increasing Productivity of Police Service Employees." <u>Public Administration Review</u>. September-October, 1975. pp. 518-522.
- Stein, John Hollister. "Enlisting Ourselves in the War on Crime.: Washington Post, July 24, 1977, p. Bl.

 $g_{\gamma} = 0$

The Urban Institute. <u>The Challenge of Productivity Diversity: Improving</u> Local Government Productivity Measurement and Evaluation: Part III: <u>Measuring Police Crime Control Productivity; Part IV: Procedures</u> for Identifying and Evaluating Innovations--Six Case Studies. Washington: National Commission of Productivity, June 1972.

a de la


- Vandiver, James. "Acquisition and Disposition of Police Frontline Information." Journal of Police Science. Vol. 2, No. 3, 1974, pp. 288-296.
- Webster, John A. <u>The Realities of Police Work</u>. Dubuque, Iowa: Kendal/Hunt, 1973.
- Whisenand, P. M., R. O. Robinson and R. E. Hoffman. Project ACE
 (Aerial Crime Enforcement), Roverside Police Department, Final
 Report. Long Beach, Cal.: Institute for Police Studies,
 California State University, June 1972.
- Wilson, James Q. "Dilemmas of Police Administration." <u>Public Administration Review</u>, September, October 1968.

Wolfle, Joan L. and John F. Heaphy. <u>Readings in Productivity in</u> <u>Policing</u>. Police Foundation. 1975.

- *6* -

6.



